Building relationships and shared understanding: Developing a community-based participatory action research project with an Inuit community

by

Rachel Anne Estelle Cassie

A thesis submitted in partial fulfillment of the requirements for the degree of

Master of Science in Epidemiology

School of Public Health University of Alberta

© Rachel Anne Estelle Cassie, 2023

Abstract

Introduction: Evidence-based, community-led health promotion initiatives can offer a culturally-grounded option to address the effects of colonization in many Indigenous communities. Community-based research offers valuable insight into community needs and priorities when planning, delivering, and evaluating health promotion initiatives. Indigenous research paradigms, including the Inuit Aajiiqatigiingniq Research Model, have been recognized as a valuable research framework to engage in strengths-based, community-led research with Indigenous communities. Community-based (CB) and/or participatory action (PA) research methods have been widely acknowledged to complement or be situated within many Indigenous research paradigms. Building relationships is an essential component of Inuit research paradigms and CB and/or PA research methodologies as these approaches to research are oriented around trusting and reciprocal relationships. Currently, most literature discussing Indigenous research paradigms and CB and/or PA research methods focuses on the outputs or findings. More research into the earlier phases of Indigenous and CB and/or PA research is needed.

Objectives: This thesis has four, inter-related goals:

- a) Provide an in-depth narrative and critical reflection of my experiences with an Inuit community during Aajiiqatigiingniq Research Model's *piliriqatigiingniq* and *inuuqatigiitsiarniq* stages.
- b) Characterize logistical requirements needed to advance research projects during the early stages of a CBPAR project.
- c) Describe practice recommendations for researchers wishing to build community relationships and foster community engagement in a research context.
- d) Identify policy recommendations for research and funding institutions to encourage and enable CB and/or PA research methods.

Methods: This thesis is grounded in the Aajiiqatigiingniq Research Model and utilizes a community-based participatory action approach. Autoethnography data generation strategies were employed to characterize and analyze the relationship building phases of this project.

Critical reflections of work completed: To build relationships and shared understanding with the community, I have participated in multiple discussions with the community's Council, mayor, and other community members. I have worked to build trusting relationships with the community by demonstrating my relational accountability when liaising with outside institutions such as the Nunavut Research Institute, University of Alberta, and Health Canada. Work I have completed includes: 1) two ethics approvals from the University of Alberta's Research Ethics Board, 2) research licensing from the Nunavut Research Institute, and 3) securing community-held funding from Health Canada. Due to COVID-19, I have collaborated remotely with the community which has led to some relationship building-related challenges. Additionally, I have experienced timeline and financial-related pressures and limitations imposed by various institutions, including the University of Alberta and Health Canada. Strategies I have learned to develop trusting relationships with community partners include respecting and following community processes, demonstrating humility, and seeking opportunities to promote community research sovereignty.

Conclusion: This study is one of few that examines the relationship building stages of CB and/or PA research methods. Indigenous research methodologies, specifically the Aajiiqatigiingniq Research Model, necessitate conscientious relationship building between outside researchers and community members as a first step in research processes. Further research is needed to explore communities' experiences engaging in relationship building with researchers and policies that will increase Inuit research sovereignty.

iii

Preface

This thesis is an original work by Rachel Cassie. It explores the early stages of a research and health promotion project with an Inuit community. The overarching project received ethics approval by the University of Alberta Research Ethics Board and a research license from the Nunavut Research Institute. Ethics application and research license numbers are not disclosed to protect the community's identity and privacy.

Dedication

I dedicate this project to the Inuit community with whom I have worked. Words cannot express my gratitude for sharing your experiences and knowledge with me. I hope my intentions to respectfully promote your research sovereignty have been adequately conveyed and that my relationship with you all continues to grow and strengthen. I am truly honoured to have worked with and learned from you all.

Acknowledgements

I want to express my sincere gratitude to the many people who made this thesis possible. Firstly, I would like to thank my family for supporting me throughout my education journey. I would not be where I am today without your guidance, encouragement, and love.

To my friends across the globe who have listened intently to my thesis' many twists and turns, thank you for your support, questions, and humour. I am fortunate to have such an incredible support network on which to land.

To the community Councillor with whom I have worked closely for the past eighteen months, thank you for your belief in me and the project. This project would not exist if not for your passion and vision.

To my supervisor, Dr. Susan Chatwood and advisory committee member Dr. Sherilee Harper, thank you for your invaluable guidance and reassurance throughout community-based research and thesis writing processes. Community-based participatory action research is challenging and uncertain at times – thank you for helping me make sense of the uncertainty.

To the Inuk Elder who sits on my advisory committee, thank you for sharing your knowledge and perspectives with me. I am incredibly grateful to you for teaching me about your culture and sharing with me your experiences facilitating healing through Inuit methods. I am humbled to have had the opportunity to work with and learn from you.

Finally, I would like to give thanks to the institutions that provided myself and the project with financial support. Funding institutions include UAlberta North (Janet and Ashley Cameron Graduate Scholarship) and Health Canada (Substance Use and Addictions Program grant). Thank you to the staff at UAlberta North for encouraging me throughout this project. To the Senior Advisor at Health Canada with whom we have worked, thank you for your accommodations and advice as we have navigated the grant process.

Abstract	
Preface	. iv
Dedication	v
Acknowledgements	. vi
Table of Contents	vii
List of Figures	X
Chapter 1: Introduction	1
Positionality	1
Thesis outline	3
Community description History, geography, and colonization	
Community context for wellness	7
Community-based participatory action research methodology	7
Relational epistemology	9
Indigenous research methodologies and methods Relational accountability	
Inuit Qaujimajatuqangit principles	. 13
Thesis framework Aajiiqatigiingniq Research Model CBPAR Methods Intersections of CBPAR Methods and Aajiiqatigiingniq Research Model	. 15 . 17
Thesis objectives	. 19
Locating research within a larger project Rationale for larger project and study	
Chapter 2: Literature Review	. 24
Definitions and discourses	. 25
Indigenization and Decolonization	. 26
Inuit research paradigms, methodology, and methods	. 27
Community-based participatory action research methods	. 31
Guidelines for researchers wishing to work with Inuit communities on health-related topics. Summary of guidelines Research sovereignty Research ethics grounded in Inuit ways of knowing and being Financial control Community-held control of knowledge	. 35 . 35 . 37 . 37

Table of Contents

Research capacity building Critiques of the Tri-Council Policy Statement 2	
Building community relationships	
Examples and discussions of Inuit research methodologies' and participatory methods' intersections 'Changing Climate, Changing Health, Changing Stories' Profile: Using an EcoHealth Approach to Explore Impacts of Climate Change on Inuit Health	. 43 . 43
Inquiry in Health Research With Inuit Women Going Off, Growing Strong: Building Resilience of Indigenous Youth	
Inuit cultural continuity and youth	
Social connection, Inuit culture, and youth	
Colonization in Inuit Nunangat	
Examples of impacts of colonization among Inuit Tuberculosis Substance use	. 54 . 54
Discussion of strategies that aim to build cultural continuity and social connection among youth in Nunavut Evidence-based best practices School-based strategies Out-of-school strategies	. 62 . 64
Potential of community-based participatory action research to explore and foster cultural continuity and social connection among Inuit youth	. 68
Chapter 3: Methods	
Applying the Aajiiqatigiingniq Research Model and community-based participatory action research methods	. 70
Analytical autoethnographic data generation and analysis	
Ethics	
Rigour	. 77
Chapter 4: Reflections of work completed during the initial phases of the Aajiiqatigiingniq Research Model	. 79
Timeline	. 79
Health Canada grant	. 80
Community Council meetings	. 81
Nunavut Research Institute Research License	. 83
University of Alberta Ethics Approval	. 85
Chapter 5: Discussion and next steps	. 87

	les I have held during this research project planning process Moving the project through colonial structures	
S	Secured funding	90
	Liaising with other institutions	
I	_iterature review	94
Cri	tical reflections on challenges and lessons learned	95
Stra	ategies I have learned to garner community engagement and build research capacity	97
Imp	pact of COVID-19 on community-based participatory action research	100
Nez	xt steps	102
Co	nclusion	102
Refer	ences	104
Apper	ndix A – Health Canada Grant	124
i.	Application	124
ii.	Grant Reporting	135
iii.	To-be-finalized Budget	141
iv.	Budget Reallocation Request	142
Apper	ndix B – Ethics Approval and Research Licensing Documents	143
i.	Nunavut Research Institute Research License Application	143
ii.	Non-technical description of project in English and the local Inuktut language	150
iii.	Detailed description of the project	152
iv.	Consent form	158
v.	Nunavut Research Institute Research License	172
vi.	University of Alberta's Research Ethics Board Approval (first)	173
vii.	University of Alberta's Research Ethics Board Approval (second)	175

List of Figures

Figure 1. Key themes discussed in this thesis dissertation.	4
Figure 2. Map of Inuit Nunangat. (16)	6
Figure 3. Iterative process of research and action in CBPAR methods.	
Figure 4. Research and health promotion project framework	20
Figure 5. The web-like nature of the literature review.	24
Figure 6. Sikumiut model for non-Indigenous researchers' role in Inuit research. (43)	71

Chapter 1: Introduction

This thesis describes the first phase of a community-based participatory action research (CBPAR) project with an Inuit community. The research goals of this work are to explore ways to foster cultural continuity and social connection among youth in the community. These goals were then embedded within a CBPAR method.

From this project's origins, I have aimed to respond to the community's identified needs and collaborate closely with a diverse group of community members. I came to this project with humility and have been reflective on my positionality to ensure the project meets the community's needs and is community-led. In this way, I have been able to position this work to align it with community goals and timelines.

In this chapter, I describe key elements that have been considered during this CBPAR's project's early development and planning phases. This includes a description of Indigenous research and CBPAR methodologies and methods, relational epistemology and accountability, the community's cultural and wellness context, and applied research frameworks. These features of CBPAR relationship building phases are not linear, but interrelated and building upon one another.

Positionality

This thesis explores my experiences as a southern, non-Indigenous Master's student building relationships with an Inuit community during the early stages of a CBPAR project. My intention throughout this thesis is to respectfully acknowledge and reflect on Inuit knowledge given the context within which these early CBPAR phases occurred.

I am a settler Canadian, born and currently living on the unceded, stolen, and ancestral lands of the Musqueam, Squamish, and Tsleil-Waututh Nations in what we now call Vancouver, British Columbia. I am fortunate that most of my family lives in Vancouver today. My maternal grandmother's family immigrated from Scotland in the early 19th century and lived in the Canadian Prairies until the late 1930s. My maternal grandfather's family is originally from

Ireland and immigrated from England to Vancouver in the 1920s. My father's family immigrated from England in the 1950s and has predominantly British ancestry.

My experiences as a fifth-generation white settler have shaped my worldviews, values, and expectations, therefore creating biases. In accordance with multiple Indigenous rights scholars, I believe settlers play a critical role in decolonizing and Indigenizing work by supporting and following Indigenous Peoples' guidance and direction. (1,2) As a settler, understanding my role in decolonization and Indigenization work by educating myself and respectfully asking questions is fundamental to meaningful allyship with Indigenous Peoples. Continual self-reflection and humility when learning about my biases and privileges is also critical to be an effective ally. Regardless of research methods used, power imbalances between communities and research institutions are present; as a non-Indigenous researcher working with an Inuit community, it is my job to acknowledge these imbalances and self-evaluate and -critique any biases I may hold to ensure just and representative research occurs. (3)

Throughout my childhood and youth, my family and educators worked hard to foster an appreciation for Vancouver's multicultural environment. As settlers of English, Scottish, and Irish descent, my family is privileged to have our culture and language be dominant in Vancouver's society, including an education grounded in our culture and taught in our mother tongue. This privilege is one of many unjust and harmful manifestations of Canada's ongoing colonization, and comes at the expense of the Musqueam, Squamish, and Tsleil-Waututh peoples.

While not necessarily deliberate, my family recognized the importance of facilitating cultural and social connection and worked hard to facilitate these experiences for me and my brother from a young age. As a child, I consumed and learned to make traditional British foods and crafts with my mother and grandmothers. I also was a competitive Irish dancer and learned various aspects of Irish culture through this activity. These experiences facilitated inter- and intragenerational social connection, thus densifying my social network. The ease with which I accessed and practiced my culture is another example of how my settler privilege has shaped my worldviews.

My views on and approaches to substance use are guided by family experience and my work as a mental health worker in Vancouver's Downtown Eastside. In this frontline harm reduction work, I have seen the life-changing impact of accessible cultural and social connection for people experiencing harmful substance use, especially for those who are Indigenous.

It is also essential that I locate myself within the research and community context. As a Master's student at a Canadian university, I am an insider to the research process. I am an outsider to the community and recognize that I will never fully understand the community members' experiences and perspectives. I have worked hard to form genuine and reciprocal relationships with community members to respectfully bridge this gap.

I began working with a community in Nunavut in May 2021 as an intern in the area of mental health. Working closely with community members, I heard many describe a need for more research into various mental health-related challenges. When my previous Master's thesis project fell through due to a devastating flood, I reached out to a community Councillor in Nunavut to inquire if they thought the community would be interested in beginning a community-based research relationship. After discussions with multiple community members, we decided to proceed. Now, more than fifteen months later, I am writing this thesis that describes the first phase of the community engagement process. Upon graduation, I have been asked to continue working with the community to develop and implement this project as a community engaged research assistant.

Throughout the community engagement process, I have listened carefully to community members and asked questions when needed in the spirit of relationality and relational accountability. I am honoured and humbled to have had the opportunity to work with and learn from many community members.

Thesis outline

This thesis is composed of five chapters. Chapter one discusses the community's cultural context, introduces the topics of Indigenous research and CBPAR methodologies, relational

epistemology, Inuit knowledges, and an Inuit research framework, and locates this thesis within a larger health promotion and research project. Chapter two gives a comprehensive literature review on Inuit research paradigms and CBPAR methods, impacts of colonization among Inuit, and evidence-based strategies to foster cultural continuity and social connection among Inuit communities. Chapter three details my role in the overall project and this thesis' methods, data generation strategies, ethics, and rigour. Chapter four presents the timeline and work I have done to date. Chapter five concludes with a reflection of my roles throughout this CBPAR project, lessons I have learned when building relationships with the community, and next steps for the project. Figure 1 provides a visual representation of this thesis' themes.



Figure 1. Key themes discussed in this thesis dissertation.

Community description

History, geography, and colonization

As this thesis explores my experience building relationships during the early stages of a CBPAR project, I will not identify the community to respect their privacy. After discussions with multiple community members about this topic, I gained the consensus that at this point in the project development, it aligns best with the community's needs and preferences to not identify them. Many community members are hesitant to share their knowledge in academia as they have experienced stigmatizing and harmful media exposure from previous research projects that identified the community during knowledge sharing without their awareness or consent.

The community with whom I am working is a remote, fly-in community in the Inuit territory of Nunavut. Inuit Nunangat translates to "Inuit homelands" (p. 2603, 4) and is composed of four distinct regions: 1) Inuvialuit Settlement Region, 2) Nunavut, 3) Nunavik, and 4) Nunatsiavut (Figure 2). While these regions and their communities are abundantly diverse with unique geography, cultural traditions, and current social contexts, they share commonalities that are grounded in their Inuit identities. (4) Nunavut, Canada's newest territory, arose from Inuit land claims after decades of negotiations by the Inuit Tapiriit Kanatami (ITK). In 1993, the Nunavut Agreement was signed which ratified the plan to create Nunavut territory; on April 1, 1999, Nunavut separated from the Northwest Territories. Nunavut is the first territory in a settler colonial nation to be governed by an Indigenous People group. (5) Self-determination is central to Nunavut's establishment, and its governance reflects Inuit values and perspectives. There are three Regional Inuit Associations within Nunavut that represent, advocate for, and provide programming to Inuit within their jurisdictions. (6)

The community with whom I am working has a population of less than two thousand, with about one-third of the community under the age of fifteen. For residents for whom there is 2016 census data, about 95% of the community's residents identified as Inuit and 90% identified the local Inuktut language as their mother tongue. Many community members currently engage in a variety of activities that promote cultural continuity and social connection. For example, families often participate in on-the-land activities such as hunting and berry gathering, preparing and consuming country foods, and sharing cultural stories. (7,8) The accomplishments of the

community's residents are attributable to their resilience and perseverance in the face of many ongoing challenges. Canada's historical and ongoing policies of colonization, including forced relocation and assimilation, residential schools, and dog slaughters, have resulted in widespread intergenerational trauma among Inuit. (9-13) This trauma has wide-reaching impacts, including breaks in cultural continuity and social connection. (9-13) For example, on-the-land activities are not equitably accessible as they require cultural knowledge that may not have been passed intergenerationally due to colonization and expensive resources such as equipment and gasoline. (6,14,15)



Figure 2. Map of Inuit Nunangat. (16)

Community context for wellness

Indigenous understandings of and approaches to wellness often focus on the collective, rather than the individual. (17,18) IQ principles reflect this collectivist viewpoint which is rooted in a cultural emphasis placed on community wellbeing. (17–22) Indigenous groups within Canada advocate for collective approaches in both health promotion and intervention work, often extending this approach to other aspects of society including education and family life. (17–20)

It is widely understood that cultural continuity and social connection promote positive mental health among Indigenous Peoples. Cultural continuity can be defined as "[one's] identity, the practice of traditional and cultural activities, and spirituality" (p.72, 19). (23) Cultural continuity is passed intergenerationally and is considered to be a social determinant of health for Indigenous Peoples. (20,24,25) Notably, resilient mental health characteristics such as a positive sense of self-identity, belonging, and culture are thought to be facilitated by cultural continuity. (11,19,25–28) Additionally, culture's powerful role in supporting and treating mental health challenges among Indigenous Peoples is widely acknowledged. (17,18,29,30) More recently, means and mechanisms to incorporate culture into upstream approaches that aim to foster positive mental health, resiliency, and problem-solving skills among Indigenous Peoples is being explored and better understood. (11,17,19,27)

Most cultural activities are administered in a group setting, thus, social connection is usually embedded into culture-based programming. (18) Facilitating intergenerational and peer social connections are an important aspect of mental health promotion, especially among youth, as these connections are thought to increase youth's positive sense of belonging and community, resiliency, and perceived problem-solving capacity. (25,27,28) Family-based programming nurtures unique familial connections that are understood to be important to child and youth development by fostering adolescents' sense of support and stability. (27,28,31,32)

Community-based participatory action research methodology

Community-based participatory action research (CBPAR) is an intersectoral methodology that combines the strengths of community-based participatory research (CBPR) and participatory

action research (PAR). (33) CBPR and PAR are common methodologies employed in healthrelated and education research, respectively. (34) There are four primary tenets of CBPR: 1) CBPR is grounded in a questioning of assumed power relationships in Western practices of knowledge creation and sharing, 2) CBPR advocates for a dismantling of these power structures, and a return of power to those being researched, 3) CBPR recognizes the value and rigour of experiential knowledge, and 4) CBPR is strengths-based and solutions-oriented. (35) CBPR is acknowledged to be "particularly useful when working with populations that experience marginalization" (p.2, 35) because it promotes community capacity for research and related subsequent programs, and aims to ensure research processes align well with the community's culture and values. Both CBPR and PAR engage community members throughout all stages of research, including the identification of the community's needs and priorities, methods, data collection and analysis, and knowledge sharing strategies. (35-37) As well, in both CBPR and PAR, communities are "considered experts of their own experiences, with complementary knowledge and skills to contribute to the research process." (p.43, 38) (35) PAR is solutionsoriented, emphasizes the research process itself to be a knowledge sharing opportunity, and seeks to create, promote, and/or support strategies that lead to community change. (36–38)

CBPAR aims to facilitate meaningful engagement with a diverse group of community members throughout the research process and ensure that all research activities and products are community-led and -owned. (33) As the research process in CBPAR is community-directed and utilizes collaborative research techniques, there is no distinction between the 'researchers' and those being researched. (33) Important to this project, CBPAR acknowledges that "complex social issues often cannot be understood... by 'expert' research" and that "there is value and legitimacy" (p.5, 33) in individuals' and communities' knowledges. CBPAR is also strengths-based and solutions-oriented, therefore it is a well-suited framework for developing and evaluating an Indigenous-led health promotion initiative. (33) Together, CBPAR's characteristics, namely its community-led approach that is guided by community needs and priorities, indicate that it complements and bolsters Indigenous research methodologies. (39,40) As a non-Indigenous scholar working in the Indigenous research space, it was important that I consider my positionality, relational epistemology, accountability to the community, and build my understanding of Indigenous research methods. There has been significant scholarship

exploring the intersections of Indigenous research methodologies and CBPR which has guided me and informed my work with the community.

Relational epistemology

At its core, relational epistemology acknowledges that knowledge is formed through one's relationships with others and the environment throughout their lifetime. This epistemology views "knowledge in relation to knowers" (p.240, 41) and assumes that everyone is a contextual, social being whose perspectives and beliefs are influenced by their physical and social environment. Relational epistemology acknowledges all beings' interconnectedness and honours these relationships by connecting one's knowledge to others and their environment. (41) As relational epistemology utilizes one's relationships throughout their life to contextualize their knowledge, there is no objective knowledge. (41) Rather, relational epistemology embeds one within their understandings of reality and views people as simultaneously impacting their knowledge as they experience it. (42)

Relationality is fundamental to many Indigenous ways of knowing and Indigenous research methodologies. (39,40) Indigenizing research that is reflective of Indigenous epistemologies and is representative of Indigenous communities must be performed within a relational paradigm. (39) Indigenous knowledge must be viewed within its relational surroundings as it is context-specific and based on relationships; this approach to knowledge is a fundamental shift from western science epistemologies that typically utilize a positivist lens, therefore separating knowledge from its relational context. (43) Relational epistemology creates a framework that allows researchers to treat Indigenous knowledges with respect and nuance, as is expected when the knowledge is shared. (39,40) Relational epistemology also provides an Indigenizing approach to learning and knowing that is especially important for non-Indigenous researchers working with Indigenous communities. (44)

In all research methods, the researcher's relationships with the community impact the research process and associated outcomes. Research that uses relational epistemology incorporates relationality from grassroots to knowledge sharing by acknowledging and accounting for these relationships in the literature and findings. (39) Relationships between the researcher and self

and researcher and research are the primary relationships considered in research that utilizes relational epistemology. (6) Utilizing a relational paradigm, my relationships with community members and the collective community existed prior to and I plan to maintain them beyond the research project. (43) When using this approach to knowledge, it is important I reflect upon and acknowledge how my relationships have influenced the research process and findings.

Establishing trusting relationships with community members has enabled this project to exist and will ensure future data generation methods are rigorous and respectful of Inuit knowledge. (39,40,45) When forming these relationships, I have become part of the community's web of relations to which I am now accountable. (45,46) My supervisory committee is composed of an Inuk Elder who is originally from the community and two researchers who have dedicated most of their careers to working with communities in northern Canada. Thus, my relationships with the community and supervisory committee have shaped my understanding of community engagement and are reflected throughout this thesis. As well, my understandings and views on cultural continuity, social connection, and colonization are shaped by my experiences of these phenomena. It is important to note that my perspectives and beliefs have impacted the way I interpret my interactions with the community and that they may view their relationships with me and/or my actions in a different light.

When using a relational epistemology, researchers must also take time to consider their positionality and critically self-reflect on their motivations, goals, and perspectives for the research and their role in the knowledge construction process. Using a relational epistemology lens, one's interests and passions arise from experiences and relationships amassed throughout their lifetime. In addition to the experiences laid out in the positionality section, my parents are both educators and our dinner conversations often centred around ways to foster social and health-promoting behaviours among children and youth in Vancouver's education system. Listening to and participating in these conversations informed my understanding of the intersections of health promotion and education. My family also worked hard to emphasize my privileges and the importance of equity throughout my childhood and youth. From a young age, they encouraged me to volunteer my time with non-governmental organizations (NGOs), including an organization that offered meals to people who are unhoused in Vancouver's Downtown Eastside. My young awareness of my privileges and participation in these activities instilled in me a passion for equity and social justice. These experiences continue to guide my personal ethics and inform my relational approach to and motivations for community engagement and research.

Indigenous research methodologies and methods

Since the early 2000s, there has been a growing body of literature discussing Indigenous research methodologies and methods. (39) Indigenous research methods are not novel, rather their acceptance in Eurocentric academia as rigorous and valid is contemporary. Indigenous research methods are grounded in Indigenous ways of thinking, are actively anticolonial, and centre relationships throughout the research process. (39,40,47) Indigenous research methods are acclaimed not only for the richness of knowledge gained, but for the processes of inquiry that promote community engagement and capacity building. (39,48)

An Indigenous research methodology, the Aajiiqatigiingniq Research Model, underpins the participatory relational work and this research. The distinctions between Indigenous and Western approaches to research require some discussion to highlight Indigenous methods' contributions to further improve Indigenous communities' research quality and sovereignty. (48) Prioritization of relationships is at the core of the differences in the orientation of power and control between Western and Indigenous research methods. Western approaches to research often prioritize the researcher's position of power to the community's detriment, which can result in research othering and supporting negative stereotypes of Indigenous communities and peoples. (39,40) In contrast, participatory¹ and Indigenous research methods are oriented around relationships and aim to privilege and promote Indigenous perspectives, knowledges, and sovereignty. (39,40,49)

Western approaches to research are often damage-centred and lack nuance and community-led solutions. (50) As an alternative to damage-centred research, Tuck suggests desire-based research frameworks that work with communities to celebrate their strengths, while acknowledging their challenges. (50) Indigenous research methods operationalize Indigenous

¹ In this context, I am using participatory to capture CBPAR, CBPR, and PAR methods.

ways of knowing and use a holistic, strengths-based lens to explore research goals, assumed knowledge, and findings, thus emphasizing the community's strengths and opportunities for change. (39,40)

Indigenous research methodologies acknowledge and promote oral knowledge sharing which is integral to many Indigenous communities' cultural and knowledge transmission. (48) While qualitative methods are increasingly being used in Western research, Western inquiry processes and analysis techniques are often inconsistent with Indigenous values. (39,48) Ultimately, Indigenous research methods centre Indigenous communities' needs and priorities, and orient research around positive community change. (39,40,51)

Relational accountability

Relationality and relational accountability are also critical for researchers to apply Indigenous research methods and situate research within an Indigenous sovereignty paradigm. (39) Relationality in Indigenous research methodologies "assumes asking only for what you need, giving back, and taking care of each other, and requires time and knowledge". (p.127, 52) Researchers engaging in relational accountability can be characterized by being accountable to the relationships they form during the research process and engaging in research processes that are led by community needs and desires. (39) To engage in relational accountability, researchers must continually reflect on their value to the research process using relational accountability's central tenets of respect, reciprocity, and responsibility. (6,39) Respect in this context applies to both those involved in the research and the knowledge shared. (6,39) Responsibility refers to researchers' duty to properly care for and distribute knowledge that has been shared with them, while acknowledging knowledge ownership by the person sharing it. (6) Reciprocity is grounded in strong and mutually held relationships with people, knowledge, and land. (6,39)

Critical reflexivity is closely tied to relational accountability. Critical reflexivity encompasses self-reflections of how one's perspectives and goals interact with and influence the research process. (51) Researchers must self-reflect throughout the research process to explore and understand how their experiences, locations, and privileges have influenced their perspectives, and ultimately their constructions of meaning. (6) Researchers wishing to work with Indigenous Peoples through Indigenous research methods must first deconstruct their own privileges and

reflect how these impact the ways in which they interact with the world. (6) This outward introspection involves a rigorous analysis of how and why Eurocentric societies developed and continue to uphold the colonial power systems within which we live, work, and play. (6)

Researchers' representations of stories and knowledge is inherent to all research. When working with Indigenous Peoples, researchers must recognize their involvement in 'the politics of representation'. (40) Throughout media and academia, Indigenous Peoples' representations are rife with negative stereotypes located within damage-centred narratives. (50) Researchers working with Indigenous Peoples must understand and appreciate their role in shaping representations as a political one that holds great power. (6,49) To create positive and empowering representations, researchers working with Indigenous Peoples must commit to respectful representations that: 1) honours the knowledge shared with them, Indigenous epistemologies, and relationships, 2) contextualizes the findings and recognizes the role of colonization and colonizing practices and policies, and 3) celebrates Indigenous Peoples' resistances and ensures knowledge accessibility to the communities they represent. (6)

Inuit Qaujimajatuqangit principles

Inuit Qaujimajatuqangit (IQ) principles translates most directly to "that which Inuit have always known to be true", (p.1, 22) or "... the knowledge, belief system, principles, and values at the core of Inuit identity and that guide/govern Inuit society" (p.413, 53). (21) Inuit Qaujimaningit is a closely related term that refers to Inuit epistemology "without reference to temporality" (54) and can be defined as "local and community based knowledge, ecological knowledge (both traditional and contemporary), which are rooted in the daily life of Inuit" (55). (53) IQ principles are the basis of Inuit culture and inform all aspects of society including wellness, knowledge creation and sharing, family life, and more. (21,22,56,57) IQ principles vary geographically within Inuit Nunangat, but there are common themes and terms used throughout the region. (21,53) These principles and values are passed intergenerationally and were traditionally modelled or learned through experience, rather than taught in a structured lesson format. (11,21,58)

Due to Canada's ongoing colonization of Inuit, there are barriers to IQ principles' traditional transmission. (9,10,12,13,22) To address these challenges, since 1989, there have been continual amendments to the school curriculum delivered in modern-day Nunavut in efforts to expand younger generations' understanding and awareness of IQ principles. (57–59) There has also been considerable work done by the Nunavut government, NGOs, and public institutions to explore ways to re-align organizational structures and values to incorporate IQ principles to better serve Inuit communities. (54,55,57,60,61)

There are hundreds of IQ principles used by Inuit throughout Inuit Nunangat. (21,53) The Nunavut Government, with Elders and other Inuit knowledge holders, has described eight main principles to align government and private sector work with Inuit culture: 1) *Inuuqatigiitsiarniq*, or "Respecting others, relationships and caring for people", 2) *Pijitsirniq*, or "Serving and providing for family and/or community", 3) *Pilimmaksarniq/Pijariuqsarniq*, or Skill development "through observation, mentoring, practice, and effort", 4) *Piliriqatigiinniq/Ikajuqtigiinniq*, or "Working together for a common cause", 5) *Tunnganarniq*, or "Fostering good spirits by being open, welcoming and inclusive", 6) *Aajiiqatigiinniq*, or "Decision making through discussion and consensus", 7) *Qanuqtuurniq*, or "Being innovative and resourceful", and 8) *Avatittinnik Kamatsiarniq*, or "Respect and care for the land, animals and the environment". (p.2, 61) Most often, multiple IQ principles apply to a situation or decision, thus fostering a holistic, critical-thinking-based approach. (62,63)

IQ principles can be viewed as providing a framework in which Inuit communities' values and goals are rooted. While Inuit Nunangat is abundantly diverse, many of the communities' shared strengths can be seen as rooted in IQ principles. For example, sharing food and hunting equipment is common and encouraged in many Inuit communities. (8) This practice relates closely to multiple IQ principles including *Pijitsirniq* and *Piliriqatigiinniq/Ikajuqtigiinniq*. In the context of this thesis, IQ principles are discussed broadly because further community discussions are needed to determine local principles that apply and how best to utilize them.

Thesis framework

I have applied a framework to position the discussion of the early phases of this CBPAR project. I have used an Inuit model to guide the work and facilitate the ongoing community-led development of this research project.

Aajiiqatigiingniq Research Model

Multiple Inuit researchers and communities have called for Inuit research methodologies to "reflect local perspectives and Indigenous ways of knowing in health research". (p.6, 64) The *Aajiiqatigiingniq* research methodology was developed by the Arviat Wellness Society to characterize "research grounded in Inuit worldviews". (65) Aajiiqatigiingniq is an IQ principle that refers to "building agreement together through a group process" (65) and was identified by Inuit Elders as important during decisions that impact collective wellbeing. (66) Employing the IQ principle of *aajiiqatigiingniq* necessitates an inclusive and participatory method built on "trusted relationship[s] with others in the process". (65) Researchers are seen as participants throughout the *aajiiqatigiingniq* research process, thus they must begin the research relationship from a place of humility and respect and commit to supporting solutions that aim to improve the community's collective wellbeing. (45,65) The iterative, solutions-seeking method of aajiiqatigiingniq is "a trusted cultural form of consensus building" (65) and follows a general progression that can be characterized by four phases: 1) meaning making to develop shared understanding of the situation to which a solution is sought, 2) sharing experiences and knowledge to contextualize the topics at hand, 3) developing innovative solutions to address the issue(s), and 4) committing to support the agreed upon "actions to achieve resolution". (65) Everyone involved has equal status and voice when using *aajiiqatigiingniq*, therefore, no one can be viewed as an "objective observer" (65) to the discussion.

The Aajiiqatigiingniq Research Model's process has four stages which relate to other IQ principles: 1) *piliriqatigiingniq*, or building relationships and meaningful community engagement, 2) *inuuqatigiitsiarniq*, or building understanding, 3) *aivaqatigiit* and *uqamangatigiit*, or personal data collection such as sharing experiences and/or knowledge, and 4) *qanurtururangniq*, or validation and relational consensus building. (65)

The *Aajiiqatigiingniq* research methodology's first phase is considered to be the most important for research outcomes as it builds relational accountability and shared understanding among all participants, including the classically-viewed 'researchers'. (45,65) *Piliriqatigiingniq* relationships are grounded in respect, trust, and equitable voice and negotiation power. (45,65) In the research context, these relationships lay the foundation for establishing research goals, processes, and definitions of success that meet all participants' needs and priorities. (45,65)

Inuqatigiitsiarniq, the second stage, develops a shared understanding of research contexts and goals. (65) *Inuqatigiitsiarniq* refers to a "place where iterative discussions" (65) facilitate equitable opportunities for all participants' perspectives to be heard and considered, thus creating relational consensus regarding all aspects of the research process. (45) Discussion topics may include terminology, research priorities and goals, identification of people who hold valuable and relevant knowledge to the research, and agreement on the research process and methods. (45,65) Additionally, community engagement continues to ensure collective awareness and support for the research process. (45,65) This stage aims to raise the community's "critical consciousness" (65), thus establishing meaningful community engagement to ensure rigorous knowledge sharing processes are followed. (45)

In the third stage, participants are able to engage in respectful data generation because shared understandings of research parameters, terms, and goals exist. (65) *Aivaqatigiit* and *uqamanggatigiit* describe ways to "express ideas and engage in heavy discussions" (65) to ensure respectful communication during data generation. Data generation methods that were decided upon during phase two are employed in phase three to create and/or gather knowledge relevant to the community's research goals. Among Inuit, oral communication is considered to be the most respected and trusted form of communication, thus oral data generation strategies, such as storytelling, are expected to be favoured. (39,40,45,64,65)

Data analysis and knowledge sharing occurs in the final phase of the Aajiiqatigiingniq Research Model, using the "IQ process of qanurtururangniq to review, assess, and validate the data collectively." (65) Data analysis, including theme identification and contextualization, occurs through iterative and collective "consensus building process[es]". (65) Once agreement has been reached, collaborative processes are employed to discuss how to apply this knowledge. (45,65) This action-oriented phase is guided by the IQ principle *isumaliuqatijiitsinirningniq* which refers to "a way of arriving at collective decisions that can be used to improve the common good." (65) Sharing knowledge with the greater community, including research findings and action plans, is a fundamental component of this phase and ensures research aligns with Inuit values. (39,45,65)

To date, my work with the community can be situated within phases one and two as laid out by the Aajiiqatigiingniq Research Model. Primarily, my work with the community has been centred around building trusting relationships and relational accountability with individual community members and the collective community; examples of strategies I have used include respectful conversations centring around my and the community's research intentions and goals and securing community-held funding for the project. I have also conducted a thorough literature review (see Chapter two) based on topics that were identified during conversations I had with community members; the knowledge I gained during these conversations and the literature review will guide my involvement in the second phase of the research model.

CBPAR Methods

During initial discussions with community members, community direction of all research activities was necessitated to ensure research is culturally competent and addresses community priorities. The community's caution when interacting with researchers and agreeing to research projects is a response to harms the community has previously experienced in the name of research. Due to this project's intersectoral and solutions-oriented approach, we decided to employ CBPAR methods in conjunction with the Aajiiqatigiingniq Research Model. CBPAR methods are grounded in CBPAR methodology which is defined as:

...a collaborative approach to research [that] involves all stakeholders throughout the research process, from establishing the research question, to developing data collection tools, to analysis and dissemination of findings. (p.5, 33)

CBPAR is an approach to research that results in community-led and -owned research. This research framework acknowledges experiential knowledge's value and place in health promotion initiatives. When using CBPAR methods, experiential knowledge is crucial not only for data generation purposes, but also when planning research activities, analyzing findings, and

developing action plans. This research approach aims to incorporate multiple perspectives and ways of knowing during planning phases, to increase the research process' accessibility and safety for all. CBPAR is solutions-oriented and "recogniz[es] that social action requires further research and social research requires further action". (p.5, 33) CBPAR can begin with either a desire for action or research, but these processes are inextricably linked, thus creating an iterative process of research and action as presented in Figure 3. (33)



Figure 3. Iterative process of research and action in CBPAR methods.

Note: These images were created to show CBPR's iterative process of research and action. (p.85,97, 67 as cited in p.5, 33)

Intersections of CBPAR Methods and Aajiiqatigiingniq Research Model

This thesis is informed by two overlapping and aligning research methodologies and methods: 1) the Aajiiqatigiingniq Research Model and 2) CBPAR methods. These approaches to research complement each other as they emphasize community-led action throughout the research process and require an inclusive, participatory, and iterative approach to research. (33,65) Additionally, among Inuit,

...knowledge can only be described as such if it is used to improve the lives of others. If one has 'knowledge' but does not share it or use it for the common good, then it is seen to have no value. (65)

This cultural understanding of knowledge's role in relational accountability aligns well with CBPAR methods which emphasize solutions-oriented action. Both the Aajiiqatigiingniq Research Model and CBPAR methodology are oriented around action that improves community wellbeing, particularly for those who participated in the knowledge generation and/or sharing process. (33,65)

Thesis objectives

CBPAR methods result in findings that are meaningful for the community as all research activities are driven by community needs and priorities. Building community relationships and fostering community engagement with a diverse group of community members is fundamental to CBPAR processes and necessitates a significant time investment before data generation planning or activities can begin. Currently, the vast majority of participatory research literature focuses on research outputs or end results. More research is needed to understand the early stages of CBPAR research processes. This thesis is a response to this gap in the literature and discusses the first two phases in the Aajiiqatigiingniq Research Model that focus on engaging the community by forming relationships and building mutual understanding. More specifically, this thesis:

- A) Provides an in-depth narrative and critical reflection of my experiences during Aajiiqatigiingniq Research Model's *piliriqatigiingniq* and *inuuqatigiitsiarniq* stages;
- B) Characterizes logistical requirements needed to advance research projects during the early stages of a CBPAR project;
- C) Describes practice recommendations for researchers wishing to build community relationships and foster community engagement in a research context; and
- D) Identifies policy recommendations for research and funding institutions to encourage and enable CBPAR methods.

Locating research within a larger project

This thesis explores the initial stages of the Aajiiqatigiingniq Research Model, 1) *piliriqatigiingniq*, and 2) *inuuqatigiitsiarniq*, or building relationships and understanding, respectively. This discussion occurs in the context of planning a community-led, health promotion project among youth. These are the first steps in a project that hopes to explore and implement initiatives for youth that will instill capacity-building traits, including resilience and problem-solving, through activities that encourage cultural continuity and social connection. Examples of activities include learning on-the-land skills such as hunting and navigation, country food preparation, and Inuit art activities. This larger project is driven by a desire to foster positive mental health among the community's youth. Specifically, the project aims to instill and promote cultural pride, healthy relationships with substances, and positive interactions with law enforcement. See Figure 4 for a visual representation of this larger project's process and goals.



Figure 4. Research and health promotion project framework. *Note*: Image inspired by conceptual model presented in (68).

This larger project framing was coordinated by myself, a community municipal Councillor, and an Inuk Elder from the community. A research committee composed of a diverse group of community members will be formed to direct all project activities in the next phases.

Currently, there are seven overarching phases planned for this health promotion project:

- Foster meaningful community engagement by building relationships and shared understanding;
- 2) Form the research committee and collaboratively discuss roles and responsibilities to create a shared understanding of work distribution and timelines;
- a) Explore how to safely conduct research with the community to maximize benefit and minimize potential harms, and

b) Plan pilot activities for youth that aim to foster cultural continuity and social connection;

- Characterize how youth are currently engaging in: a) cultural continuity and social connection, and b) substance use;
- 5) Qualitatively explore how youth would like to engage in cultural continuity and social connection (i.e., what activities or initiatives would they be interested in?);
- 6) Implement pilot projects grounded in IQ principles that serve the needs and interests of youth with ongoing activity evaluation by youth; and
- Re-characterize how youth engage in: a) cultural continuity and social connection, and b) substance use.

This research thesis discusses my experiences during the project's first phase which aligns with the Aajiiqatigiingniq Research Model's first two steps. As noted by the Arviat Wellness Society, relationship building "is a process that cannot be overlooked or rushed through". (65) Building strong relationships with community members will ensure this research project effectively engages in CBPAR going forward and is grounded in Inuit culture. This initial phase of the project has three goals: 1) form relationships with a diverse group of community members to ensure relational accountability, 2) informally learn about community members' interests, needs, and priorities, or "explor[e] contexts ... [and]... seek[] shared understandings/starting points" (65) and 3) gain support from Council-identified stakeholders, including the community's

mayor, mental health nurse, elementary and secondary school principals, and the education board, or "community outreach/awareness". (65)

Relationships formed during the first phase will guide the research committee composition as we will employ snowball sampling. Phase three is essential to ensure research processes are safe and that the necessary supports and safeguards are in place for participants experiencing distress. Knowledge created during the fourth phase will inform qualitative topics explored during the fifth phase, as well as activity delivery and type during phase six. Phase four results will also provide the baseline data with which we can compare phase seven results to evaluate the project.

Rationale for larger project and study

The community with whom I am working is a young, resilient, and resourceful community whose members are committed to finding solutions to challenges they are experiencing. In 2018, a community needs assessment was conducted during which community members voiced concerns over challenges they experience related to structural violence, including insufficient housing and food insecurity. Addressing these challenges forms the larger project's foundation. As many of these challenges can be linked to historical and current colonization policies that disrupt Inuit ways of life, the larger project will implement programs that aim to address the effects of colonization, by fostering cultural continuity and social connection among youth. (11) Further, community members, particularly Elders, have long called for initiatives that promote intergenerational knowledge sharing between Elders and youth. Therefore, this project's community-led, upstream, strengths-based approach is rooted in Inuit cultural values and IQ principles and aims to facilitate appealing and accessible cultural and social programming for youth.

During initial discussions with the community, it was made clear that total community direction and ownership over the research process is critical. Various Councillors have expressed a desire for the project and support the project's goals, but the collective Council has necessitated various community stakeholders' support before proceeding to phase two. The Council also emphasized the importance of thoroughly planning the research committees' composition, roles, and responsibilities. Thus, this project demonstrates a CBPAR approach as community members have been meaningfully engaged from grassroots and direct all research and program activities.

22

Conscientious planning of the research committee will ensure purposeful engagement with and direction from a diverse group of community members throughout the research process. As noted previously, building relationships with the community is an essential first step to ensure the subsequent phases of this larger project are conducted in ways that are culturally appropriate and maximize positive impact while minimizing any potential harms.

In this chapter, I have introduced Indigenous research paradigms, CBPAR methodology and methods, the community's historical and cultural context, and situated this thesis within a larger health promotion project.

Chapter 2: Literature Review

This chapter highlights the findings of literature review which provides a critical assessment of the literature pertaining to community engagement during CBPAR. I provide an overview of definitions and discourses, Indigenization and decolonization, Inuit research paradigms, CBPAR methods, guidelines for researchers wishing to work with Inuit communities, building community relationships, cultural continuity and social connection among youth and strategies to foster these health-promoting characteristics among Inuit youth, colonization in Inuit Nunangat, and CBPAR as a health promoting initiative itself. The literature review's findings capture the depth of scholarship and Indigenous knowledges guiding participatory research with Inuit communities. Figure 5 illustrates the interconnected nature of this literature review's discussed phenomena.



Figure 5. The web-like nature of the literature review.

Definitions and discourses

It is important to define terms within spatial and temporal contexts as definitions are shaped by these characteristics. (6) Indigenous research paradigms are inherently defined by the community employing them, thus, defining an Inuit research methodology is essential to understand this thesis' approach. (39,40) CBPAR methods are often used synonymously with other qualitative methods; therefore, a definition of CBPAR in this context is necessary to understanding this research project's methods. Cultural continuity is a relatively new term in Indigenizing literature that is intrinsically defined by the people group using it. (69,70) Within Canada, the mechanisms and ramifications of colonization vary geographically and with people group. Additionally, social connection and substance use are terms that have evolved with time and reflect changes in our understanding of human psychology. The aforementioned terms are highly context-specific and vary temporally, spatially, and culturally. Thus, defining these terms' applications in this work is crucial to contextualizing my experiences beginning a CBPAR project with an Inuit community.

Discourses allow us to understand current social processes or phenomena, in the context of historical practices or circumstances and their public discussions. (6,71) Indigenous research paradigms reflect the great diversity seen between Indigenous People groups; thus, further discussion of this thesis' Inuit research methodology is needed. CBPAR methods vary greatly between applications, therefore a discussion of this thesis' approach to community engagement, funding, ownership over research, and next steps is required. Cultural continuity and social connection are dynamic, complicated processes and phenomena that have evolved significantly over time and vary widely between and within people groups. The discourses surrounding these processes and phenomena reflect their dynamic and context-specific nature. (72–75) The lens with which colonization and its ramifications are viewed is embedded in cultural values and norms. Discourses surrounding this oppression structure have changed significantly in the last century, with current academia focusing on ways to enable and promote decolonization and Indigenous Peoples' self-determination. Collectively, these terms' nuanced meanings and applications highlight the need for a context-specific review of their discourses to understand this work's context, findings, and implications.

Indigenization and Decolonization

In academia and wider society, there is great discussion over the use of the terms Indigenization and decolonization. While these terms have varied definitions, Indigenization within research is generally described as research methodologies and methods that promote the culture, values, knowledges, and goals of the Indigenous Peoples with whom the research is being conducted and "[move] beyond tokenistic gestures of recognition or inclusion to meaningfully change practices and structures". (76) Indigenizing processes "should be led by Indigenous Peoples and should respect Indigenous intellectual sovereignty". (77) Indigenization includes work such as developing and utilizing Indigenous research methodologies, methods, and data generation strategies. (77,78) On the other hand, decolonization processes divest colonial power from "bureaucratic, cultural, linguistic, and psychological" (p.101, 79) spheres to "[bring] about the repatriation of Indigenous land and life". (p.1, 80) In the Inuit context, decolonizing initiatives have been described as "action oriented solutions [that] empower Inuit community members to engage in problem solving, create relationships, and honour community knowledge". (p.9, 81) Decolonizing research work includes research methods that actively further Indigenous sovereignty by promoting research capacity building and knowledge sharing that "decentre[s] ... Euro-Western structures of knowledge" (77) and results in increased self-determination. (79) Thus, while distinct processes and phenomena, decolonization and Indigenization are complementary. There is currently a debate in academia over whether Eurocentric institutions aiming to 'Indigenize the academy' (2), must first decolonize by challenging and transforming the "ways in which they reproduce settler colonial power". (77)

Recently, decolonization has become a 'buzz word' in academia and larger society, with people using the term haphazardly, thus neglecting its true meaning. As asserted by Tuck and Yang, "the metaphorization of decolonization makes possible a set of evasions ... that problematically attempt to reconcile settler guilt and complicity, and rescue settler futurity." (p.1, 79) Decolonization only describes work that dismantles colonial structures and restores Indigenous Peoples' self-determination over their land, leadership, and/or decision-making, thus researchers must be careful when applying the term to ensure its specificity, power, and urgency are preserved. (2,77,79) Non-Indigenous people have an important role to play in decolonizing and Indigenizing work by simultaneously "identifying, challenging, and dismantling settler colonial
privileges and structures" (p.358, 82) and stepping aside to allow Indigenous Peoples to lead these movements. Thus, researchers who engage in decolonizing work with Inuit must 'tread lightly' to ensure they are acting in ways that affirm and align with "community leadership, structures, values and beliefs". (p.9, 81)

Throughout this thesis, I respectfully use the term Indigenization to reflect the current project's work and acknowledge that I have applied Indigenous knowledges with the intent of amplifying their presence within academia. Using the term decolonization to describe this thesis' work would be an act of metaphorization as I focus on my experience, as a non-Indigenous graduate student, building a CBPAR project with an Inuit community. Additionally, the project is in its early stages of capacity building to achieve research self-determination and has not yet dismantled any colonial structures. However, as necessitated by many community members, this project's goalposts are decolonization and research sovereignty, thus the language with which we describe the project will continue to evolve.

Inuit research paradigms, methodology, and methods

Research paradigms can be defined as "the beliefs that guide our actions as researchers." (p.13, 39) A research paradigm includes the researcher's ontology, epistemology, axiology, and methodology. (38,39) These facets of research together guide the research goals, processes, and outcomes. Briefly, ontology examines what we can know or understand about the world, epistemology explores how we have gained and currently view knowledge, axiology scrutinizes our morals and ethics, and methodology studies research methods or how research produces or gathers knowledge. (38,39) Thus, methodology is just one component of the research paradigm and it must align with the project's ontology, epistemology, and axiology. (38) Research methods, while often used synonymously with methodology, are located *within* their respective methodology, and are defined as the "collection of research strategies and techniques based on theoretical assumptions that combine to form a particular approach to data and mode of analysis". (p.2, 83 as cited in p.31, 38) While I have been privileged to engage in informal discussions with community members about Inuit ontology, epistemology, and axiology, I am not Inuit myself, thus it would be inaccurate and an act of cultural appropriation to reflect on

these aspects of Inuit research paradigms. Thus, this thesis focuses on my experience employing a published Inuit research methodology.

An Inuit research methodology is described by Healey and Tagak as an approach to research that acknowledges and supports the relationships that form "the foundation from which we [Nunavummiut] move forward to achieve wellness" (p.11, 45) by being inclusive, participatory, and action-oriented. (65) Inuit research methodology is grounded in a relational epistemology which describes knowledge as originating in "people's histories, stories, observations of the environment, visions and spiritual insights." (p.3, 45) Inuit epistemology is holistic, thus research using Inuit methodology encourages multi-sectoral collaboration to develop solutions. (45,65)

Prioritizing relationships is critical to meaningfully engage and apply an Inuit methodology. (45,65) Accountability to these relationships lays the methodology's foundation by ensuring respectful participant engagement, research ethics, data generation and analysis strategies, and knowledge sharing. (39,40,45,65) Researcher characteristics that are important to engaging in relational accountability include, "responsiveness and openness... methodological coherence, and reflection upon intentions and relationships" (p.10, 45). (65)

In practice, relational accountability influences the research process because one who is engaged in relational research must consider and act in ways that strengthen their relationships. (45,65) Wilson (2008) notes that relational accountability impacts their work in four, interconnected ways: 1) research question formation, 2) data generation strategies, 3) data analysis, and 4) knowledge sharing. (39) For example, community relationships provide researchers with a nuanced understanding of the community's histories and current realities to allow them to meaningfully contribute to the collaborative question formation process. (45) Healey and Tagak also discuss an example of how relationships can impact participant engagement and recruitment, "snowball engagement method[s]... [focus] on the establishment of trusting relationships... which strengthens the response to the project, as well as contributes to greater rigour and accountability overall". (p.6, 45)

Researchers engaging in collaborative discussions with communities around data generation must reflect on their relationships and their knowledge to determine which data generation techniques to propose that will result in meaningful outcomes and be respectful of the community's needs and preferences. Oral history and culture are important aspects of Inuit ways of being and knowing. (45,65) Data generation methods which encourage storytelling and sharing narratives allow Inuit to engage in research in ways that respect cultural values and norms. (45,65) Data generation strategies which facilitate storytelling are "grounded in [an] Inuit epistemology" (p.6, 64) by enabling multiple truths and reflect Inuit societal values by ensuring the researcher and community members are engaged as partners. In relational epistemologies, stories are offered in the spirit of sharing, rather than giving. (45) In practice, this means that data generation may resemble a conversation where the researcher first shares their experiences and available resources related to the topic, while creating space for the participants to share their stories. (45) The language in which knowledge is shared or gained influences its interpretation; thus, research processes, especially data generation and knowledge sharing, provide opportunities to further strengthen relationships by respectfully communicating through the local Inuktut language. (45)

Consistent with relationality, it is important for the researcher engaging in data generation to reflect upon and acknowledge their role in shaping the conversation to determine the stories' meaning and intent. (45) When analyzing the knowledge and experiences shared, researchers must contextualize findings and align them within an Inuit epistemology; concurrent member checking by engaging those who participated in the data generation process during data analysis is also essential to be relationally accountable to the community. (45) It is important that researchers allow themselves sufficient time to immerse themselves in the knowledge that has been shared with them and allow ideas to organically form. (45) From a relationality perspective, discussing proposed findings' validity and potential with those who hold relevant knowledge, including community members and academic colleagues, and situating these ideas within a literature review, is fundamental to rigorous data analysis. (45)

Relationships also play an important role in knowledge sharing. When sharing others' stories, researchers must ensure the knowledge holder consents to the story being shared on their behalf, take care to honour the holder's intent when it was shared, and give credit, if desired. (45) When stories are shared with researchers, they are shared in the spirit of relationality; this means that the knowledge contained within the stories and the narrative are meant to co-exist to allow

individuals to derive relevant lessons for themselves. (45) However, in Western academia, qualitative data is typically grouped thematically and separated from its context as there is limited textual space in academic publications or presentations. (45) In alignment with Inuit methodology, when sections of stories must be omitted due to imposed constraints, researchers must seek approval from the story holder and could honour the omitted sections by acknowledging their existence. (45)

As discussed above, sharing and applying knowledge to improve the community's collective wellbeing is fundamental to Inuit research methodology. (45,65) This approach to knowledge sharing is grounded in relational accountability as that knowledge was originally shared with the mutual understanding that the research will benefit the community. Prior to knowledge sharing, a diverse group of community members should be engaged to decide upon the content and method of knowledge sharing to ensure it aligns with community needs and goals. Efforts must be made to make knowledge sharing formats accessible to all community members, particularly those who participated in the research process, by translating all materials, using plain language and visuals throughout, and hosting oral presentations with an interpreter present.

In Inuit research methodology, research ethics are grounded in relational accountability. (45,65) Inuit values are unique from Eurocentric ones; non-Inuit researchers working with Inuit communities must engage a diverse group of community members to ensure research processes align with community values. (45,65) The ways in which Inuit values direct the research process varies between projects and depends upon community needs and priorities. For example, the cultural emphasis placed on collective wellbeing could prioritize community-level knowledge or solutions over individual-oriented action. (61,62) Consent processes must also act from a place of strengthening relationships by discussing formal consent procedures and offering options that respect oral communication and language preferences. Similarly, confidentiality must utilize a relational approach by ensuring safeguards are in place that meet the community's needs. (45) These practices should be collaboratively developed with the community to ensure the research process is accessible and safe for all community members.

Community-based participatory action research methods

CBPAR and other participatory research methods' popularity have been increasing in recent years, especially among university-based researchers working with Indigenous Peoples. (84) Similar to Indigenous research methods, participatory research methods are not "novel approach[es] to research outside the academy" and have been used since the 1970s "as an intervention for positive social change". (p.162, 84) CBPAR methods are rooted in a desire to work towards change the community views as beneficial, while addressing social inequities and power imbalances that manifest in conventional research approaches. (84)

There is no unanimous definition of CBPAR. In this thesis, CBPAR is defined as an approach to research wherein a diverse group of community members direct all research activities, including research question formation, research methods, and knowledge translation strategies, and research is solutions-oriented to benefit the community. (33,35,85–87) As community members are both conducting the research and the ones being 'researched', the line between the 'researchers' and 'researched' is blurred. (33,35,85-87) One can also view the distinction between 'researchers' and community members as sitting on a "participation continuum" (p.7, 33) where traditional Western research methods have less community participation and researchers and participants are distinct from one another, and CBPAR methods have greater community participation and there is minimal distinction between 'researchers' and 'participants'. CBPAR aims to not only validate experiential knowledge that is often not recognized in Eurocentric academia, but also build research capacity within the community. (33,35) As CBPAR is a method or "collection of research strategies and techniques based on theoretical assumptions that combine to form a particular approach to data and mode of analysis" (p.31, 38), various qualitative data collection strategies can be used. (4,83) For example, CBPAR can be utilized with focus group, photovoice, or participatory film strategies. (38)

Distinctions between CBPAR, CBPR, and PAR are unclear in the literature and these terms are often used interchangeably. (38) These terms' use varies based on discipline, with CBPR and PAR often describing indistinguishable approaches to public health and education work, respectively. (34) In this thesis, CBPAR is used to respectfully acknowledge calls for research with Inuit communities to be community-based, participatory, *and* action-oriented, this project's

interdisciplinary nature and intent to build research capacity, and minor differences in these terms' definitions. CBPR can be defined as

... an approach to research that involves collective, reflective, and systematic inquiry in which researchers and community stakeholders engage as equal partners in all steps of the research process with the goals of educating, improving practice, or bringing about social change. (p.2, 35)

This work utilizes a CBPAR framework over CBPR because this work is solutions-oriented and all research activities are oriented around a goal to positively change the community. (88) PAR can be defined as a form of qualitative research where researchers collaboratively work with the 'researched' to understand human experiences, and take constructive action to make positive change. (85) This work utilizes a CBPAR framework over PAR because CBPAR is "grounded in the needs, issues, concerns, and strategies of communities". (p.5, 33) While some view CBPR and PAR as inherently solutions-oriented and community-based, respectively, this work utilizes CBPAR to emphasize the critical roles of both action-oriented methods and Inuit knowledges and perspectives throughout research planning, implementation, and follow-up phases. (88) However, in this thesis, I collectively refer to community-based (CB) and participatory action (PA) research as 'CB and/or PA research' to reflect their similar meanings and practical implications.

There are three main strengths of CBPAR methods. (33) Firstly, in CBPAR, research is community-led and -owned which facilitates: 1) research that addresses needs prioritized by the community, 2) strategies and solutions that are sustainable and align well with the community's cultural, physical, and social context, and 3) a research process that fosters community research capacity growth. (33,35,88) Secondly, relationships with a diverse group of community members ensure varied perspectives are considered during the planning and implementation phases to increase equitable opportunities to participate in the research process. (33,38,88) This not only increases the equity of the research process, but strengthens the research's rigor as a more representative group of people will be able to participate. (33,88) Lastly, CBPAR is solutions-oriented and utilizes a desire-based approach, thus providing the community with ways forward from challenges they are currently experiencing. (33,38,88) Not only does this facilitate research

that aligns with an Inuit epistemology, but it strengthens and reasserts community sovereignty over health promotion initiatives. Tuck emphasizes the important intersections of sovereignty and solutions-oriented work by stating, "a framework that accounts for and forwards our [Indigenous Peoples'] sovereignty is vital." (p.423, 50)

Challenges researchers experience when engaging in CBPAR are largely grounded in academia's and other institutions' characteristics and requirements. (88) While many institutions have begun to take steps to ameliorate these challenges for researchers and communities, some barriers persist. For example, research ethics boards (REBs) typically require researchers to specify or outline all research activities from the outset; this requirement is not amenable to CBPAR's iterative research and action phases that develop organically based on knowledge shared and gained. (38,89) While many REBs allow researchers to update their methods throughout the research process, this can be highly time-consuming and therefore impractical or inaccessible for researchers and/or communities. (89,90) Further, many REBs' standards and guidelines are not appropriate for CB and/or PA research methods, especially for research with Indigenous communities, as they focus on individual wellbeing over collective. (89,91,92) Additionally, CB and/or PA research requires a lengthy time investment to foster trusting, long-term relationships between the institution-based researchers and community and to develop sustainable solutions based on findings. (87,93) Researchers often receive pressure from their affiliated institution(s) to publish research results regularly which can be challenging given CB and/or PA research project timelines. (84,93) As well, research grants are often short-term, thus creating barriers for researchers and communities wishing to engage in CB and/or PA research methods. (93–95) When funding is attained, these short-term grants can undermine the sustainable and long-term nature of CB and/or PA research work. (93-95)

Suggested strategies to ensure REBs are appropriate tools to evaluate CB and/or PA research methods are grounded in ethical principles of respect, beneficence, and justice. (92) Such recommendations to REBs include utilizing a flexible and holistic definition of authorship so that communities are recognized in academia, requiring a community-created collective agreement that describes the research process and all potential impacts prior to REB approval, and ensuring community rights to and ownership of knowledge are acknowledged. (92) There is also a push for REBs to move beyond the individual-centred risk analysis and consider the rights of and risks

to the community during ethics review. (96) More research into the experiences of communities and institution-affiliated researchers engaging in CB and/or PA research is needed to effectively address these challenges and increase access to these valuable research methods and frameworks. Additional challenges associated with the REB process in the context of CB and/or PA research with Indigenous communities are discussed in the following section.

Guidelines for researchers wishing to work with Inuit communities on health-related topics

Worldwide, but especially in communities experiencing inequities, research is needed to support evidence-based policies that will "support optimal outcomes" and inform "actions that create safer, healthier, and more resilient communities". (p.3, 97) However, research with Indigenous communities that utilizes colonial approaches and/or methods undermines research's potential and causes harm. (97) Now, more than ever, "research governance bodies, policies, and practices must be transformed to respect Inuit self-determination in Inuit Nunangat". (p.4, 97) These calls for collective, community-based Inuit research leadership aim to maximize research impact and limit research-related harm(s). (97) As asserted by the Inuit Tapiriit Kanatami (ITK), research with Inuit communities must function as a tool to achieve sovereignty and empower community members to seek solutions to meet the community's needs and priorities. (97)

Currently, there are fiscal and human resource limitations preventing research sovereignty among Inuit. Federal research funding must be amended to allow Inuit representational organizations to access funding as lead institutions or principal investigators. (97,98) To address the human resource limitations, Eegeesiak, former president of the Qikiqtani Inuit Association, "recommend[s] that the governments of Canada and Nunavut provide training and other support that will allow Inuit to actively participate in Arctic ... studies and activities." (p.80, 9) Inuit communities who operate community-based research centres can be exemplified for other communities wishing to scale-up their research capacity. (64) Additionally, as affirmed by the ITK and the Murdered and Missing Indigenous Women and Girls' (MMIWG) Call for Justice 16.26, a university in Inuit Nunangat is needed to build research capacity among Inuit which will ultimately address the "continued domination of Inuit Nunangat research by non-Inuit researchers based outside of Inuit Nunangat" (p.4, 97). (99) Indigenous rights movements across Canada have repeatedly called for resources to develop research that is respectful of community needs, priorities, and epistemologies. The Truth and Reconciliation Commission's (TRC) Calls to Action 58, 65, and 78 highlight research's critical role to further reconciliation and provide opportunities for communities to share their truth, healing, and reconciliation journeys. (100) The MMIWG's Call for Justice 5.25, "call[s] upon all governments to resource research on men who conduct violence against Indigenous women, girls, and 2SLGBTQQIA people." (p.186, 99) The Qikiqtani Truth Commission recommends that the "Qikiqtani Inuit Association should develop a framework (principles, policies, and techniques) for all private, public, and research agencies to use in conducting consultations with Inuit" (p.67, 9) to further strengthen Inuit governance.

Summary of guidelines

To address the "growing recognition of unethical research practices involving Indigenous Peoples in Canada" (p.163, 84), multiple Indigenous organizations and governmental bodies began publishing research guidelines and policies in the 1980s. In this section, I will synthesize and analyze ten organizations' published guidelines for research-related policy change and researchers working with communities in Inuit Nunangat. These organizations include: ITK, Qaujigiartiit Health Resource Centre (QHRC), Nunavut Research Institute (NRI), TRC, MMIWG Report, QTC, Association of Canadian Universities for Northern Studies (ACUNS), Canadian Federal Panel on Research Ethics, Canadian Institute of Health Research (CIHR), and Social Sciences and Human Research Council (SSHRC). These research guidelines and policy recommendations can be viewed along an impact scale from institutional policy to individual researchers working with communities. For the purposes of this thesis, I will focus the discussion on recommendations for individual researchers working with Inuit communities. The ITK's five priority areas are used as a framework to analyze and synthesize the other guidelines: 1) research sovereignty, 2) research ethics grounded in Inuit ways of knowing and being, 3) financial control, 4) community-held control of knowledge, and 5) research capacity building. (97)

Research sovereignty

To promote Inuit self-determination in research, Inuit and their representational organizations must be "engaged as partners" (p.11, 97) when determining research project timelines, focus, methods, data analysis strategies, and knowledge sharing. (9,98,101–106) Article 9.6 of the Tri-

Council Policy Statement 2 (TCPS 2) highlights the importance of engaging "all relevant sectors – including individuals and subgroups who may not have a voice in formal leadership" (103) when building community partnerships. As all other research recommendations can be seen as tools to further Inuit sovereignty, researchers must reflect on their role in and actions to support Inuit research self-determination. (97)

Participatory methods grounded in trusting and respectful relationships were highlighted by most of the guidelines as fundamental characteristics of research that aims to advance Inuit research sovereignty. (9,97,98,101–103,105,106) Many of these guidelines highlight the importance of relationship building as an important first step when beginning a research partnership with an Inuit community. (98,100,102,103) When building relationships with communities, it is important that researchers critically reflect on their understanding of and commitment to relational accountability as these relationships are often the primary mechanism communities employ to protect their interests and centre their needs throughout the research process. The NRI and TCPS 2 acknowledge that not all research projects will inspire the same degree of community engagement, but all projects can be initiated from a place of respectful relationship building. (98,103) A collaboratively-drafted research agreement between all parties has been suggested as a way to ensure respectful and reciprocal research is planned and delivered and that the research process strengthens community sovereignty. (98,102,103,105)

The importance of contextualizing findings within Inuit epistemologies is another crucial way for researchers to foster research sovereignty. (9,102,104–106) At a workshop with Inuit from across Nunavut, the importance of respectfully including Elders and others who hold Inuit knowledges in research processes was emphasized. (101) However, plans to incorporate Inuit knowledge must be accompanied by rigorous planning and resource commitment as, "token references to traditional knowledge are viewed as insincere attempts by researchers to leverage funding or fulfill requirements." (p.4, 98) (103,105) Strategies researchers can use to ensure they are respectfully applying the Inuit knowledge shared include thorough member checking, encouraging the knowledge holders to participate in the research and/or action planning process, and seeking consent and feedback prior to any knowledge mobilization or action phases.

Research ethics grounded in Inuit ways of knowing and being

Inuit values are located within a relational paradigm, thus researchers working with Inuit communities can consider research ethics as manifestations of relational accountability. (39,102) As Inuit consider themselves as being in relationships with other people, wildlife, and the environment, researchers must also be accountable to these relationships to ensure ethical research with Inuit. (97) Additionally, confidentiality and consent processes must be collaboratively determined with community members to ensure they align with Inuit values; special consideration must be given to ensure individuals' confidentiality in small, densely-connected communities. (103,105–107) Consistent with Inuit interpretations of knowledge and reciprocity, research must be action-oriented and seek to improve the collective wellbeing. (65,98,101,105,106)

Guidelines based on discussions with Inuit community members from across Inuit Nunangat affirmed the importance of equitable "treatment of researchers and community knowledge holders" (p.5, 101) throughout the research process and that research ethics must incorporate Inuit values. Researchers must also ensure that community members who shared their time, skills, and/or knowledge are recognized as authors and/or contributors, if they desire, in all knowledge sharing products. (98,103–106)

Financial control

Access to funding and financial control is a keystone element to furthering research sovereignty and capacity building among Inuit. (97) While policy changes by funding institutions are the primary means to increase financial control, researchers wishing to begin a research relationship with an Inuit community could discuss adding the respective Inuit representational organization to grant applications to facilitate funding control. (108) Additionally, researchers should "assist community members in pursuing and securing research funding, wherever possible". (p.5, 98)

Appropriate compensation given community members' time, skills, and/or knowledge shared is an essential piece of ethical research and one that must be determined during project planning with the community. (98,102) While under-compensation is the more-frequent concern, overcompensation can be argued as equally problematic because it could function as a form of coercion, thus encouraging community members to participate when they otherwise would not have. (102,107)

Community-held control of knowledge

Most guidelines highlighted the importance of supporting and encouraging community-held control of knowledge and research data. (98,101–103,105) Eurocentric academia instills and promotes researcher-favoured proprietary structures. Therefore, researchers, especially those who are experienced in traditional Western research methods, must critically reflect on their biases and the implications of community-held knowledge control and ownership. In practice, community-held control results in community members having access to and authority over research data, final decision-making power in collaborative discussions around knowledge sharing, and long-term data storage in communities. (98,103,105,106) As participatory research methods engage community members throughout the research process, they facilitate the community's retention of control and ownership over their knowledge. (101,102) Explicit parameters around the researcher's rights to and usage of knowledge created must be stated in the research agreement to ensure the community agrees to this power distribution. (98,103,105,107) Additionally, biological samples from humans and other organisms may be seen as sacred within some communities; explicit parameters around sample extraction, handling, usage, and long-term possession must be discussed with the community. (105)

Integrated knowledge sharing is encouraged to facilitate a transparent research process and enable larger community engagement with the project. (98,105) Researchers must ensure that all knowledge sharing products are collaboratively created and/or provided in a timely manner to the community in "useful and understandable" (p.2, 101) formats, including oral presentations and visual and textual materials in the local Inuktut language. (98,102) Researchers must also ensure Inuit-appointed entities own all knowledge and research products. (97,98,105) To affirm the community's control and ownership during knowledge sharing with groups from outside the community (e.g., academic journals or presentations), researchers must work collaboratively with community members to determine the material's content, format, and organization or institution with whom the knowledge will be shared. (98,102,103,105,106)

Research capacity building

Research processes that build Inuit research capacity are needed to further research sovereignty within communities in Inuit Nunangat. (9,97–99,103,105) Capacity building also promotes reciprocal relationships between institution-based researchers and communities. Examples of ways researchers can foster research capacity include learning about and respectfully utilizing Inuit research methodologies to amplify their presence in academia and meaningfully engaging community members, especially youth, throughout the research process to learn about and foster an appreciation for research. (97,101,109) Hiring local research assistants and giving credit for their contributions is another strategy institution-affiliated researchers can employ to build Inuit research capacity and invest in the local economy. (101)

Critiques of the Tri-Council Policy Statement 2

While the TCPS 2 was created "in the spirit of respect" (103) to promote ethical and respectful research with Indigenous communities in Canada, the ITK discusses three main critiques of this document when applying it in the Inuit research context. (97) Firstly, REB review is the primary mechanism to ensure researchers are compliant with the TCPS 2, "making it a relatively inefficient mechanism for facilitating Inuit self-determination in research." (p.24, 97) While communities can report research misconduct to the NRI and/or affiliated REB, this would require community members having a detailed understanding of research ethics and act in a way that may go against their views of relational ethics. Issues around REBs' role in TCPS-2 application and regulating mechanisms lays the foundation for the ITK's subsequent concerns. REB review processes typically have a vested interest in the research proceeding, thus their review and consideration for TCPS 2 policies may be undermined by biases in favour of the researcher or research project. (97) Lastly, TCPS 2 applies only to "institutions and entities that receive federal funding from the Tri-Council Agencies" (p.24, 97), most of which are located outside of Inuit Nunangat; thus, research risk is typically evaluated by people who are not Inuit and do not consider the broader definitions of Inuit wellbeing.

These concerns, among others, are echoed in the literature. Many researchers have expressed frustration and concern with the REB process as reviewers often do not "fully appreciate the tenets of community-based participatory Indigenous research." (p.9, 110) For example, Moore *et al.* and Stiegman and Castleden highlight concerns around REBs' requests for "detailed

timeframe[s] and activities related to the research" (p.9, 110) which are contradictory to CB and/or PA research approaches that are organic in nature and respectful of the community's timeline. (90) While many REBs are amenable to project alterations, revision processes are time-intensive and divert the researcher's attention from the community. (90) Stiegman and Castleden also highlighted that the TCPS-2:

...does not give any guidelines to REBs or researchers in terms of how to navigate the tensions that arise when the ethical guidance of Indigenous peoples contradicts that of a university REB. (p.1, 90)

Thus, researchers engaging in community-based research with Indigenous partners are often placed in situations where they must either compromise their relationships with the community or violate REB processes and thus risk lose funding and/or career progression. (90) These circumstances indicate that the TCPS-2 not only enables, but often requires, tokenistic Indigenous jurisdiction over research projects.

These concerns highlight the importance of prioritizing relationships and employing relational ethics in all research projects. Stiegman and Castleden posit that REB review could be transformed to empower Indigenous partners and invert the "balance of power between the academy and Indigenous research partners". (p.4, 90) Expanding upon this, they offer that when analyzing projects, reviewers should focus on

...evaluating the strength of the researcher-community partnerships and structures of mutual accountability that have been established while acknowledging the jurisdiction of the nation in question and deferring to their authority... (p.5, 90)
As REB review remains the primary mechanism to implement the TCPS-2, communities' and researchers' concerns around the REB process are fundamental challenges that must be

addressed to appropriately reflect Indigenous research methodologies and CB and/or PA research methods. (90,97)

Building community relationships

As relationships form the foundation of Indigenous research methodologies and CB and/or PA research methods, investing time and resources into these relationships' formation is an important first step to begin participatory research projects with Inuit communities. (90)

Community relationships function to: 1) establish trust and relational accountability between outside researchers and community members to foster robust community engagement with the project, 2) situate the research within community needs and narratives to facilitate shared understandings of research goals and parameters, and 3) ensure, to the extent possible, that research outcomes and actions benefit the collective community and community members. (64,90,98,108,110,111) In a series of health research ethics workshops, Nunavummiut highlighted prioritizing relationships early in the research process as it "relieves stress for both research partnerships that cross "cultural, socioeconomic, and Indigenous-Settler bounds" (p.2609, 4), these relationships enable richer, more rigorous findings by utilizing data generation strategies that are consistent with cultural values and norms and facilitating discussions around the "complexities and interrelationships" (p.90, 108) during data analysis. (110)

Relationship building, and participatory research approaches generally, require more time- and initial resource-investment that traditional, Western research approaches. Often, participatory research projects exist within relationship-building and planning phases for several years before any data generation can occur. (64,84,90,103,108,110) These initial research phases are often ineligible for funding as there is no defined research project. (84,90) In 2022, the CIHR initiated a grant for researchers engaging in "planning activities, [and] partnership development" (112) to increase funding accessibility for participatory research projects. (113) To the best of my knowledge, there are about 162 grants available, and this is the only grant provided by CIHR that supports community partnership and project planning. (112,113) Additionally, researchers experience pressures to publish frequently which are often incompatible with participatory research timelines and processes. (84) Even within health research spheres, researchers not engaging in participatory research are often unaware of the time- and resource-investment necessary to build trusting and collaborative research relationships with communities. (84)

Respectful relationships with community members are important in all research projects. (98) Community relationships can be characterized as existing along a community engagement spectrum from "basic consultation" (p.11, 98) in which the communities who may be impacted by the research are informed of the project and asked for feedback on methods, to "communityinitiated and directed research" (p.11, 98) in which a community identifies a gap in knowledge, develops, and delivers a research project, potentially with the help of outside researchers. Relationship depth will vary based on the community's interest and engagement in the project; for example, a project exploring "geomagnetic phenomena" (p.10, 98) will likely garner less interest and engagement than a project exploring "narwhal distribution and abundance". (p.10, 98) Regardless of research content, it is crucial that researchers initiate these relationships from a place of humility and respect and that the community has been respectfully and meaningfully offered the highest degree of feasible community involvement.

It is widely acknowledged that ideally, participatory research relationships are communityinitiated. (84,98,103,110) Examples of how these community-initiated partnerships can begin include meeting at knowledge sharing events such as conferences or successfully working together on a previous project. (84) However, a study exploring researchers who engage in collaborative research with Indigenous Peoples found that most CB and/or PA research projects were researcher-initiated. (84) There are a myriad of explanations for this discrepancy between ideals and practice, including researchers unknowingly, or otherwise, engaging in colonial methods to initiate participatory research projects, time and workload pressures placed on community members that prevents them from initiating a research project, and communities' previous negative experiences with researchers and financial limitations. (84)

Practical suggestions for researchers wishing to establish relationships with communities centre around communication and community engagement strategies that are rooted in partnership and community autonomy. (45,98) Community needs and expectations around their role(s) and responsibilities are important to establish early in the research relationship building process to create mutual understanding around the community's capacity and guide the researcher's approach. (45,98) It is also critical that accountability, communication, and decision-making methods and processes are collaboratively established early to inform researchers of effective strategies they can employ that will align with community needs and foster respectful and reciprocal relationships. (45,98) Other recommendations to researchers looking to foster respectful relationships with Inuit communities include: 1) critically reflecting upon and honestly communicating intentions and goals, 2) following community timelines and processes, 3) respectfully utilizing the local Inuktut language throughout the research process, 4)

incorporating, and giving credit to community contributions, 5) engaging a diverse group of community members, especially Elders and other knowledge holders, 6) researching previous and current research projects with the community and, if available, any community-identified research priorities, and 7) promoting capacity-building activities to foster research sovereignty. (84,98,102) Once the community has indicated that they wish to proceed with a research relationship, discussions around funding, data generation and analysis strategies, consent processes, and knowledge control, ownership, and sharing can occur to facilitate a collaborative drafting process for the community research agreement.

Examples and discussions of Inuit research methodologies' and participatory methods' intersections

In this section I will review and analyze three studies that utilize Inuit research methodologies and CB and/or PA research methods in the context of the ITK's National Inuit Strategy on Research's (NISR) five priority areas. (97) These studies also illustrate the priority areas' interconnected nature.

Changing Climate, Changing Health, Changing Stories' Profile: Using an EcoHealth Approach to Explore Impacts of Climate Change on Inuit Health

This paper discusses an EcoHealth approach to explore the impacts of climate change on Inuit health in Rigolet, Nunatsiavut. (108) Community research sovereignty was promoted throughout the research process by utilizing "participatory, community-run methodological strategies to explore climate-health relationships" (p.93, 108) and data generation strategies that resulted in knowledge that is meaningful, useful, and beneficial for community members and researchers. This paper describes multiple techniques researchers utilized to ensure research ethics were grounded in Inuit epistemologies. For example, this research project was action-oriented and aimed to address a community-identified need and produce rich data to best inform evidence-based policies. (108) Authors also note that the project's success can be linked to their strength-based approach which aligns with Inuit values and epistemologies and its "unique funding structure [which] allowed community members to control the project finances" (p.99, 108). (39,40,45,65) Community-held control over knowledge was facilitated through digital storytelling methods that respect and honour Inuit "oral traditions and cultural stories". (p.97, 108) This data generation technique enabled maximal data control by community members

because "it is the participants, not the researchers, who have the power to decide what information is important, relevant, and appropriate for the research topic." (p.96, 108) Lastly, "capacity development was a project priority" (p.96, 108) whereby six community members were trained to develop and host digital storytelling workshops for other community members, including youth, and gained experience presenting research at academic conferences. This community-led approach to research has enabled the community to develop "their own approach to the research process" (p.96, 108), thus promoting research sovereignty.

"Sewing Is Part of Our Tradition": A Case Study of Sewing as a Strategy for Arts-Based Inquiry in Health Research With Inuit Women

This article discusses a project that explored Inuit women's childbirth experiences through artsbased inquiry. (4) To contextualize the research topic, at 36 weeks' gestation, women within Nunavut's Qikiqtaaluk Region typically fly to Iqaluit for birthing care. As the project was held in Iqaluit with women from across the Qikiqtaaluk Region who have experienced pregnancy, I have conceptualized this group to be the research community for the purposes of this analysis.

This project was developed "in light of calls for Inuit-led and Inuit-directed research" (p.2612, 4) and demonstrates sewing as a means of inquiry that has the potential to expand "Inuit leadership in research design and implementation, [and root] research in an Inuit epistemological and ontological framework". (p.2612, 4) Inuk women were involved throughout the research process as researchers, holders of Inuit sewing and other cultural knowledge, interpreters, and participants. Inuk researchers were critical to this project's success as they shared "the nuances of how and when to ask questions, and what questions were appropriate" (p.2609, 4) to ensure question formatting "invite[d] conversation in ways that resonated with Inuit culture and respected people's dignity and differing comfort levels in sharing." (p.2609, 4) At the intersections of research sovereignty and ethics, sewing was incorporated into this research project as "a strategy for arts-based inquiry that facilitated and enhanced the data collection component of the research" (p.2605, 4) by aiming to create a comfortable atmosphere where women feel welcome to engage in conversation and share their experiences. (4) This project was highly action-oriented as the method for inquiry itself and research goals were meaningful and beneficial for the women participating. Relationality was strengthened through the sewing activity as women shared knowledge and expertise, thus, sewing can be situated within Inuit

research methodologies by encouraging a "relational research space and co-creating research through storytelling and sharing of knowledge". (p.2611, 4) Understanding the community to be Qikiqtaaluk Inuit women who have experienced pregnancy, efforts to increase community-held control of knowledge are demonstrated by ensuring members of the research team include community members and member-checking findings with women who participated in the research. Using sewing methods, capacity building was woven into this project's structure as women were able to share knowledge with and learn from each other about this Inuit craft. Relationships among participants and between participants and local team members were also born or strengthened through this research process. Thus, sewing as an inquiry strategy can be seen "as a way to teach and carry on Inuit culture, and embody Inuit knowledge and traditions." (p.2609, 4)

Going Off, Growing Strong: Building Resilience of Indigenous Youth

Aullak Sangillivalianginnatuk (or, Going Off, Growing Strong) is a grassroots, health promotion program grounded in Inuit knowledge and culture for youth in Nain, Nunatsiavut. (68,114) In this program, youth engage in on-the-land activities run by community harvesters and other knowledge holders. Aullak Sangillivalianginnatuk was developed in response to a community-identified need to foster resilient mental health among youth through activities that promote cultural continuity and social connection. Research methods are employed to conduct program evaluation which is an "integral part of Aullak Sangillivalianginnatuk in order to measure and capture the program's impact and provide valuable information on which to make operational adaptations along the way". (p. 81, 114) Research sovereignty, ethics grounded in Inuit ways of being and knowing, community-held control of knowledge, and community-led health promotion project.

In this example, authors describe the ways in which research sovereignty is promoted through a PAR approach that guides the program evaluation's process, goals, and conduct. Program evaluation planning and methods are conducted by community members associated with Aullak Sangillivalianginnatuk, including staff members and harvesters. Additionally, this program is hosted in a pre-existing community space that was established "to provide a hub for traditional food sharing, and to help reduce food insecurity issues." (p.81, 114) Program activities and

format were based around youth-identified interests, thus Aullak Sangillivalianginnatuk "was born to meet the needs of these youth." (p.80, 114) Grounded in Inuit values and relationality, each program's cycle occurs over a period of more than a year which provides youth and program staff enough time to build trusting and safe relationships. The program also takes a strength-based approach and utilizes cultural activities to promote mental health resilience. Some community-held control of knowledge was demonstrated as both papers listed multiple community members as authors. Additionally, community representatives "helped guide Aullak Sangillivalianginnatuk towards its core goals and objectives" (p.71, 68) and continue to "advocate on behalf of youth, help select program participants, and act as key points of contact for any concerned adults and for community agencies." (p.71, 68) Capacity building among youth is a fundamental goal of Aullak Sangillivalianginnatuk; through activities, youth are encouraged to engage in cultural and social connection and youth who have completed the program are "given the opportunity to become Junior Leaders for the next cohort". (p.81, 114) For example, the Aullak Sangillivalianginnatuk program has evolved to encourage youth to distribute country food to community members who experience challenges when trying to access the community freezer (e.g., Elders and single-parent households). Community capacity is also developed among adults involved with Aullak Sangillivalianginnatuk as they gain experience developing programs that aim to support and foster positive mental health among youth.

Inuit cultural continuity and youth

There are a range of terms used to describe connection to culture among Indigenous Peoples, including cultural continuity, cultural connectedness, and enculturation. (69) In this thesis, I utilize cultural continuity as it reflects the active process by which culture is shared and gained. The historical use of the term cultural continuity in the Indigenous context forms the basis for how it is defined and used today. In 1956, Eggan first used cultural continuity in an Indigenous wellness context to describe the Hopi Nation's "transmission of cultural heritage from one generation to another – and … the means by which that transmission is accomplished". (p.347, 115) In 1998, Chandler and Lalonde described cultural continuity among First Nations youth in British Columbia (BC), Canada as "understanding themselves as connected to their own past and building future". (p.208, 70)

The evolution of the term cultural continuity reflects changes in our understandings of social, cultural, and psychological phenomena and processes. Today, we understand culture to be necessarily both fixed and evolving, to ensure its continuation within "the social realities of each generation". (p.78, 116) More recent descriptions of cultural continuity express this paradoxical nature. In 2007, Kirmayer et al., expressed that cultural continuity was expressed as depending on the "notion of culture as something that is potentially enduring or continuously linked through processes of historical transformation with an identifiable past of tradition" while emphasizing culture's "fluidity and negotiated realities". (p.77, 116) They go on to discuss how cultural identity and traditions provide resources for people to construct a social and psychological identity, in a world with interacting cultures. (116) In 2020, Newell, Dion, and Doubleday described cultural continuity in Inuit Nunangat as "the links between food, culture and health, ... the continuation of traditional culture in current society, or as the degree of social and cultural cohesion within a community". (p.64, 69) In 2021, Auger described a holistic view of cultural continuity in the context of Métis youth in BC as encompassing "identity, the practice of traditional and cultural activities, and spirituality". (p.72, 19) Pulling from the above definitions, this thesis uses cultural continuity to describe individual and community identity, participation in traditional, cultural, and spiritual activities, spirituality, and how these phenomena and knowledges are transmitted.

More recently, cultural continuity has been explored as a social determinant of health among Indigenous Peoples. (19) Social determinants of health can be defined as "non-medical factors that influence health outcomes". (117) Thus, understanding cultural continuity to be a social determinant of health illustrates its role in wellness outcomes among Indigenous Peoples. Indigenous Peoples have long called for holistic interventions and practices that are designed to foster and enhance wellness beyond the biomedical sense. (13,19,20,22,118) These sentiments are echoed in the literature and emphasize the importance of collaborative and intersectoral initiatives that use upstream, strengths-based, and community-led approaches. (13,20,22,24,118) According to Greenwood and de Leeuw, to effectively address the social determinants of health among Indigenous youth,

Approaches must be flexible, while also addressing historical and contemporary determinants and should include decolonizing strategies... Interventions should not target

individual behavioural change or focus solely on proximal determinants of child health. Instead, interventions should account for broader contexts and distal determinants that continue to influence the context and, thus, the health of the child. These broad contexts require collaborations across and between sectors and disciplines; medical or even health sectors alone cannot address or influence these determinants of health and must work in concert with other sectors such as education, child welfare, housing and justice, among others. (p.383, 24)

Cultural continuity is affirmed and protected within multiple legal documents. The Canadian Constitution affirms and protects cultural continuity, especially among children and youth. (24) The United Nations (UN) Convention on the Rights of the Child "tak[es] due account of the importance of the traditions and cultural values of each people for the protection and harmonious development of the child" (p.2, 119). (24) The UN Declaration on the Rights of Indigenous Peoples (UNDRIP) emphasizes the importance of culture and cultural continuity throughout the report. (120) While the adoption and speed with which the UNDRIP has been implemented by the Canadian government have been widely criticized, on June 21, 2021, Canada's UNDRIP Act, also known as Bill C-15, received Royal Assent and was enacted. (121–123) More research into Canada's implementation of the UNDRIP is needed.

Previous studies exploring the relationship between cultural continuity and health among Indigenous youth have found cultural continuity to significantly protect against mental health challenges including depression, anxiety, harmful substance use, self-harming behaviours, and suicide. (19,70,116) Chandler and Lalonde's (1998) influential work explored the association between cultural continuity and suicide rates among First Nations' youth in BC. (70) Using six variables measuring attempted cultural rehabilitation, they found that communities with more factors of cultural rehabilitation showed reduced suicide rates among youth. Subsequent studies have found similar findings, illustrating the critical role of culture and cultural activities in mental health promotion initiatives for Indigenous youth. (24,28,116,118,124) It is thought that cultural continuity provides youth with resources and skills to manage challenges in life; when there are disruptions to cultural continuity, youth experiencing challenges may be more likely to engage in harmful coping mechanisms as they lack the advantageous resources and skills derived from cultural and social connection. (70,116) There is also evidence of the relationship between cultural continuity and physical health outcomes including diabetes, tuberculosis, and HIV among Indigenous Peoples, further illustrating the fundamental role of culture in individual and community wellness. (125–127)

For decades, Inuit have advocated for initiatives that promote cultural continuity in their communities. With the use of CB and/or PA research methods, academia has begun to explore how Inuit communities define wellness and culture's role in wellness. Inuit holistic understandings of and approaches to wellness are grounded in IQ principles that emphasize community wellbeing and incorporate the "the social/emotional, spiritual, cognitive and physical wellbeing". (p.3, 22) Similar to research findings among First Nations and Métis, research exploring the links between culture and wellness among Inuit have found significant associations, particularly among youth and when exploring mental health phenomena. (69) Cultural continuity has also been found to be associated with food security among Inuit, emphasizing culture's central role in other social determinants of health. (128) Previous research has highlighted that initiatives aiming to improve wellness among Inuit require approaches that are community-led and grounded in IQ principles and Inuit culture. (69) More community-led research exploring ways to strengthen cultural continuity among Inuit, particularly youth, is needed to adequately inform health promotion initiatives that aim to foster community and individual wellbeing.

Social connection, Inuit culture, and youth

While dependent on others, social connection, or social connectedness, is a highly personal phenomenon. Social connection can be defined in a myriad of ways, and it can be argued that each person's description will vary depending on their needs. Baldwin characterized social connection as a "cognitive structure representing regularities in the patterns of relatedness". (p.461, 129) Lee, Draper, and Lee described social connection as "an attribute of the self that reflects cognitions of enduring interpersonal closeness with the social world *in toto*". (p.310, 130) They later contrast social connection to belongingness as social connection having an emphasis on the "independent self in relation to others" (p.310, 130) and belongingness being "defined by group membership or peer affiliation". (p.310, 130) In contrast to Eurocentric societies, Inuit identity is thought to be ecocentric, thus an Inuit-aligned definition of social

connection that reflects relational accountability is needed. (81) Drawing on previous definitions and descriptions of social connection among Inuit, this thesis describes social connection as one's reciprocal connection with and accountability to people, ancestors, and future generations that provides emotional, spiritual, and physical resources which one can access. (131,132)

There is strong evidence supporting the critical role social connection plays in individual and community wellness among Indigenous Peoples. (131–133) It is theorized that positive effects of social connection, including sense of support, belonging, and satisfaction, buffer health challenges one may face and may encourage people to engage in health-promoting behaviours. (131) Some researchers have even suggested that social connection could be as important to health outcomes as "established disease risk factors" (p.1423-1424, 131), including smoking and high blood pressure, thus illustrating social connection's pivotal role in individual and community wellness.

Social connection can be viewed as a network of social ties, of which there are varying levels. (131) Communities with higher density networks, or greater social connection, are more likely to share knowledge and resources and provide "social feedback" (p.1424, 131) which can work to re-align an individual's actions that are deemed harmful by the larger group. (134) Richmond and Ross discuss Gottlieb's three levels of social connection: micro, mezzo, and macro. (131,134) The micro level refers to one's most intimate relationships, including partners and family with whom you live, and is defined by relationships that provide "deep and nurturing emotional" (p.1424, 131) support. The mezzo level refers to those one interacts with on a regular basis for reciprocal support and respect, such as friends or teachers. (131,134) The macro level is composed of one's most distant social connections, including acquaintances or people with whom there is limited contact. (131,134) Reflecting relational accountability's role in social connection, varying levels of connection are important to community wellbeing as they embed and intertwine individuals within "the social context of their families and communities." (p.1424, 131)

As discussed above, social connection has many well-established positive associations, including increased sense of belonging, resiliency, and self-esteem. However, some research indicates that a high degree of social connection does not necessarily have a positive impact on community

wellness. Richmond and Ross discuss First Nation and Inuit communities that appear to experience high levels of social connection, but are also prone to experiencing wellness-related challenges. (131) They propose that social connection, via conformity pressures, can foster negative phenomena such as harmful substance use, in closely connected communities. (131) Conformity pressures are thought to be heightened by reduced community resources which is commonly experienced by remote and rural First Nations and Inuit communities. (131) However, conformity pressures are not inherently negative and can promote community wellness, especially in collectivist cultures. (135) Thus, one can view social connections that routinely promote negative conformity and do not provide reciprocal support and respect, as disconnections or breaks in social connection. Using this lens, patterned negative conformity pressures can be understood as a type of social disconnection that a community or individual may experience.

Social connection among Inuit is an important part of their collective wellbeing and holistic wellness. In traditional Inuit societies, relationships with immediate and extended family, and community, were essential to survive and thrive. (81) Inuit Elders have emphasized intergenerational relationships' importance to collective and individual wellness, and have highlighted the particular significance of family connections. (81,131) It is important to view Inuit social structures with a decolonial lens, as Eurocentric assumptions of family structures can be damaging for Inuit. Inuit families often live intergenerational relationships are fundamental to Inuit's societal fabric. (81) Among other harmful outcomes, colonization has disrupted Inuit society by encouraging families to live separately due to overcrowded, western-style homes and by removing children from homes deemed 'unsafe' by government employees who are often southern and non-Indigenous. (136–138) These disruptions to Inuit society have created social disconnections; however, across Inuit Nunangat, Inuit Elders and other community leaders are encouraging social connection through cultural activities and modelling positive social behaviours. (81,114,139,140)

Previous research with Inuit has found that central guiding values among Inuit include responsibility, respect, sharing, and reciprocity. These values are grounded in IQ principles and

embedded within social connections among Inuit. (132) Further, social connection was prevalent in traditional Inuit societies, with many cultural traditions and activities occurring in groups, including hunting, food preparation, and ceremonies. (8,114,141) Today, these traditions and activities are being used to facilitate healthy and reciprocal social connections among Inuit.

Social development and connection among youth is a widely researched area. However, there is limited research into social connection among Inuit youth. As discussed above, there are many facets of Inuit society that are unique from Canada's Eurocentric, non-Indigenous society. Thus, discourses and frameworks discussing youths' social development and connection in a non-Indigenous setting do not necessarily apply well in an Inuit context. (74,142–144) More research into social development and connection among Inuit youth is needed to better understand these phenomena and processes.

However, there are discourses and theories of social development and connection that apply to this thesis. These include the motivating factors for social connection such as a sense of belonging, resiliency, and self-esteem, as well as effects of harmful social ties including negative conformity pressure. (81,131,144) Additionally, generational disconnect due to social media and mobile communication that has been explored in other cultures likely exists among Inuit. (145,146) As intergenerational relationships are fundamental to Inuit society, harm associated with this disconnect may be amplified among Inuit compared to non-Indigenous populations. (81) More research into digital media's impact on social connection among Inuit is needed.

Colonization in Inuit Nunangat

This section provides a highly superficial summary of Canada's historical and ongoing colonization of Inuit. In the 1950s, Inuit began to experience regular contact with non-Indigenous people and government-directed relocations from on-the-land camps to permanent settlements commenced. (136) Inuit actively resisted these changes to society, however, non-Indigenous people continued to act in harmful ways against Inuit, including the killings of *qimmiit*, or dog teams, which were integral components of Inuit society and essential to hunt and travel long distances. (9) Beginning in the 1960s, many Inuit children were removed from their families and forced to attend residential schools or sent to live with non-Indigenous families,

sometimes thousands of kilometers from their families. (9) In these residential schools, the children were "made to forget their Inuit roots" (p.34, 9) and were punished for speaking an Inuktut language. Among Inuit and other Indigenous Peoples, residential schools created a "deep cultural and generational divide" because children could "no longer communicate with parents and grandparents" and had lost "cultural teachings, beliefs, values, and skills, especially those needed for activities on the land." (p.36, 9) Inuit continue to live with the harmful impacts of former colonization policies and tactics. (9,13)

Colonization is an ongoing practice in settler colonial states such as Canada. Colonization has displaced Indigenous societal structures, including those of wellness, justice, and education; thus, colonization continues to have "profound effects at community, social, familial, and individual levels" (p.2, 81) among Inuit. (24) While the territory of Nunavut was founded by and for Inuit, it exists within Canada's colonial structures which continue to undermine and obstruct Inuit's sovereignty. (5,97,147) For example, even though education, healthcare, and justice systems are under territorial jurisdiction, federal funding for these services is associated with requirements that are rooted in colonial values and expectations. (147–149) Further, Canada's criminal laws and the Supreme Court of Canada are under federal jurisdiction, thus their implementation in Nunavut by the Royal Canadian Mounted Police (RCMP) continues to impose colonial structures upon Inuit. (148,149)

Research conducted with Inuit "has largely functioned as a tool of colonialism" with Inuit often viewed "as either objects of study or bystanders". (p.5, 97) Regardless of researchers' intent, this exploitative and racist research style at its best magnifies pre-existing challenges and traumas experienced by communities, and at its worst manifests new challenges and traumas for Inuit. (97) This legacy has resulted in many communities and individuals having a traumatic relationship with research that "continues to be reflected in current approaches to research governance, funding, policies, and practices". (p.5, 97) As noted above, multiple Inuit political and advocacy institutions, including the ITK and QHRC, have called for research to be led by Inuit and promote Inuit sovereignty. (97,109,150)

Examples of impacts of colonization among Inuit

As noted above, colonization has disrupted all facets of Inuit's pre-colonization society. To exemplify colonization's wide-ranging effects, this section briefly discusses two examples of its harmful effects among Inuit and current initiatives Nunavummiut are taking to address these challenges.

Tuberculosis

Tuberculosis (TB) is a highly infectious disease that mainly targets the respiratory system and spreads by airborne droplet nuclei. In this thesis, active TB refers to someone who is currently experiencing a TB infection, may or may not be symptomatic, and can transmit the disease. (151) Overall, Canada is considered to be a low-incidence country for active TB. (152) However, in 2016, Inuit living in Inuit Nunangat experienced active TB rates of more than 300 times those experienced by Canadian-born non-Indigenous people, thus illustrating a health disparity between non-Indigenous people and Inuit in Canada. (151,152) The impacts of these high rates of TB are exacerbated by barriers to culturally-appropriate healthcare that many communities across Inuit Nunangat experience, including many communities lacking a local physician and healthcare workers being unaware or neglectful of Inuit cultural values and norms. (153,154)

TB was initially spread to Inuit Nunangat by "early European explorers and whalers" (p.7, 151). (154) Throughout the first half of the twentieth century, medical officers continued to report "concerns about the high rates of TB in their areas and the need for improved services." (p.7, 151) Early interventions to address the growing TB crisis among Inuit "marked a general shift in healthcare from Inuit ways to Euro Canadian ones" (p.38, 154) and included screening, removal of infected people from communities, and immunization campaigns. (151) By the middle of the twentieth century, "it was estimated that one in seven Inuit was living in a southern sanitorium" and "incidence rates of TB among Inuit … were in the 1500-2900 per 100,000 range." (p.8, 151) The staggering impact of TB among Inuit "cannot be overstated." (p.8, 151)

In the 1960s, a mass screening, testing, and treatment program began throughout Inuit Nunangat. (151) Under this program, the incidence of active TB among Inuit decreased fifteen percent each year until 1997. (151) While similar strategies of community-wide screening, testing, and treatment have been ongoing since 1997, the rates of TB among Inuit, mainly in the Eastern

Arctic, have increased substantially. (151) In 2016, the incidence rate of TB among Inuit in Inuit Nunangat was estimated to be about 183 per 100,000. (151) Additionally, TB outbreaks and TB-related deaths among youth are becoming increasingly common in Inuit Nunangat. (151)

The persistently high rates of TB in communities across Inuit Nunangat reflect health inequities present among Inuit. (151) Specifically, contributing factors include the region's prevalence of food insecurity, overcrowded housing with poor ventilation, and culturally-inappropriate TB surveillance and treatment protocols. (152,154) Poverty has also been cited as a contributing factor with non-Indigenous people in Inuit Nunangat earning almost \$70,000 more annually than their Inuit counterparts. (151,154) These contributing factors, among others, can be linked to colonization, illustrating the web-like results of Canada's historical and ongoing colonial practices and policies. For example, during government-directed relocations, the Canadian government built Eurocentric housing that did not take into consideration Inuit's intergenerational family structures, thus encouraging overcrowded homes. (9,151,153,154) Today, quality housing shortages remain a crisis across Inuit Nunangat. (9,151,153,154) Additionally, food sovereignty is associated with TB as current TB medication must be taken with food and experiencing malnutrition increases one's risk of a severe TB infection. (153,154) Inuit's barriers to food sovereignty are widely considered to be a direct consequence of Canada's historical and ongoing colonization; colonial policies and practices that can be linked to Inuit food sovereignty include government-directed relocations and federal food security programs that are incompatible with and disruptive to Inuit culture (e.g., Nutrition North). (6,7,9,15,153,154)

In 2018, the ITK and Government of Canada committed to "reduce the incidence of active TB by 50% ... by 2025, and eliminate the disease by 2030 in Inuit Nunangat." (p.699, 152) (151) Throughout the report, a holistic and coordinated approach that addresses the social determinants of health was highlighted as essential to reduce and eliminate TB among Inuit. (151) Until recently, strategies to prevent and control TB have largely focused on medical interventions. (151,154) As proposed by the Inuit-specific TB Strategy (2013), programs that aim to address the interconnected social determinants of TB should consider: 1) housing, 2) food sovereignty and nutrition, 3) mental wellness, 4) income distribution, and 5) accessibility of health services. (151,153) It is well-acknowledged that prior to colonization, these social factors were equitably

distributed among Inuit society. Thus, challenges Inuit experience today related to these social determinants of health are a direct consequence of historical and ongoing colonization. (151,153)

The Inuit TB Elimination Framework provides both "strategic direction for Inuit regions ... [to] develop and implement region-specific TB elimination action plans" and "an evidence-based, transparent tool for ensuring accountability and measuring progress toward TB elimination." (p. 16, 151) Thus, the framework does not prescribe detailed plans, rather gives regions and communities the tools and knowledge to transform TB care and prevention and develop tailored programming. (151) The Inuit TB Elimination Framework is grounded in principles such as collaborative decision-making processes, respect for Inuit values, knowledge, and culture, and equitable and efficient resource use. (151) The priority actions laid out in the framework emphasize the importance of increasing Inuit sovereignty, addressing the social determinants of health, and empowering and engaging communities throughout program planning, delivery, and evaluation. (151)

Grounded in the Inuit TB Elimination Framework, Nunavut Tunngavik Incorporated (NTI), in partnership with the Government of Nunavut, created the Nunavut Tuberculosis Elimination Action Plan, 2020-2023. (155) This regional action plan has ten focus areas including housing, food sovereignty, and Inuit-governed research. (155) As of November 2022, the full action plan is unavailable to the public. The territory's previous strategy to prevent and manage TB was published in 2017 and focused on clinical aspects such as screening, contact tracing, and treatment and care plans. (156)

A study that evaluated the ITK's goal of reducing the annual TB incidence in Inuit Nunangat by 50% by 2025 found it to be implausible; they found that "if active TB cases are identified rapidly within one week of becoming symptomatic" (p.699, 152) a 50% reduction in active TB incidence could be achieved between 2025 and 2030. This study also found that eliminating TB from Inuit populations altogether "would extend beyond 2030" and would "require high rates of contact-tracing". (p.698, 152) Given these findings and Nunavut's ongoing TB outbreaks, program evaluations are needed to determine effectiveness and provide realistic timeframes and goals.

Substance use

Substance use can be defined as the consumption of psychoactive substances including tobacco, alcohol, and drugs. (157) There are an array of widely-debated terms to describe the harmful effects of substance use including substance addiction, substance dependence, substance use disorder, substance abuse, and problematic substance use. Substance dependence can be defined as a physical dependence on a substance, characterized by increased substance tolerance and withdrawal symptoms. (158) Addiction can be defined as "the loss of control over the intense urges to take the drug even at the expense of adverse consequences". (p.764, 158) Substance abuse or problematic substance use can be defined as "drug use [that] dominates a person's life at the expense of other activities and has negative mental and/or physical side effects". (p.345, 159) Currently, physicians in Canada use the Diagnostic and Statistical Manual of Mental Disorders (DSM), Fifth Edition to diagnose patients with substance use disorder. (157) Discussion of the DSM-5 and its applications is beyond this thesis' scope.

In this thesis, the term 'harmful substance use' is used to encompass the phenomena of substance dependence, substance abuse, and addiction. Harmful substance use is preferred over substance abuse or problematic substance use as it emphasizes the experience of the phenomenon, rather than placing blame on the individual. Thus, harmful substance use is considered less stigmatizing than substance abuse or problematic substance use. (160) Harmful substance use can be defined as a "persistent preference for drug rewards that provide immediate reinforcement... but [with] longer term costs in important life-health domains... as compared to drug-free alternatives that typically have lower short-term, but higher long-term value". (p.740, 161) In other words, harmful substance use occurs when someone experiences a pattern of substance use that provides an immediate reward (e.g., euphoria, anxiolytic, or analgesic effects), but results in delayed or long-term negative consequences (e.g., harmful physiologic effects). In comparison, healthier substance use can be characterized by engaging in substance use in a way that may lower immediate rewards (e.g., depressive symptoms), but there are delayed or long-term benefits (e.g., healthy familial relationships and academic or professional achievement). In this thesis, I use the phrase 'healthy relationships with substances' to describe abstinence or substance use patterns that mitigate the harm to the individual and/or their community.

Prior to European contact, Indigenous Peoples of Canada did not consume alcohol, opiates, and stimulant drugs such as cocaine or methamphetamines. (162,163) It is widely understood that the intergenerational trauma experienced by Indigenous Peoples is reflected in the disproportionately high prevalence and rates of harmful substance use among Indigenous Peoples, compared to non-Indigenous people in Canada. [111,112] For example, the 2012 Aboriginal Peoples Survey found that the rate of 'heavy drinking' (defined as "five or more drinks on one occasion, at least once a month in the twelve months preceding the survey" [p.8, 164]) among off-reserve First Nations, Métis, and Inuit were 35%, 30%, and 39%, respectively. In comparison, the 2011 National Household Survey found 23% of non-Indigenous people to be heavy drinkers. (164) However, these same surveys found that off-reserve First Nations (30.9%), Métis (24.9%), and Inuit (37.6%) were more likely to abstain from drinking altogether than non-Indigenous people (23.5%). (164) This indicates that drinking habits, and likely general substance use, is a highly complex social phenomenon among Indigenous Peoples that requires a nuanced, holistic approach. (163)

As noted previously, when non-Indigenous people began interacting with Inuit regularly, they often did so in ways that caused harm and disrupted Inuit society. (136) In the substance-related context, non-Indigenous people introduced and weaponized alcohol, tobacco, and other substances. (163) As seen among other Indigenous populations, alcohol's introduction, in particular, caused harms among Inuit that are still felt today. For example, in Nunavut between 1999 and 2007, 23% of all premature deaths were linked with heavy alcohol consumption and 30% of all homicides were associated with substances. (165) There is limited information on drug usage in Nunavut, however, there are RCMP reports of illegal drugs being trafficked and sold throughout the territory. (166,167) More research into drug usage in Nunavut is needed.

Many of the perceived ramifications of harmful substance use can be viewed instead as consequences of colonization-related trauma. (163) This approach aims to emphasize the larger system's role in substance use patterns, and minimize stigma and blame placed on the individual or community. Alternatively, utilizing a strengths-based approach, harmful substance use can be viewed as an adaptive means to survival after exposure to trauma. (168)

Currently, there are a variety of strategies employed in Nunavut to prevent and address harmful substance use. In 2016, the Government of Nunavut published a report detailing fifteen initiatives aiming to reduce harms associated with alcohol use in the territory. (165) These initiatives are grouped into four categories: 1) prevention and education, 2) harm reduction, 3) treatment, and 4) enforcement. The prevention and education stream aims to empower Nunavummiut with the knowledge and skills to make informed choices related to alcohol consumption and increase accessibility to resources and supports for all Nunavummiut. (165) The harm reduction stream aims to ensure Nunavummiut receive collaborative, intersectoral care and empower Alcohol Education Committees (AECs) and Community Health and Wellness Committees (CHWCs). (165) Under the harm reduction stream, Nunavut's Finance Ministry has opened beer and wine stores in Igaluit and Rankin Inlet to increase access to alcohol to "disrupt bootlegging" (p.12, 165) in the territory. (169) The treatment stream aims to ensure screening, treatment, and referral services are available and culturally appropriate for all Nunavummiut. The enforcement stream aims to "control and influence the availability and distribution of alcohol" (p.14, 165) by improving the permitting system which allows Nunavummiut to import alcohol into the territory, increasing support for RCMP activities that disrupt crime and harmful behaviours, and establishing alcohol pricing controls. Evaluation of the implementation of this report's strategies and initiatives is needed. Research into unregulated and pharmaceutical drug use among Nunavummiut is needed to facilitate a comparable strategy for other substances.

Across Nunavut, there are a variety of community-level strategies aiming to prevent and address harmful substance use. For example, AECs and CHWCs are composed of locally elected community members who are passionate about community wellbeing. (165,170,171) In communities where alcohol is permitted, AECs function to control alcohol importation and educate community members on responsible alcohol consumption. (171) CHWCs are responsible for the health and wellbeing of the community, including identifying and supporting community needs and priorities. (170) These committees are intended to act independently, under the council of Nunavut's Liquor Commissioner, thus aiming to instil autonomy and community-led decision-making regarding substance use and greater community wellbeing. (172) The Nunavut government also facilitates training of community-based, paraprofessional

counsellors who can offer culturally-appropriate counselling. (165) More research into these community-level strategies' effectiveness is needed.

Substance use among youth is a complex social phenomenon with considerable research exploring its underlying causes, patterns of engagement, and prevention and treatment strategies. A full literature review of substance use among youth is beyond the scope of this thesis, however, the underlying causes, patterns of engagement, and prevention strategies are briefly discussed below.

Worldwide, there are an array of factors that are known to encourage healthy relationships with substance use among youth. (173–179) These include resilient mental health, sense of belonging and support from family members and community, and models of healthy relationships with substances. (173–179) For Indigenous Peoples, there are additional social factors that are thought to encourage healthy relationships with substances including connection to their land, culture, and language. (19,20,24,118,124) Canada's historical and ongoing colonization reality has created barriers to these health promoting factors among Indigenous Peoples, however, many Inuit communities are making efforts to foster and grow these factors.

(4,9,10,12,13,108,114,180)

There are a variety of strategies aiming to prevent harmful substance use among youth in Canada. These prevention strategies are often deployed in schools and can be viewed as grounded in values along a spectrum of abstinence to harm reduction. Examples of abstinence-oriented strategies include the well-known 'Just Say No' and Drug Abuse Resistance Education (DARE) campaigns which encourage youth to abstain from all substances and aim to foster skills that help youth avoid engagement in substance use, organized criminal activity, and violence. (181) Critics of abstinence-based approaches often claim these initiatives promote fearmongering, stigmatization, and do not adequately provide youth with the skills and knowledge to make informed choices. (182–190) Harm reduction-oriented approaches aim to foster youths' informed decision-making skills regarding substance use, by educating youth on peoples' motivations to use substances and risks associated substance use, and by providing youth with skills to abstain from or engage in substance use in healthier ways. Critics of harm reduction approaches claim these initiatives do not adequately emphasize the risks of substance

use to youth, assume youth will inevitably engage in substance use, and require personnel who are educated on harm reduction principles. (191) Eurocentric approaches to harm reduction can be "contentious and contested" in Indigenous communities, usually on "moralistic grounds that are thought to be a result of centuries of colonization and Christianization". (p.10, 192) However, Indigenous approaches to harm reduction can be "rooted in Indigenous Knowledges and worldviews" and focus on "mitigating the egregious harms of colonization". (p.10, 192) Indigenous approaches to harm reduction are varied and often include activities aiming to foster cultural continuity which may alleviate many Indigenous Peoples' concerns of harm reduction strategies. (192)

In Canada, municipal, regional, and/or territorial governments usually determine governmentprovided substance use programming for youth within the region, with some schools or communities selecting additional programs based on their identified needs. (193) Laws prohibiting people under the age of majority buying, possessing, and consuming alcohol and cannabis are also examples of strategies to prevent substance use among youth. There are six communities in Nunavut where alcohol is prohibited for all, but people who buy alcohol for or sell alcohol to youth are subject to additional charges and/or fines. (172)

There are a plethora of community-led initiatives employed throughout Inuit Nunangat aiming to prevent and address harmful substance use among children and youth. These initiatives are usually developed and administered through the local community or a non-governmental organization. (193) Examples of these initiatives include the Makimautiksat Wellness and Empowerment camp and Aullak Sangillivalianginnatuk program. (114,194) The success of these initiatives can be attributed to their grassroots, community-led approach that seeks to build upon the community's strengths. (114,194)

Discussion of strategies that aim to build cultural continuity and social connection among youth in Nunavut

As discussed above, Canada's historical and ongoing colonization practices and policies, as well as the resulting intergenerational trauma, have created barriers to cultural continuity and social connection among Inuit. (9,10,12,13,180) Increased cultural connection is known to be positively correlated with social engagement and other health-promoting behaviours. (20,24,140) The

inverse causal relationship is also seen, or, Inuit who are more socially connected or engage in other health promoting behaviours are more likely to be better connected to their culture. (118,124) These inverse relationships are important to acknowledge as they illustrate the interconnected nature of these phenomena. This interconnectedness implies that when an intervention is taken to support the growth of one or more factors, all other factors will benefit, which in turn amplifies the initial intervention's impact. As social connection is known to encourage health-promoting behaviours, and cultural continuity is considered a social determinant of health among Indigenous Peoples, programs that promote cultural and/or social connection can be viewed as health promotion initiatives. (19,131,132)

Evidence-based best practices

Health promotion strategies that aim to build cultural continuity and social connection among youth in Nunavut require an evidence-based approach to ensure effectiveness. In this section, I will detail three of many evidence-based best practices for health promotion work among Inuit: community-led, strengths-based, and land-based.

For decades, Indigenous Peoples have called on academia and public institutions to utilize a community-led approach in health promotion work. Community-led approaches to health promotion are "defined and implemented by the community, who are in control of all resources, parameters, and decisions". (p.2, 195) Community-based is often used synonymously with community-led as both terms refer to collaborative approaches between a community and an outside actor, such as an NGO or research institution. (196) However, in health promotion, community-based approaches can vary in the extent of community engagement, and fundamentally differ from community-led approaches due to community-based approaches' shared control of resources and decision-making power between the community and outside partner. (195) In contrast, in community-led health promotion, the community retains their power, and the outside actor supports program activities' planning and delivery by providing resources, skills, and/or knowledge as requested. (195) Importantly, community-led approaches use a collective decision-making process with a diverse group of community members to determine community needs, priorities, and interests. (195,196) Community-led approaches to health promotion result in strategies that are either community-created or -tailored, by altering an existing strategy's framework to the community's needs, interests, and culture. Community-led
processes result in culturally-appropriate, community-oriented initiatives, and thus tend to be more effective than homogenous national or regional strategies. (195,196) The Aullak Sangillivalianginnatuk program discussed previously is an example of a community-led health promotion initiative. (114) This program was developed with a diverse group of community members and a mental health professional working with the community, is grounded in IQ principles and Inuit culture, and sought to meet community-identified needs and priorities. (114)

Strengths-based approaches utilize and enhance communities' strengths, while also considering their challenges. (197) Strengths-based approaches are solutions-oriented and focus on ways forward, rather than dwelling on current challenges. (197) Tuck discusses the importance of celebrating a community's strengths, while acknowledging their challenges, "even when communities are broken and conquered, they are so much more than that – so much more that this incomplete story is an act of aggression". (p.416, 50) Highlighting a community's strengths functions to assert a desire-based framework and focuses on solutions, rather than amplifying challenges of which the community is already well-aware. (50,197) Solutions determined through strengths-based approaches typically engage community members more effectively than damage-centred narratives as they encourage participation by providing a positive way forward. (197–199) Additionally, strengths-based initiatives typically use resources efficiently as they consider a community's strengths and challenges related to human resources, funding, and space during planning, implementation, and evaluation activities. (199)

Indigenous Peoples often advocate for land-based, health promotion initiatives for youth. These initiatives are either implemented entirely or partially on-the-land and aim to connect youth with their land. Elders are often engaged throughout the planning and delivery of land-based initiatives. (193,194) Land-based initiatives can be seen as a decolonizing approach to health promotion because they aim to reconnect peoples to their land, thus,

...if colonization is fundamentally about dispossessing Indigenous peoples from land, decolonization must involve forms of education that reconnect Indigenous peoples to land and the social relations, knowledges and languages that arise from the land. (p.1, 200)

Land-based initiatives also foster cultural and social connection as these activities are grounded in culture and IQ principles and administered in a group setting. Inuit Elders have called for more land-based initiatives to share knowledge with youth and reconnect them to their land and culture. (193,194) Land-based initiatives are inherently strengths-based and typically use a community-led approach, thus incorporating multiple evidence-based best practices. (139,200)

School-based strategies

Since 1989, Inuit in Canada have advocated for and worked towards a self-determined curriculum that best serves Inuit children and youth. (57) The Nunavut government has made efforts to facilitate cultural continuity and social connection in the school system. Nunavut's curriculum is based on the "landmark document" (p.296, 57) 'Inuit Qaujimajatuqangit: Education Framework for Nunavut Curriculum', which credits the knowledge of more than 55 Inuit Elders and 65 northern and Inuit educators. (201) Learning from the knowledge shared by Elders, Nunavut's curriculum goals are based on IQ principles and Inuit laws and values. (57) The curriculum is composed of four integrated strands, aiming to better approximate "the holistic nature of Inuit knowledge" (p.115, 202): 1) *Nunavusiutit:* heritage, culture, economics, world news, and environmental sciences, 2) *Iqqaqqaukkaringniq*: math, technology, practical arts, problem-solving, and analytical and critical thinking, 3) *Auglajaaqtut*: personal safety, wellness, goal setting, and society and volunteerism, and 4) *Uqausiliriniq*: creative arts, language, and reflective and critical thinking. (57,203) The ITK has made clear the continued need for curriculum development grounded in IQ principles and Inuit culture. (57,204)

Strengths of Nunavut's school-based strategies to foster cultural continuity and social connection include required Inuktut learning for all Nunavummiut children and efforts to ensure curriculum is delivered in a holistic manner that aligns well with IQ principles and Inuit cultural values. (201,203) Consistent with holistic approaches to teaching and learning, land-based education is also encouraged. (201) In Inuit culture, Elders are seen as teachers; Nunavut's curriculum reflects this by including Elders and their teachings as critical components of the school curriculum. (57,201,203,205) Additionally, whenever possible, Inuk community members are employed in the school system, further facilitating cultural continuity and pride. (57)

Challenges of Nunavut's school-based strategy to foster cultural continuity and social connection mainly centre around curriculum delivery and human resources. (206) Similar to healthcare workers, educators in Nunavut are highly transient and often culturally incompetent. (205,207–

209) This high turnover creates an unstable learning environment where students are often unable to form long-term, trusting relationships with educators. (57) As of 2013, two-thirds of teachers in Nunavut were from southern Canada (210), and in 2018, 88% of principals were "English-speaking non-Inuit". (207) Nunavut's Department of Education provides a 'Teacher's Key Essentials Guide' for all incoming teachers which briefly discusses IQ principles and ways to apply them in the classroom. (206) As well, Nunavut's Department of Education delivers an optional, but highly encouraged, virtual orientation to teaching in Nunavut. (211) In the past, orientation activities have included virtual conference calls, presentations, and online courses. (211) Material covered in these orientation activities include expectations for life in Nunavut, cultural sensitivity and trauma-informed practice, and discussions of IQ principles and how they apply in the classroom. (211) Given Nunavut's Inuit-specific curriculum, southern teachers' limited training is often inadequate and can result in poor incorporation and delivery of Inuktut languages, IQ principles, and Inuit culture in the school system. (206–208)

To address the low prevalence of teachers who can speak an Inuktut language, schools often assign their few Inuk teachers to deliver language classes. (57,212) However, Nunavut's curriculum is intended to be holistic and incorporate IQ principles and Inuit culture throughout all subject areas. Thus, the high prevalence of southern non-Indigenous educators often results in a siloed approach to pedagogy, common in schools in southern Canada, that does not align well with Inuit culture. (57,204,212,213) Nunavummiut's concerns over the quality of education in the territory cannot be overstated; in October 2021, the NTI filed a lawsuit against the Nunavut Court of Justice regarding Nunavut's lack of Inuit culture and Inuktut language delivery in schools. (214)

Learning experiences outside the classroom are highly encouraged by Nunavut's Department of Education who has partnered with Nunavut's Department of Environment. (201) Examples of this partnership include land-based camps whose programming has been co-developed by the departments of Education and Environment, in conjunction with local and regionally-based organizations. (215) Other initiatives include school presentations by conservation officers, community clean-ups, and the 'Students on Ice' program. (215) There are also examples of unique partnerships between schools and local organizations that facilitate on-the-land learning across Nunavut. (193) However, while land-based education is promoted, often educators must

take the initiative to plan and/or deliver these programs which can be highly time-consuming. (216) Similarly, including Elders in school-based learning is encouraged, but in practice, it is often dependent on individual educators making these connections which results in ad-hoc inclusion. (205) In Eurocentric education systems, Elders' inclusion and land-based learning are typically viewed as curriculum enhancements, rather than fundamental components of students' education. (217–219) Thus, southern, non-Indigenous teachers are less likely to adequately engage in these educational practices, especially given their limited training. (57,212,213) Previous research has indicated that Elders view school-based education as inadequately fostering a connection to the land among Inuit youth, further emphasizing the current gaps present in Nunavut's educational policy and practice. (216)

It is well-acknowledged that increasing the prevalence of Inuit educators in Nunavut would address many of these challenges. Since 1979, there have been efforts to increase the accessibility of education careers for Nunavummiut. (220) Today, the Nunavut Teacher Education Program offers Nunavummiut a Bachelor of Education through the Nunavut Arctic College and Memorial University. This program's accessibility and effectiveness needs to be evaluated. The Nunavut government could also increase its teacher orientation's rigour by thoroughly discussing the curriculum and cultural context, and requiring a basic level of an Inuktut language for non-Inuit teachers. (221) Additionally, southern, non-Inuit teachers should be encouraged and incentivized to meet families outside of the school context and participate in community events, when invited, to better understand the "milieu in which [they have] chosen to work." (p.253, 208) Lastly, requiring, rather than encouraging, educators and schools to meaningfully incorporate land-based learning, engage with local Elders, and ensure Elders are respectfully included in planning land-based learning opportunities would likely improve the quality of education for Inuit youth.

Out-of-school strategies

There are a variety of strengths associated with out-of-school strategies aiming to foster cultural continuity and social connection among youth, including afterschool programs, on-the-land camps, and whole-family programming. These programs are often community-led and delivered by a local or regional organization, resulting in initiatives that are developed and delivered in culturally-appropriate processes that meet community-identified needs and priorities. (193)

These initiatives vary based on content, format, and community, but they are often land-based and include Elders during planning and delivery phases.

There are a variety of out-of-school strategies implemented throughout Nunavut. In Gjoa Haven, an Elder-youth land camp was developed to explore the connection between community wellbeing and caribou. (216) During initial discussions with Elders in the community, the researchers learned that Elders find Elder-youth land camps to be the "most effective means to share their knowledge, for youth to learn, and researchers to engage in respectful research." (p. 261, 216) With the Elders and other community members, the researchers developed, facilitated, and evaluated three land camps using the Qaggiq Model for Inuktut knowledge renewal. The authors discuss some of the complexities of cross-cultural research and highlight the importance of local leadership, thorough discussions around ethics, safety, and responsibilities, facilitating opportunities for experiential learning, and program sustainability. (216) This influential study provides a framework for land-based learning that other communities wishing to learn from Elders can utilize and adapt to their needs. (216)

The QHRC's 'Makimautiksat Wellness and Empowerment' camp and 'Eight Ujarait/Rocks Model' is as an example of a community-led health promotion program that was deployed in six communities across Nunavut. (194) The program is intended for youth, can be delivered in English or an Inuktut language, and is administered over eight to ten days, with two to three days on-the-land. Parents have expressed pride that this initiative was "developed by Nunavummiut, for Nunavummiut" (p.5, 194) and has received wide praise for its culturally-grounded approach to health promotion. Overall, the program has been found to facilitate cultural continuity and social connection through activities that promote intergenerational knowledge sharing and foster a sense of belonging. (194)

Similar to complexities surrounding CB and/or PA research methods, communities planning and implementing out-of-school health promotion programs often experience challenges related to funding institutions' requirements regarding program design, capacity, and evaluation. A systematic review found funding to be the main resource barrier to planning and implementing out-of-school initiatives. (222) Funding for out-of-school strategies can be challenging as grants are the main funding source for these types of initiatives and their applications are often onerous

and inaccessible to people not affiliated with a research institution. (94,95,223) Additionally, grant funding is often short-term, thus creating challenges when planning sustainable initiatives, and grants' required program evaluation is often not culturally competent for Indigenous Peoples. (95,224,225) Previous research has indicated that Inuit Elders prefer to engage with a small number of youth at a time. (216) Decreasing the ratio of youth to instructors increases the programs' proportional cost which can be challenging to justify to funding agencies. (216) Importantly, and in contrast to school-based strategies, challenges surrounding out-of-school initiatives appear to centre around funding, rather than content and approach. This fundamental difference is important to acknowledge when exploring policies that aim to increase the prevalence of community-led cultural initiatives for youth.

Potential of community-based participatory action research to explore and foster cultural continuity and social connection among Inuit youth

As discussed above, CBPAR methods are a solutions-oriented approach to research that situate community members at the helm of the research to lead and/or direct all research activities. (33,88) CBPAR methods that are meaningfully rooted in the Aajiiqatigiingniq Research Model and informed by the ITK's NISR, can function to promote research self-determination in an Inuit context. (33,35,65,88,89) However, these research methods themselves can further promote community wellbeing by fostering cultural continuity and social connection among youth. Specifically, a diverse group of community members guiding all research activities in an Inuit context would likely facilitate intergenerational knowledge sharing among community members and create opportunities to discuss IQ principles' application in a modern context. It has been well-acknowledged that intergenerational knowledge sharing is urgently critical to facilitate cultural continuity as Inuit Elders today may have experienced life before the governmentdirected relocations that began in the 1950s. (22,136) Often, IQ principles and Inuit cultural values are viewed as 'traditional knowledge', undermining their dynamic nature and relevance in today's world. (226) Experiential learning about IQ principles' and cultural values' application in various contexts is critical to enable cultural continuity; CB and/or PA research methods offer an opportunity to demonstrate to youth the modern value and applicability of these principles and values. (11,22,56,58,62,63) Therefore, research that utilizes CBPAR methods and is grounded in the *Aajiiqatigiingniq* research methodology and ITK's NISR, has the capacity to facilitate a rich

and ongoing intergenerational discussion of IQ principles and cultural values, thus promoting cultural continuity *through* the research process.

As seen in other CB and/or PA research projects, involving youth throughout the research process can facilitate and strengthen reciprocal relationships among youth and other community members, and foster a sense of collective pride. (108,227) As CBPAR methods and the Aajiiqatigiingniq Research Model necessitate action-oriented approaches to research, youth can be empowered as researchers and actively participate in the decisions that will impact their community. These collaborative decision-making processes have the potential to promote a sense of belonging, relational accountability, and self-determination among Inuit youth. (39,108,227) Thus, CBPAR research methods grounded in a culturally-relevant methodology can extend beyond a knowledge generation and sharing framework, and have the potential to be a health promotion initiative that fosters cultural continuity and social connection among youth.

In this chapter I have appraised, analyzed, and synthesized literature related to the topics of: the terminology surrounding Indigenization and decolonization, Inuit research paradigms, CBPAR methods, guidelines for researchers wishing to work with Inuit communities, community engagement through relationship building, cultural continuity and social connection among youth, strategies to foster wellness-promoting behaviours among Inuit youth, historical and ongoing colonization in Inuit Nunangat, and CBPAR as a health promoting initiative itself. As demonstrated throughout the chapter, these topics are complex, context-specific, and interrelated thus, a nuanced approach is needed when applying these knowledges.

Chapter 3: Methods

In this chapter, I describe my application of CBPAR methods within the Aajiiqatigiingniq Research model, autoethnographic data generation, and strategies I have employed to analyze ethical conduct and rigour.

Applying the Aajiiqatigiingniq Research Model and community-based participatory action research methods

The definitions and appropriate applications for the terms methodology and methods are nuanced, but important to keep in mind. Methodology is essentially "the study of methods." (p.30, 38) While methodologies are well-researched and thoroughly characterized, a research project's methodology is necessarily both dynamic and static as it is, "the established and evolving approach to and foundation of a research study." (p.11, 228) Within methodology, methods are the collective research strategies and processes that together determine one's approach to participant recruitment, data generation, handling, and analysis, and knowledge sharing. (38) Data generation and analysis strategies are the techniques used to create and analyze data and include tools such as focus groups, surveys, or interviews; in this thesis, I utilize analytical autoethnographic data generation and analysis as discussed in the next subsection. (38,228)

The Aajiiqatigiingniq Research Model has been discussed in length in Chapters one and two. Given the project's current phase and the value in discussing CBPAR projects that are in an early planning stage, this work discusses my experiences engaging with an Inuit community during the first two stages of the Aajiiqatigiingniq Research Model. These initial phases focus on building relationships and mutual understanding through discussions of current realities, research goals, existing knowledge around the focus areas, and roles and responsibilities throughout the research process. (65) Further discussions and relationship building are needed before we can proceed to data generation, analysis, and knowledge sharing phases. It is important to note that in this project's application of the *Aajiiqatigiingniq* research methodology, the described phases will not be discrete or singular; rather, they will occur concurrently and cyclically to ensure data saturation is achieved and community-identified research goals are met.

CBPAR methods are grounded in a community-led, action-oriented approach to research, thus, they complement the Aajiiqatigiingniq Research Model. CBPAR's characteristics have been thoroughly discussed in the previous two chapters. In summary, CBPAR was selected for its prioritization of relationships and community-directed, solutions-centred approach to data generation and knowledge sharing. Additionally, CBPAR methods are grounded in a flexible approach to research which allows them to be easily combined with other methodologies and supports a variety of data generation and analysis techniques.

When applying the Aajiiqatigiingniq Research Model and CBPAR research methods, my approach and understanding of my role can be visually described by Wilson *et al.* (2020)'s Sikumiut model (Figure 6). (43) This model was collaboratively created with Mittimatalik community members and is grounded in the ITK's NISR. (43) The Sikumiut model illustrates the role of non-Indigenous research partners in Inuit, community-led research projects as fundamentally promoting Inuit self-determination by encouraging community-led decision-making processes, prioritizing community-identified needs, aligning the project with community values, and fostering Inuit youth capacity. (43) As a southern non-Indigenous researcher, I will continue to apply and reflect upon this model in future work with the community.



Figure 6. Sikumiut model for non-Indigenous researchers' role in Inuit research. (43)

Analytical autoethnographic data generation and analysis

Autoethnography is a relatively new qualitative data generation strategy that encompasses other strategies such as personal narrative, interpretive biography, and socioautobiography. (38) Autoethnographic techniques are a self-narrative tool wherein one connects and critically analyzes themselves in relation to one or multiple cultural context(s). (38,228) Analytical autoethnography is a type of autoethnographic inquiry in which the researcher is considered and identified as a participant, and one that aims to increase and improve "theoretical understandings of broader social phenomena" through analytic research. (p.375, 229, as cited in 228)

I utilized Smith and Sparke's figure, "A typology of narrative analyses" (230), to select autoethnography based on my role as the storyteller in this thesis. Authoethnography questions, "What am I learning by examining my identities, power, privileges, and penalties within one or more cultural contexts?" (p.7, 228) Unlike positivist data generation strategies, autoethnography embraces subjectivity and engages first-person voices and perspectives throughout the data. (228) Autoethnographic techniques necessitate critical reflexivity which enables unique findings based on the intersections of personal perspectives and values, literature, current realities, and meaning in greater society. (228) These characteristics collectively highlighted autoethnography as the most appropriate and effective data generation technique for this thesis because I explore my experience during the early phases of a CBPAR research project with an Inuit community.

Autoethnographies centre personal accounts of the researcher's experiences with the intent of expanding understanding. (38,228) The way in which the experiences are collected and shared, and the level of detail provided in autoethnographic accounts, varies based on research methodology, methods, and goals. (228) For example, researchers may generate their autoethnographic data through journal entries, discussions with others, reflections based on memory, analysis of work completed, and/or dramatic performance. (228) Researchers must consider their methodology and methods when determining what data sources to utilize, to ensure all aspects of the research design align and complement one another. (38) In this thesis, I use personal reflection and analyses of work completed to generate the data as these sources align with our action-oriented methodology and methods and do not create additional burdens for the community with whom I am working.

72

There are five main characteristics of autoethnography: 1) critical reflexivity, 2) educative experiences, 3) privilege-penalty experiences, 4) ethical concerns, and 5) salient experiences assembled and shared. (228) Engaging in critical reflexivity requires one to challenge existing knowledge and understandings, in the context of new knowledge and experiences. (228) In line with critical reflexivity, autoethnographers must consider their own roles in creating their realities and analyze their responses to challenges as along a scale of resistance to complicity. (228) Researchers engaging in critical reflexivity must identify and consider educative experiences, or experiences that have transformed their awareness or perspective on a topic, to effectively analyze their understandings of the autoethnographic phenomenon. (228) Privilegepenalty experiences describe one's relation to and experiences of oppression systems such colonialism, ethnicity, socioeconomic status, and gender. (228) Experiencing privilege and penalty can occur simultaneously and are context-dependent, thus illustrating the nuanced approach autoethnographers must take when considering their "privilege[s] and penalt[ies] alongside the social forces they perceive, identify, and study in relation to themselves." (p.23, 228) Educative experiences and one's privileges and penalties are often discussed in a researcher's positionality statement; for self-narrative data generation, it can be valuable to incorporate these experiences and positionalities into the discussion to enrich context-specific discourse. (228)

Relational ethics are fundamental to autoethnographic strategies as the inquiry focuses on the researcher's experience *in relation* to others and the environment. (228) Autoethnographic studies must take a prudent and holistic approach to ethics, to ensure privacy for the people and communities involved is upheld and that the researcher's biases do not overwhelm the narrative and therefore jeopardize the study's credibility. (228) There are three ethical obligations autoethnographers must consider: 1) protecting the privacy and safety of those discussed in the autoethnographic work, 2) if and how to engage those discussed in the work through member checking, and 3) all possible interpretations and implications of the work. (228) Various researchers engaging in autoethnography have noted that to respect and affirm relational ethics, the study's overall intention must be prioritized over specific details' inclusion, particularly those of a sensitive nature that could reflect poorly on the community and therefore undermine the researcher's relationships. (228)

Salient experiences assembled and shared refers to how autoethnography data generation occurs and can be assessed. (228) Salient experiences can be defined as "memorable personal stories from one's life that can be supported by evidence from related critical social research literature." (p.26, 228) In other words, salient experiences form the foundation of the autoethnographic data. Situating these life narratives within a rigorous literature review supports autoethnography as a valid and rigorous research tool and draws attention to any knowledge gaps or inconsistencies that may exist in the literature. (228) Assemblage refers to the gathering and synthesizing of salient experiences and relevant literature to create a "rich, multilayered account[] of a particular time, place, or moment". (p.27, 228)

Autoethnography can be a challenging data generation strategy to employ as it requires a critical analysis of one's experiences and actions, thus making oneself vulnerable. (228) While reflections of one's experiences are necessary, they are insufficient alone; researchers wishing to engage in autoethnography must be reflexive by "being critically conscious while considering how they might be complicit" (p.17, 228) in the challenges they experience(d). Autoethnography has received critique from some researchers claiming that the strategy is "too subjective to meet the scrutiny of rigorous research." (p.27, 228) However, many researchers have rebutted this sentiment by reinforcing that because autoethnography necessitates a connection of the self to the cultural, it enables a discussion that "focus[es] outward on the sociocultural aspects of one's experiences, [thus] it is research." (p.41, 38)

Ethics

As my research relationship with the community stemmed from previous work as an intern with the community, I have engaged in relational ethics from the project's grassroots. Relational ethics can be described as ethics through the lens of relational accountability. (39,52) This approach to research ethics is especially important when working with Indigenous communities to ensure current and future research is conducted in a manner that affirms their sovereignty and is respectful of their culture, people, and land. (97) Both the *Aajiiqatigiingniq* and CBPAR methodologies necessitate ethics grounded in relationality and relational accountability. Relational ethics are also emphasized and required by the NRI and TCPS-2, however, their applicability extends well beyond REB approval and research licensing. (103)

The principles of relational ethics are grounded in relational accountability and include respect, reciprocity, and responsibility. (6,39,52) I have engaged in relational ethics by critically reflecting on my presence's impact on the community and influence in the project planning process. As a southern non-Indigenous person who is working within institutional research, it is my responsibility to ensure all research goals, activities, and outcomes are ethical for the community. I grew up and currently live in a markedly different sociopolitical environment than the project's setting; I cannot assume what I believe to be just, moral, and principled is ethical for this Inuit community. I have also practiced relational ethics by utilizing continual and conscientious reflexivity to evaluate my actions and motivations, encouraging community direction for all research activities, and learning from and applying community feedback. These efforts have functioned to ground this project's ethics in relational accountability. To facilitate further discussions and mutual understandings of relational accountability, I plan to establish terms of reference with community members involved in the research process and other institution-based researchers. These terms of reference will be collaboratively created by community members and university-affiliated researchers to ensure all parties understand their respective roles and responsibilities; this document will also provide guidance regarding relational accountability, ethical conduct, and conflict resolution. (98)

There are some autoethnographic-specific ethical considerations that deserve emphasis. (228) Namely that, while autoethnography strategies draw from a personal experience, these experiences are related to larger social and cultural phenomena which are often discussed and described in detail; these discussions, while intending to provide context for readers, can cause harm to the community or individuals involved if done so carelessly. (228) Thus, autoethnographers must take particular care to not identify the community or individuals involved that wish to remain private. (228) Given these ethical considerations that are grounded in relational accountability, throughout this thesis, I have omitted details and literature that could inadvertently identify the community. I have elected to not ask community members to engage in member checking for this thesis because: 1) I am not identifying the community or any individuals, 2) it would add a significant burden to community members without much benefit for them as individuals or the collective community, and 3) this thesis utilizes autoethnographic data from a researcher's perspective during the early stages of a CBPAR relationship. However, I

75

have engaged in member checking by discussing my experiences with my thesis committee who have been instrumental to this project's process and colleagues who are aware of the project's progression and activities. Knowledge sharing products that discuss research findings or the community's experiences will be collaboratively produced with the community, and therefore, community-based member checking will be inherent to the creation process. Researchers must also consider all possible interpretations and implications for their work. (228) While communities across Inuit Nunangat are incredibly diverse, many non-Indigenous people still view Inuit, and even Indigenous Peoples, as a homogenous group. (180) Given this, some readers may interpret my experience with one community as reflective of all communities within Nunavut or Inuit Nunangat. While this is not my intent, I have considered it as a possible interpretation and have therefore taken great care to ensure that I utilize a strengths-based approach throughout the thesis so that the research highlights the community's positive attributes. I have also highlighted the diversity within Inuit Nunangat throughout my thesis to prevent this interpretation.

Formal ethics requirements are important to obtain and can function as a 'checkpoint' for researchers and communities engaging in CB and/or PA research methods. (90) I completed mandatory research ethics training from the University of Alberta before any interaction with the community. As is common with CB and/or PA research projects, the process to obtain formal ethics approval from the University of Alberta and NRI was highly complex. About three months into my discussions with the community regarding a collaborative research project, I received the initial ethics approval from the University of Alberta. The NRI's process to obtain a research license was intricate and necessitated important discussions with the community about research goals, parameters, and processes. In June 2022, after five months of discussions with community and the NRI, I obtained a research license. In light of the project's changes, a new ethics approval from the University of Alberta's REB was required; we received our second ethics approval from the University of Alberta in June 2022. Ethics approval and research license numbers are not disclosed in this thesis as these documents are for the future data generation and analysis activities and could expose the community's identity.

Rigour

Relational accountability extends beyond ethics and plays a critical role in creating rigorous research. Rigour can be described as "demonstrating how and why (through methodology) the findings of a particular inquiry are worth paying attention to". (p.100, 38) Essentially, questions of rigour aim to establish that logic was applied when making decisions throughout the research process, and that "what we are writing came from our data and not a vision in the sky". (p.106, 38) Wilson notes Western concepts of rigour such as validity and reliability do not apply well to relational research as these require phenomena to be viewed in isolation, rather than as part of a network of relationships; authenticity and credibility are offered as alternatives to validity and reliability, while shifting the emphasis to the research's relationality. (39) In relational research, one should analyze the research's accuracy by asking how it is 'encircled' or fits within what is already known, rather than applying the Western technique of triangulation. (39) Assemblage is a similar technique described by Hughes and Pennington that utilizes multiple sources of knowledge (e.g., academic literature, government reports, member checking) to situate one's experiences within current understandings of reality. (228) Encircling or assemblage via literature reviews and discussions with people involved and/or aware of the phenomenon is the primary strategy this project employs to assess and ensure rigor. This rigour analysis technique also highlights any trends, gaps, or inconsistencies in the literature. (228)

This project also utilizes methodological coherence, theoretical thinking, and prolonged engagement to verify rigour. (38) Methodological coherence refers to all aspects of the research, including the project's ontology, epistemology, method, data generation strategies, and more, supporting and working well together. (38) As noted previously, the *Aajiiqatigiingniq* and CBPAR methodologies complement each other given their similar prioritization of relationships, community-led processes, and action-oriented approach. Autoethnographic data generation and analysis align with these methodologies because autoethnography views relational accountability as an essential component of the research process, promotes encircling or assembling findings, and fosters an 'outward-facing' discussion of personal experiences in the context of collective experiences or phenomena. Theoretical thinking and prolonged engagement are both inherent to the autoethnographic process CBPAR and *Aajiiqatigiingniq* methodologies. Theoretical thinking can be described as 'dwelling' with the data and viewing it "from a macro-micro perspective"

(p.110, 38); this careful approach to data analysis is essential for researchers to meaningfully engage in critical reflexivity throughout the analysis process. (38) Autoethnography moves beyond prolonged engagement, or "spending a considerable amount of time in the setting" (p.111, 38), by focusing the discussion on the researcher's experience in relation to the greater cultural and social context. Therefore, the researcher's personal connection to the data enables a rich discussion of the experience and possible implications.

In this chapter, I have described my approach to and utilization of CBPAR methods in conjunction with the Aajiiqatigiingniq Research Model, my use of autoethnographic data generation, and applied research ethics and rigour analysis techniques.

Chapter 4: Reflections of work completed during the initial phases of the Aajiiqatigiingniq Research Model

In this chapter I will narrate and reflect upon the work I have completed thus far in the project. Through this work, and the related discussions, I have fostered trusting relationships with community members and built shared understanding around community needs, research priorities, and established processes within the community. Thus, this work can be situated within the Aajiiqatigiingniq Research Model's first two phases, *piliriqatigiingniq* and *inuuqatigiitsiarniq*, or building community relationships and understanding, respectively.

Timeline

In May 2021, I was hired to work with the community as an intern. Working with multiple community members in the field of mental health, I developed educational resources, researched services currently available to community members, and explored ways to increase the community's capacity for mental health support. The relationships I formed during my internship laid the foundation for the research project discussed in this thesis.

In August 2021, I asked my internship supervisor if they thought the community would be interested in developing a research project with me; after discussions with multiple community members, we decided to proceed as they thought the community could benefit from this opportunity, and we began discussing potential ideas. In September 2021, I established my university advisory committee and began my literature review. I am fortunate to have an Inuk Elder originally from the community relation, we drafted a project proposal for Health Canada's Substance Use and Addictions Program (SUAP) grant that explored ways to foster cultural continuity and social connection among youth in the community. After REB approval and the initial NRI research license review, we presented our project proposal to the community's Council in February 2022. The Council was interested in the project and felt it explored an important topic; however, they voiced concerns over this proposed project's pace, and wanted to further explore community-led research processes and necessary safeguards to ensure research leads to positive change within the community while minimizing any harms.

Learning from this feedback, I revised the proposal to include an initial research phase that explores: 1) how to conduct CBPAR methods in a way that aligns with community goals and values, and 2) necessary safeguards throughout the research process to minimize risks and any potential harms. This proposal has not yet been officially accepted as the Council wishes to ensure key community stakeholders are supportive prior to finalizing their approval. Discussions of ways to include youth during the research and health promotion planning process are also ongoing.

Health Canada grant

In September 2021, my advisory team identified Health Canada's SUAP grant as a possible funding option for the research and health promotion project. I was the lead author on the grant application with support from my community supervisor and advisory team (see Appendix A, section i). As a Master's student who has taken graduate-level courses on public policy, health promotion, and community-based qualitative research methods, I was well-positioned and equipped to lead the writing process. My approach to the grant writing process was guided by conversations with community members and a community needs assessment that was conducted in 2018. I also conducted a literature review to learn about community-led health promotion initiatives, particularly in the Inuit context, and completed a policy and program scan to prevent program redundancy. I purposefully wrote the grant in general terms (e.g., 'programs to prevent harmful substance use among youth') to ensure that a community-led approach for program processes, activities, and goals could be achieved if funds were received. With the community's mayor, my community supervisor, and my advisory team, we collaboratively decided to submit the grant under the community's name so that funds could be held and controlled by the community. In March 2022, we learned that we were successful in our application and received \$172,000 from Health Canada.

Since March 2022, I have been involved in collaboratively finalizing the program's activities, timeline, and budget with community members (see Appendix A, *section ii and iii*). Health Canada requires detailed descriptions of all program activities, timelines, and justifications for every budget item. These details are incongruous with a community-based approach to research and health promotion, as activities usually develop organically depending on what is learned in

previous phases. Given these circumstances, I initiated discussions with Health Canada to explain the community's current circumstances and the project's approach, and learn how to align their reporting needs with the community's identified preferences for program processes. I also successfully advocated on behalf of the community to allow the program's budget to be modified given the community's circumstances and Health Canada's lengthy application analysis (see Appendix A, *section iv*). Following these interactions with Health Canada, I facilitated discussions with community members about the funding and reporting requirements, to ensure that they are aware of Health Canada's requirements and determine if they wish to proceed with the grant; once we had reached a consensus that they do wish to proceed with the funding, I facilitated discussions of strategies we could use to ensure that Health Canada's requirements are met and a community-led approach is maintained.

Community Council meetings

Throughout the project planning process, I have attended multiple Council meetings over virtual or teleconferencing software. Joining the meetings remotely has created some communication challenges as it can be hard to convey one's respect and humility through voice alone and the Councillors meet in-person, so discussion amongst them can be difficult to navigate. Additionally, similar to many rural and remote communities in Canada, the community's communication infrastructure often provides a weak or lagging connection.

Despite these challenges, the Council meetings have been instrumental to foster engagement with a diverse group of community members. The Councillors are nominated and elected by the community, thus they are often viewed as community representatives. Currently, all NRI-approved research projects must also be approved by the Council before research activities can begin. As noted previously, I presented our initial project's proposal at a Council meeting in February 2022. The Councillors collectively expressed interest in the project and felt it addressed an important topic to improve community wellbeing, but were concerned that the project's pace did not allow for a cautionary approach to data generation. Multiple Councillors shared stories of previous harm they have experienced from research projects and emphasized that safety during the research process was a principal concern. Learning from their stories, I transformed the proposal to include a research phase to first explore how to safely conduct research with the

community and ways to ensure the research process and outcomes are community-led and owned. The community had limited capacity for a new research project between April to June 2022 and many community members leave the community to spend time on the land in the summer. In July 2022, the new proposal was discussed at a Council meeting, but there were not enough Councillors present to make a decision. Thus, the new proposal was not voted on until September 2022.

I am fortunate to have an Inuk Elder on my advisory committee who is originally from the community and working in the field of community-based research and mental health. When I was discussing the Councillors' concerns and the resulting project's direction with my advisory committee, the Inuk Elder offered to write a letter of support for the project to explain why they believe community-based research is important; this letter was translated and offered in the spirit of relational accountability with the intent of creating an environment where the Councillors felt empowered and safe to make their decision regarding this project. In September 2022, I presented the new proposal and letter of support to the Council; they now agreed with the project's goals and processes, but wanted to ensure that other community stakeholders, including the mental health nurse, school principals, and education board, were in agreement before finalizing their approval. Currently, my main community relation is working to obtain letters of support from these stakeholders. While the project has not yet been approved, these community-led discussions around research priorities and health promotion initiatives have furthered discourses around community wellness planning.

These meetings were important relationship building opportunities. In the spirit of respect and in accordance with the NRI's requirements, I ensured that all documents presented to the Councillors were translated into the local Inuktut language. I also outlined the university-affiliated researchers' role in this project as following the community's direction and supporting research activities as requested by community members. I also re-iterated that terms of reference would be collaboratively drafted to address concerns related to community-held ownership and control over the research process, findings, and knowledge sharing products. During these meetings, I received multiple questions from the Councillors around my goals and underlying motivation to engage in this type of research project. These questions provided me with a valuable opportunity to reflect on my intentions and demonstrate my genuine desire to promote

the community's research and health promotion sovereignty. When responding to these questions, I aimed to demonstrate my humility and respect for the community's self-determination, but I found this nuanced communication challenging to achieve over a conference call.

Nunavut Research Institute Research License

An NRI research license is mandatory for all research occurring within Nunavut. The NRI's research license process aligns with the ITK's research strategy to promote Inuit research sovereignty and requires researchers to demonstrate their commitment to rigorous and ethical research. The research license application requires: 1) a non-technical project description that summarizes the research project's objectives, methods, data generation, benefits and risks, and knowledge sharing; 2) a detailed project description that describes the aspects discussed in the non-technical description in greater detail and provides additional information on the project's methodology, consent processes, data storage and accessibility, and biological sampling; 3) all consent forms that will be used in the project; 4) proof of community support and engagement during project planning; 5) a declaration of any needed assistance from Nunavut's Health Department; and 6) an REB approval from a research institute (e.g., university or other research institute) (see Appendix B). The non-technical summary and consent forms must also be translated into the local Inuktut language (see Appendix B, *section ii and iv*).

I submitted the original proposal to the NRI in January 2022 and received the Council's feedback on it in February 2022. The NRI's reviewers also commented on some of the project's strengths, namely its community-led approach and relevant topic, and their concerns which focused mainly on appropriate compensation for community members and proof of community engagement. As noted previously, we wrote a new research proposal based on what we had learned during the Council meeting and addressed the NRI reviewers' concerns; in summary, this new proposal narrowed the research study's scope to the community engagement phase and focused on further exploring the research process and necessary safeguards. We submitted a new application to the NRI for this proposal in March 2022. This new proposal was initially viewed by the NRI's reviewers as an extension of the first project and received criticism for its lack of response to the reviewers' comments on the initial project. Once the miscommunication was rectified and the reviewers understood this to be a new research license application, the NRI asked for further proof of community engagement to which we provided a letter of support from the mayor. In June 2022, the NRI approved our new application (see Appendix B, *section v*). While the NRI research license process was onerous at times, it also facilitated important discussions with the community regarding research priorities, processes, and goals.

Writing the research license application was a time-consuming process as many questions arose as I necessarily considered the project from a macro-micro perspective. When writing the first application, I applied and synthesized the knowledge I had gained as an intern, the information captured in the 2018 needs assessment, and the project ideas that had been collaboratively explored with my primary community relation and advisory team. In retrospect, this first proposal was based on some assumptions we had made about community priorities and wellness goals. The Councillors' feedback demonstrated how some of my underlying beliefs and assumptions had influenced the project and their valuable feedback functioned to align the project with the community's needs. Research, and therefore research design, is never objective; however, CBPAR projects, in contrast to projects that utilize traditional research methods, are guided by the community's values and goals rather than the researcher's. In the second proposal, I consciously reflected on my biases throughout the writing process and applied the Council's feedback. When writing the second application, I recognized that my interpretation of the Council's feedback is not necessarily what they meant to convey; in response to realization, I emphasized a community-led approach throughout the application to ensure that the research process can continue to be modified to meet the community's needs.

For the translations, my main community relation identified a community member who would be interested in doing some paid translation work. Working with a community member functioned to simultaneously bolster the community's engagement in the project and invest in the local economy. Upon reflection, I have come to realize the import of having a strong relationship with at least one community member who wishes and is able to contribute to research activities during the NRI licensing process. My primary community relation has been imperative for aspects of the research process that require in-person communication or knowledge of the community's dynamics. Further, as my primary community relation is a Councillor, they have acted as a liaison between myself and the rest of the Council; for example, they advocated for me to be a

delegate at the Council meetings and ensured the project's inclusion on the Council meetings' agendas.

University of Alberta Ethics Approval

The University of Alberta's REB affirms and holds researchers to the TCPS-2's standards and guidelines. Among other aspects of the research design, the University of Alberta's ethics application requires researchers to discuss research objectives, participant recruitment and data generation strategies, consent processes, data storage and accessibility protocols, and planned knowledge sharing and all other uses for the research. The ethics application has a specific section for researchers working with Indigenous Peoples to discuss community engagement in the planning process and the project's cultural sensitivity.

In October 2021, I began my first REB application. Similar to the NRI research license process, I found this initial application helpful for me to think through the proposed project's objectives and protocols. While many of the questions were challenging to answer given our communityled approach, these questions highlighted the various means by which community engagement and direction would be essential. My initial REB application was shuffled between various reviewers due to the project's methods and multiple reviewers' COVID-19-related leaves of absence. This shuffling resulted in my application falling between the institutional cracks and was not acknowledged until I followed up six weeks later. After some discussion regarding the proposed consent forms and participant recruitment strategies, I received my first ethics approval in January 2022 (see Appendix B, section vi). In April 2022, after re-writing the NRI's research license application, I updated the REB to inform them of the new research focus and methods and reasoning for the changes. In response to the changes, they required me to submit an entirely new ethics application, despite many sections being identical and the new project being an expanded-upon version of the community engagement planned for the first project. Repeating the ethics process was highly tedious and undermined my ability to prioritize community relationship building during this time period. In July 2022, I received the ethics approval for my second application (see Appendix B, section vii).

The University of Alberta's REB has taken efforts to ameliorate many of the challenges researchers engaging in CBPAR methods experience. During my ethics reviews, they allowed me to state that I will update the REB on research activities as community-led planning progresses. The University of Alberta's REB has also aimed to effectively hold researchers working with Indigenous communities to the TCPS-2's standards by asking detailed questions related to research methodology, community risks and benefits, proof of cultural sensitivity in the research plan, and requiring researchers to show community support prior to research initiation. However, they undermined the CBPAR process by requiring a detailed timeline with all research activities and the project's cultural sensitivity by necessitating lengthy consent forms and initially resisting oral consent processes.

In this chapter, I have described and reflected upon the work I have completed thus far in the engagement phase of the project. This work, and the related conversations I have had with community members, has contributed to the building of trusting relationships and shared understanding around community needs, research goals, and established community processes. Given my work's scope and content, it can be located within the Aajiiqatigiingniq Research Model's first two phases that explore building community relationships and understanding, or *piliriqatigiingniq* and *inuuqatigiitsiarniq*, respectively.

Chapter 5: Discussion and next steps

In this section, I use critical reflection to analyze and encircle my experiences beginning a CBPAR project with an Inuit community and critical reflexivity to analyze and position myself within the various power structures within which these experiences occurred. Ng *et al.*'s comparison of critical reflection and reflexivity is useful to delineate these similar, yet unique processes. (231) Critical reflection can be defined as "A process of examining assumptions (i.e., individual and societal beliefs and values), power relations, and how these assumptions and relations shape practice." (p.1123, 231) Additionally, critical reflections analyze one's held beliefs and experiences and the associated outcomes inform one's everyday practice. (231)

There are a myriad of definitions of critical reflexivity "arguably mirroring the concept of reflexivity itself." (p. 529, 232) Ng *et al.*'s. definition of critical reflexivity aligns with my approach and method,

A process of recognizing one's own position in the world in order both to better understand the limitations of one's own knowing and to better appreciate the social realities of others. (p.1123, 231)

They also note that critical reflexivity analyzes power relations and societal norms, structures, and discourses and that the related findings lead to systems-level change. (231)

Roles I have held during this research project planning process

Moving the project through colonial structures

One of the primary roles I have filled during the research planning process has been to move the project through colonial structures, namely the University of Alberta's REB and Health Canada, while affirming and promoting the community's research sovereignty. As asserted by the Aajiiqatigiingniq Research Model, establishing shared understandings and trusting relationships built on "respectful, open communication" (65) are critical for research projects' success. As a non-Indigenous researcher affiliated with a research institution, I am an insider to various colonial structures that have informed my beliefs and biases. It is important to acknowledge that I have generally benefited from these colonial structures that have caused and continue to inflict harm on Inuit communities. Throughout my engagement with the community, but particularly when liaising with colonial structures, I have reflected on my biases that are a result of my

previous experiences with these institutions and taken efforts to align my actions and beliefs with the community's needs and goals. The Aajiiqatigiingniq Research Model emphasizes that "the *piliriqatigiingniq* relationship is grounded in equal voice, negotiation and open view sharing, in exploration that sets goals and guidelines for how a task progresses and is viewed as successful." (65) Thus, critical reflections of my own biases and beliefs have and will continue to be essential to ethically move this project through colonial structures on behalf of the community.

As noted previously, REBs' reviews often do not align well with community-led research methods such as CBPAR. Challenges communities and researchers experience during REB review include required specification of all research activities and sole consideration for the research project's impact on the individual, rather than the community. (38,89,91,92) The University of Alberta's REB has taken efforts to ameliorate these challenges by allowing applicants to update the REB on research activities and requiring proof of community support prior to research initiation with Indigenous communities. However, as noted by Stiegman and Castleden, REB processes are time-consuming, thus updating the REB can be an onerous process that often undermines the community's self-determined process or momentum for a research project. (90) REBs are often "steeped in a positivist tradition" (p.3, 90), resulting in the lens with which research projects are viewed, and standards to which research projects are held, often contradicting Indigenous research methodologies and therefore functioning to undermine Indigenous research sovereignty. The REB-related burdens I experienced during the two separate review processes were not simply frustrating for me, they also undermined my ability to invest time and energy into community relationships; this REB-related phenomenon has been noted by other researchers as significantly impairing their ability to effectively engage with the community. (90)

As laid out in the TCPS-2, REBs are not positioned to "override or replace ethical guidance offered by Aboriginal peoples themselves" (103 as cited in p.1, 90) (103), thus, there appears to be a prevalent disconnect in REB policy and practice. As noted previously, REB review represents an opportunity for community empowerment by evaluating the strength of,

... researcher-community partnerships and structures of mutual accountability that have been established, while acknowledging the jurisdiction of the nation in question and deferring to their authority... (p.5, 90)

For REBs to realize their full potential and act as an empowering process for Indigenous communities and researchers alike, researchers have called for review boards to shift their focus to the "four Rs' of Indigenous research: respect, reciprocity, relevance, and responsibility." (p. 5, 90)

Community engagement and support are based on nuanced relationships between the researcher, research project, and community. Engagement with a diverse group of community members during project planning has been widely recommended for researchers working with Indigenous communities. (39,40,78,98,103,108) As communities can be viewed as a network of complex relationships within a diverse group of people, proof of community engagement and support cannot be wholly captured in a letter of support from an individual. (39,43,102) During REB review and the NRI's review process, I provided a letter of support from the mayor to demonstrate community engagement; I sought a letter from the mayor because they are relationally accountable to the Council and engage in collaborative decision-making processes with them. However, support letters from an individual or small group of community members do not necessarily reflect the collective community's values and priorities and could function to undermine the community's processes and protocols. Therefore, consistent with the 'four Rs', REBs must holistically and critically examine researchers' community engagement and support to effectively evaluate researchers' genuine collaboration with and support from Indigenous communities. (84,90) Additionally, alternative mechanisms to formal letters of support to demonstrate community engagement that align with Indigenous values and traditions could be collaboratively developed with Indigenous Peoples and implemented by REBs, such as submissions of email records or facilitating phone or video calls. (84)

Communities, especially those which are marginalized and experiencing challenges, often have limited human resources for research projects. (110) Therefore, researchers must recognize a community's capacity and desire to engage on a research project and ensure they are not adding to the community's burdens. (110) When initiating a research project, it is important that researchers transparently discuss the community's capacity and desired level of involvement for the project. (98) The extent of community engagement lies on a spectrum with one extreme representing communities that wish to be involved in the minutia of the research process, and the other extreme representing communities that wish to establish their overarching goals and

89

processes with a researcher who then independently manages much of the research process and updates the community regularly with the project's progress. (98,110) Communities' level of desired engagement often varies temporally and can be dependent on the activity at hand or other participating organizations. (98) In the early planning stages, I learned that the community with whom I am working, while interested in proceeding with the research project, had many ongoing priorities and would prefer a lower level of community engagement during the planning phases. Based on this knowledge, I discussed project goals, overall project processes, and reporting expectations with community members to ensure that I act in a way that aligns with community needs and values. These discussions have influenced my liaising with the University of Alberta's REB and Health Canada and resulted in me leading the ethics and grant application process and seeking opportunities for research self-determination that are not burdensome for the community. I have sought regular feedback on my level of engagement and actions to tailor my approach to best meet the community's needs. For example, initially I did not or steer the meetings' conversations or send community members meeting agendas or minutes as I did not want to appear authoritative; however, based on feedback, I learned that most community members would appreciate meeting agendas and minutes and a more directive approach to project discussions to enable efficient and productive meetings. This example of action correction aligns with the Aajiiqatigiingniq Research Model as I have sought to establish communication strategies that respect the community members' time and processes. (65) Additionally, my understanding of the community's capacity led me to omit community member checking for this thesis.

Secured funding

I was the lead author on a Health Canada grant application that my advisory team had identified. My education and positioning within a research institute equipped me well to lead the grant application process. Researchers engaging in CB and/or PA research are often relied upon to secure funding due to their institutional affiliations and understanding of the complex, and often laborious, grant application processes. (64,84,90,108) Despite multiple calls for funding institutions to widen the eligibility criteria for recipients, grants are often only accessible to those affiliated with a research institution; thus, researchers affiliated with recognized research institutions are often depended upon to obtain financial resources for research projects. (64,90,97) Further, funding institutions' applications' onerous formatting and requirements creates additional barriers for organizations and individuals outside of the research world. When writing the grant application and communicating with Health Canada, my approach was guided by my intention to respectfully promote the community's research sovereignty by increasing their funding access and control. My objective to increase the community's financial sovereignty manifested primarily by: 1) writing the grant in general terms, to optimize the community's control over the program planning and delivery, and 2) initiating discussions around the community's capacity to hold the grant funding and promoting this funding model.

The general diction I used throughout the grant application has created a flexible funding scaffold that allows the community to utilize the funds for youth health promotion however they see fit. In the grant application, I framed the project around a general goal, 'for research and programming to prevent substance use among youth'. I intentionally employed vague terms and phrasing to enable the community to direct and/or lead all research and health promotion activities, if desired. In this way, my approach to the grant writing process aligns with the Aajiiqatigiingniq Research Model as it fosters trusting relationship building and facilitates total community direction during later phases. (65) This approach also aligns with the ITK's NISR (2018) priority to increase financial sovereignty in research projects with Inuit communities. (97)

Within the health research sphere, the importance and impact of community-led research and health promotion programs has been well-established. (97,102,108) However, funding agencies often undermine researchers' ability to effectively engage in CB and/or PA research projects by requiring grant applicants to provide detailed timelines, budgets, and research and/or program activity plans and grant recipients to provide an even higher level of detail for program plans prior to funding-mobilization. (90,110) These reporting requirements, similar to those seen in REB reviews, are rooted in colonial structures and processes. (90,110) However, there are strategies researchers can employ to work around these reporting requirements, including: 1) writing the application in vague terms to allow the community to determine how best to utilize the funds, 2) advocating on the community's behalf by asking the funding body if and what changes to the budget or program activities are allowed after funds have been approved, and 3) collaborating closely with community members during the writing process. Each of the aforementioned strategies has their own benefits and drawbacks and the researcher and

91

community must collaboratively decide which strategy or strategies best fit the funding institution's requirements and the community's needs and capacity. (64) In our situation, Health Canada was able to accept programming goals over explicit activity descriptions and consider budget adjustments after funds had been approved. However, Health Canada is unwilling to allow budget adjustments after funds have been dispersed, thus, I am continuing to work closely with the community members to finalize the budget.

Grant administration, reporting, and management is highly time-consuming, posing additional challenges for communities wishing to hold funding; in our application, we addressed this by including a stipend in the budget for a community member to administer and manage the grant. This approach not only reinforces the community's direction over the research process by situating community members at the helm of the financial resources necessary for research activities to occur, it also invests into the community's economy. In traditional and often many CB and/or PA research projects, the researchers hold the funds and therefore have greater control over the research process; financial sovereignty functions to flip the traditional researcher-community power dynamic, thus empowering the community to self-determine and address their research and health promotion needs. (108) Community-held financial control is thought to have contributed to another Inuit community's research project's success by enabling a genuine community-led approach that facilitated relevant activities and resulted in community members' pride over the research process and outcomes. (108) While this project's success cannot yet be evaluated, the community's control over the funding has promoted trust between myself and community members, therefore strengthening relationships.

Liaising with other institutions

Thus far in the project, my interactions have been largely limited to the University of Alberta's REB, Health Canada, and the NRI. My interactions with the University of Alberta's REB and Health Canada have been described in detail in the previous two sub-sections and Chapter four. My approach to obtaining a research license from the NRI has been largely guided by my understanding of the community's desired level of engagement during early planning phases. In my experience, the NRI's research license process effectively evaluated the strength of my relationship with the community by asking for additional proof of community engagement and support. Importantly, they wanted to see that the community is driving the research process, and

that the research team is cross-culturally competent and that local control and ownership over the data is established. (90) In this way, the NRI's feedback facilitated important discussions among my advisory committee and the community's Council regarding the research process and ways that Inuit cultural values and IQ principles can be incorporated into the research design. Further, the NRI's process aligns with the Aajiiqatigiingniq Research Model as it necessitates researchers to carefully engage in relationship and understanding building with the community, prior to approval. (65)

The NRI was the only organization with whom I engaged that required documents for the community to be translated into the local language. Translation respects the community's culture, language, and language sovereignty and ensures project perspectives can be heard from all community members, especially Elders. (233) Further, community-based translations offer an opportunity for researchers to engage with more community members and invest in the local economy. Thus far in the project, translating the non-technical project description, consent forms, and project briefings has been a mechanism I have utilized to demonstrate my commitment to relational ethics.

Mutually beneficial relationships between myself, the NRI, and the community's Council required a strong community relation who could advocate for the project and myself. Other researchers initiating a research relationship have cited the importance of strong community relations as communities will understandably have many questions and concerns at the beginning of a research project. (39,40,43,64,84,98,108) As discussed previously, my primary community relation, who is a member of the Council, was instrumental during the NRI research license process as they were able to: 1) bridge the relational accountability between myself and the Council due to their positive relationships with both parties, 2) promote the project's discussion during Council meetings that often have lengthy agendas with competing priorities, and 3) identify a community member who could provide translation work. The importance of strong community relations is widely acknowledged to be accentuated for researchers working remotely as some planning work is necessarily or preferentially conducted in-person. (6,84,234) In this regard, my primary community relation has supported me by garnering community members' interest in the project through casual conversation and liaising with stakeholders, including the education board and school principals.

Future liaisons with regional or local organizations will be guided by a similar nuanced approach to community engagement that is grounded in the Aajiiqatigiingniq Research Model and based on community-expressed desire and/or capacity.

Literature review

Literature reviews are important for research projects to encircle findings and make rigorous evidence-based recommendations. (39,45,97) As a graduate student in public health, I have the resources and formal education to effectively scan the literature and identify, appraise, and analyze relevant sources. In an ideal situation, community members would lead the literature review process as they are best informed on how to relate current knowledge to the community's context. (43,45,97) However, for this project, I have led this important task as there are no community members currently able to participate. To effectively conduct the literature review and situate other studies' findings, I have applied knowledge that community members have shared with me. This knowledge has most often been shared through organic, "iterative discussions [that] lead to relational consensus building around the research itself". (65) Throughout the literature review process, I have attempted to ensure my work aligns with the Aajiiqatigiingniq Research Model by laying the foundation for evidence-based "collective decision[s] about how to apply ... [the] research" (65) in later research phases.

The community's literature review capacity stems from colonial policies and practices that have simultaneously disrupted Inuit ways of sharing knowledge and created barriers to Eurocentric knowledge systems. (9,13,99,200) These historical and ongoing policies and practices contribute to non-Indigenous people, particularly those of European descent, being over-represented in research spheres. (40,43,45,80) However, the prevalence of Indigenous researchers in Canada is growing which can be attributed to Indigenous Peoples' resiliency and strength and self-determined education initiatives' success. (45,97) Recommendations to increase research capacity in Inuit Nunangat, including the establishment of a university, would further promote Inuit communities' ability to fully engage in and lead research projects. (97) Intersectoral, evidence-based, Indigenous-led policy-scale interventions are needed to address the underlying causes of Indigenous Peoples' underrepresentation in research. (97)

I, a white, non-Indigenous person from southern Canada, leading the literature review reinforces a colonial power dynamic between myself and the community where I hold knowledge to which the community does not have full access. (43,45,84) While I acknowledge this imbalance in perspectives, it is also important that I acknowledge that I have benefitted from the structures and systems that oppress Indigenous Peoples and undermine their self-determination. To minimize the impact of this toxic power dynamic and build research capacity within the community, I plan to create accessible resources (e.g., translating article summaries, infographics, and/or oral presentations) explaining the literature review process and findings. (43,102) I will also attempt to invert this power dynamic by employing innovative strategies to engage community members during literature appraisal, but there is currently limited research on this topic. These strategies will be collaboratively determined with community members, based on their interest and/or capacity, and other relevant organizations such as the QHRC.

Critical reflections on challenges and lessons learned

The primary challenges I experienced during these early phases of a CBPAR project centred around institution-imposed timelines and financial pressures. As a full-time, thesis-based Master's student, I am required to complete my degree within four years; this timeline requirement contradicts community-led research processes as many community-based research projects are in relationship building and project planning phases for several years before any data generation occurs. (43,84) Graduate students are required to pay tuition throughout their degree, regardless of taking classes, adding significant pressure to financial burdens that many students already experience; this added pressure encourages students to complete their degrees in a timely manner that is not conducive to community-led approaches. Further, grants and scholarships available to Canadian graduate students working with communities during early research phases are currently limited. (84,90) The structure and reporting requirements for most graduate funding are based on traditional research methods. (43,84) While data generation is often one of the most resource-intensive phases for traditional research methods, community-based research can be costly during the planning phases as community members must be compensated for their time and knowledge, translations are often needed, and researchers may be required to travel to the community. Thus, grants that are well-suited for traditional research methods often do not align with community-based research timelines or methods. (43,84) While our Health Canada grant

application was successful, the community must finalize the budget and activities before funds can be mobilized; in the meantime, I have personally paid over \$1000 for translations to enable project progression. I am privileged to be positioned to afford this important expense; however, it is important to acknowledge that many students would not have sufficient savings to financially contribute to this extent, thus preventing these projects from proceeding. Timeline accommodations and greater funding opportunities for graduate students engaging in the planning phase(s) of CB and/or PA research projects are needed to facilitate and encourage students' involvement in this important research field. (43,84) Further, sustainability and longevity are defining features of many CB and/or PA research projects, therefore, increasing funding for researchers early in their careers to build these relationships can be seen as a longterm investment in community wellbeing.

These fiscal and timeline-related pressures are not unique to graduate students as wellestablished researchers doing CB and/or PA research with Indigenous communities have expressed similar challenges,

...financial services administrators ... 'need to be more aware that the process of doing [Indigenous] research is not cut and dry like a survey or a quick interview. It's much more engaging and it's much more involving the participants.' (p.10, 110) These prevalent experiences among students and researchers alike demonstrate that research and funding institutions must revise their policies to increase the accessibility and feasibility of community-led research methods. Additionally, Indigenous representation and activist organizations like the ITK, QHRC, First Nations Information Governance Centre have necessitated community-led approaches to research with Indigenous communities, thus further emphasizing the need for institutional policy reform in favour of CB and/or PA research methods. (97,102,235)

While navigating these institution-imposed challenges, I have learned the importance of preserving respect for community timelines and processes by not transferring these pressures to the community. These boundaries can be challenging to achieve and maintain as the institution-imposed pressures have substantial ramifications and it can be tempting to undermine the community-led process by imposing actions or decisions that quickly and/or easily meet the institutions' needs. However, in the spirit of relational accountability, it is important to uphold

these trusting relationships and find alternative solutions. (43) This idea can be extended beyond the singular-project ethics scale,

If researchers and their mentors (Indigenous and non-Indigenous) don't advocate for the time and funding required to do decolonizing research, it will do little to decolonize the university. (p.149, 43)

While the work described in this thesis cannot be described as decolonizing, the challenges I experienced are common to researchers engaging in decolonization work and my responses to these pressures functioned to affirm and promote institutional changes that align with decolonizing agendas.

Often, it can be helpful to share institutional challenges with community members to enable collaborative decision-making when seeking solutions; however, researchers should approach these conversations with caution to ensure that they do not place inadvertent pressures on the community, resulting in solutions that are oriented around the institution's or researcher's needs, rather than the community's. (43,84) In this situation, I was able to mobilize my own funds to advance the project and minimize the financial pressures by basing my thesis on the relationship building and planning phases of a CBPAR project. As noted previously, not all scholars wishing to engage in CB and/or PA research have these opportunities, thus, policy changes are needed to address the underlying causes of these challenges. (43,84)

Strategies I have learned to garner community engagement and build research capacity

Throughout this experiential learning process, I have gained and refined skills and strategies to foster community engagement and research capacity. These relationships lay the foundation for all subsequent research activities and outcomes, therefore

The most critical [research stage] is the first ... which centres around coming alongside participants to build the trusted relationship and shared understandings of the contexts. (65)

Primarily, these skills and strategies are rooted in respect and relational accountability. Wilson *et al.* lists seven questions they asked themselves throughout their research process with an Inuit community to ensure they are acting in ways that affirm relational accountability and align with community-identified goals. (43) Three questions to which I have found myself routinely coming

back: 1) "What am I suggesting? Is it based on a western or decolonizing research perspective?", 2) "What skills do I bring that can support community research needs so I can give back?", and 3) "Am I getting caught up in southern timelines and deliverables and forgetting that it's not about the results, it's about the process?" (p.149, 43) These questions, along with others, have pushed me to critically reflect on my goals and motivations throughout this relationship-building and planning process.

As discussed previously, engaging a diverse group of community members to lead and direct research activities is crucial to ensure the project aligns with the collective community's needs and priorities. (39,40,102,103) As an outsider to the community who has worked remotely, this goal has been a learning opportunity for me as I have had to rely heavily on my community relations. This community's dynamics, as with all communities, are complex and storied; thus, building diverse connections, primarily through other community members, has created both unique relational opportunities and challenges. (84) Ensuring a diversity of perspectives are heard and included will be an ongoing priority for the project and strategies to better engage a representative group of community members will continue to be discussed. I plan to respectfully promote equitable research participation by considering current collaboration opportunities' characteristics (e.g., time commitment, primary language utilized, location, and/or meeting times) and collaboratively determining alternative options to increase project accessibility for all. (43,98,105,110) Additionally, financial compensation and other incentives will need to be further explored to ensure they appropriately reflect community members' contributed time, knowledge, skills, and/or resources. (98,102,103)

Accessible research and knowledge sharing materials are widely acknowledged to be essential for robust community engagement when utilizing community-led research methods. (43,45,98,101,102) Strategies we have used thus far to make the research planning materials accessible include translating all documents, using plain language, bullet points, and tables to display information, and favouring oral communication over written when discussing research goals and approach. Future strategies we could employ to equitably share knowledge about the research project's progress and findings include the creation of infographics and other printed visual resources, in-person and radio presentations, and/or participatory data generation techniques such as photovoice or videos. (43,98,227) Consistent with a community-led approach,
these strategies will need to be collaboratively discussed with community members to ensure they are effectively accessible for all. (45,98,101,102)

Effective communication is critical to encourage community engagement and build trusting relationships. (39,43,84) I have routinely sought feedback on my written and oral communication style to ensure it meets the community members' needs and preferences. For example, I have modified my oral communication style to speak slower, especially when there is an interpreter present, as I received feedback that my naturally fast speech was challenging for some community members to understand. In alignment with the Aajiiqatigiingniq Research Model's inuuqatigiitsiarniq phase, or building shared understanding, I have also engaged in ongoing discussions of the community's expectations for me as a researcher and project partner (65); these conversations will guide the collaborative drafting of our terms of reference that describe all parties' roles and responsibilities. (102) Based on community-led research principles and my experience of the research planning process thus far, I am planning to take a flexible and evolving approach to my involvement in the project. This strategy will allow me to modify my contributions based on the community's needs and preferences which may vary seasonally as community members go on the land, and temporally as community capacity builds. (43,98,102) This holistic and adaptable approach to my participation is rooted in intentions to increase the community's research and health promotion sovereignty.

In accordance with community-led research processes and relational accountability, I have learned the importance of not speaking *for* the community, unless they have explicitly asked me to do so. (43,110) This can be a challenging balancing act as I do not want to overburden community members with questions and research engagement, while aiming to uphold community-led processes and research self-determination. Upon critical reflection, I have balanced these priorities and needs by utilizing a nuanced approach grounded in relational accountability and humility to determine if, when, and how to act on behalf of the community. Depending on the circumstances, my actions have been determined by conversations I have had with community members and/or existing knowledge about the community's needs and current situation. (110) For example, when communicating with Health Canada, I have not sought engagement for every meeting or email, as that would be burdensome to the community based on their stated current capacity. However, if I were to reach out to a new partner or gain new information from an existing partner, I would seek collaboration before acting. Obtaining regular feedback on my approach and actions has been especially important when acting or speaking on behalf of the community. (110) As in any relationship, there are times I have acted in ways that are incongruous with the community's goals or priorities due to miscommunication and/or underlying biases on my behalf; in these situations, I have found it paramount to respond with humility and genuine remorse and demonstrate a desire to rectify the action(s) and collaboratively discuss strategies to prevent future wrongdoings.

The aforementioned strategies I have learned to encourage community engagement and build research capacity align with the research guideline themes discussed in the Literature Review. The ITK's NISR asserts that research with Inuit communities must function to promote research sovereignty and that all other policies or approaches can be seen as contributing factors to this overarching goal. (97) Overall, I have aimed to promote research sovereignty by deferring to community members' decisions and upholding community-identified priorities and goals. In alignment with the Aajiiqatigiingniq Research Model, my approach to facilitate engagement has been guided by my relationships with the community, thus affirming my relational accountability which is the keystone of Inuit research ethics. (45,65) As discussed in this chapter's previous sub-sections, transparent community-held control over project finances and knowledge generated and/or shared has enabled more trusting relationships between myself and community members. Lastly, my flexible approach to the extent and mode of my involvement promotes capacity building as it creates organic and ongoing opportunities for community members to utilize new skills and/or knowledge as the project evolves.

Impact of COVID-19 on community-based participatory action research

Prior to the COVID-19 pandemic, the community with whom I am working already experienced barriers to quality and culturally-appropriate healthcare including a lack of an in-community physician and limited access to other in-community healthcare services; the COVID-19 pandemic highlighted and amplified these healthcare-related challenges. Due to the risk of COVID-19 transmission, I have collaborated with the community entirely via remote interaction. Time spent in communities has traditionally been critical to forming trusting relationships

between researchers and community, as noted by Wilson *et al.* (2020), "Understanding from the outset that I needed to prioritize time and relationships in [the community] ..." (p.148, 43) However, I, like many researchers, had to adapt my methods and approach to a virtual format given the pandemic's restrictions and community's concerns for outsiders transmitting the virus.

Remote collaboration has caused me to rely more heavily on my primary community relation. This dynamic has created unique collaboration opportunities and at times facilitated more trusting relationships due to my inclusion in the community's relational accountability web; however, it has also added a significant burden to my community relation's workload and created barriers for me to form relationships with other community members, independent of my community relation. Many researchers engaging in CB and/or PA research projects have shared similar impacts of the COVID-19 pandemic on their community relationships. Hunter *et al.* notes that, "the shift from face-to-face engagement to virtual recruitment and relationship building hit roadblocks in many AI/AN [American Indian and Alaska Natives] communities." (p. 188, 234) This sentiment is echoed by Valdez and Gubrium who note that while there are conveniences associated with remote collaboration, "online-based CBPR removes a sense of solidarity that is inherent to attending meetings ... the online format did not facilitate the desired relationship-building effect." (p.7, 236) Valdez and Gubrium's identified conveniences are not generalizable to this project's context as they were working with an urban community and therefore did not experience similar communication infrastructure challenges. (236)

Similar to many communities in northern Canada, the community with whom I am working regularly experiences internet and/or telephone service outages or impairments. My remote collaboration with this community has illustrated to me the wide-ranging impacts and degree of challenges that communities experience due to poor communication infrastructure. Across Inuit Nunangat, the internet "tends to be slow and unreliable, and is frequently jeopardized by weather and technical difficulties". (p.2, 237) Currently, only one community in Inuit Nunangat has access to the federal government's target broadband connectivity and the current maximum internet speeds in Nunavut are less than half of the government's target. (237) There are a myriad of reasons for this poor level of connectivity, including remote communities and sparse population. (237) However, other circumpolar countries such as Norway, Sweden, and Finland have access to high-speed internet nationwide as their governments prioritize this service. (237)

The Canadian federal government must continue to improve affordability and access to highspeed internet in Inuit Nunangat, especially in the post-pandemic era of increasing virtual health and justice services and online business opportunities. (237)

Next steps

The project will continue to iteratively progress through the Aajiiqatigiingniq Research Model phases. My involvement in this project is guided by Wilson *et al.*'s (2020) Sikumiut model that is grounded in motivations to increase the community's sovereignty. (43) Thus, my participation will be entirely determined by community members' needs and preferences. If the community wishes to proceed with the project and my engagement, I will continue to support the community in the mobilization of Health Canada's funding to implement youth programming in the summer and fall of 2023. I will also offer to continue supporting the research planning process by participating in discussions of research methods that meet the community's needs and collaboratively developing experiential learning opportunities and workshops to strengthen the community's research capacity. Currently, discussions around research that explores ways to prevent and address harmful substance use have paused to respect the Council-stated desire to explore safe research methods *prior* to any other research. On a personal level, I will continue to reflect upon my role and duty as a fifth-generation settler to support and promote Indigenous Peoples' sovereignty and the ways in which I can meaningfully engage in and contribute to decolonizing research.

Conclusion

This thesis captures an academic perspective on the early stages of a CPBAR project in the context of a larger health promotion project with an Inuit community. In Chapters one and two, I positioned myself and my relations within the work, described my research approach which is grounded in the Aajiiqatigiingniq Research Model and utilizes CBPAR methods, analyzed and synthesized evidence-based best practices for researchers working with Inuit communities, described the relationship-building phases, and strategies researchers affiliated with institutions can employ to create shared knowledge with communities. In Chapter three, I detail my CBPAR methods which are rooted in the *Aajiiqatigiingniq* research methodology, describe

autoethnographic data generation, and rigour and ethics analysis strategies employed. In Chapter four, I narrate and reflect upon the ways in which I have furthered my relationships with the community and advanced the project's progression. In Chapter five, I situate my experiences within the Aajiiqatigiingniq Research Model and the literature to contextualize findings and highlight any knowledge gaps.

This work emphasizes the complexities of CB and/or PA research methods and the intersectional, nuanced approach that is required to build trusting and reciprocal relationships in the context of wellness research with Inuit communities. This work can be used to inform both the health promotion project planned with this Inuit community and relationship building with other Inuit communities; some aspects of this work can be applied to CB and/or PA research methods with Indigenous communities in the international context.

References

- 1. Erica-Irene A. Daes. Indigenous peoples and their relationship to land. United Nations Economic and Social Council; 2000.
- 2. George CT. Decolonize, then Indigenize: Critical insights on decolonizing education and Indigenous resurgence in Canada. Antistasis. 2019;9(1):73–95.
- Chávez V, Duran B, Baker QE, Avila MM, Wallerstein N. The Dance of Race and Privilege in CBPR. 2008 [cited 2022 Oct 6]; Available from: http://rgdoi.net/10.13140/RG.2.2.20860.95367
- Brubacher LJ, Dewey CE, Tatty N, Healey Akearok GK, Cunsolo A, Humphries S, et al. "Sewing Is Part of Our Tradition": A Case Study of Sewing as a Strategy for Arts-Based Inquiry in Health Research With Inuit Women. Qual Health Res. 2021 Dec;31(14):2602– 16.
- 5. Marecic CJ. Nunavut Territory: Aboriginal Governing in the Canadian Regime of Governance. Am Indian Law Rev. 2000;24(2):275.
- 6. Horlick S. The Impact of the COVID-19 Pandemic on Food Security and Food Sovereignty in Nunavut Communities [Master's of Science in Global Health]. University of Alberta; 2021.
- 7. Egeland GM, Charbonneau-Roberts G, Kuluguqtuq J, Kilabuk J, Okalik L, Soueida R, et al. Back to the future: using traditional food and knowledge to promote a healthy future among Inuit. In: Indigenous Peoples' food systems: the many dimensions of culture, diversity and environment for nutrition and health. Rome: Food and Agriculture Organization of the United Nations, Centre for Indigenous Peoples' Nutrition and Environment; 2009. p. 9–22.
- 8. Searles E (Ned). To sell or not to sell: Country food markets and Inuit identity in Nunavut. Food Foodways. 2016 Oct;24(3–4):194–212.
- Qikiqtani Truth Commission, Qikiqtani Inuit Association, Canada, Canadian Heritage. QTC Final Report: Achieving Saimaqatigiingniq [Internet]. 2013 [cited 2022 Oct 14]. Available from: https://www.qtcommission.ca/sites/default/files/public/thematic_reports/thematic_reports_e nglish_final_report.pdf
- Crawford A. "The trauma experienced by generations past having an effect in their descendants": Narrative and historical trauma among Inuit in Nunavut, Canada. Transcult Psychiatry. 2014 Jun;51(3):339–69.
- 11. Kluane A, Mackay T, White M. Integrating traditional practices into Inuit Mental Wellness Programs [Internet]. Walter & Duncan Gordon Foundation; 2015 Sep. Available from: https://gordonfoundation.ca/resource/integrating-traditional-practices-into-inuit-mentalwellness-programs/

- 12. Looking Forward, Looking Back: Report of the Royal Commission on Aboriginal Peoples [Internet]. Ottawa, Canada: Royal Commission on Aboriginal Peoples; 1996 Oct. Available from: https://www.bac-lac.gc.ca/eng/discover/aboriginal-heritage/royal-commission-aboriginal-peoples/Pages/final-report.aspx
- 13. Honouring the Truth, Reconciling for the Future: Summary of the Final Report of the Truth and Reconciliation Commission of Canada [Internet]. Truth and Reconciliation Commission of Canada; 2015 [cited 2022 Feb 5]. Available from: https://irsi.ubc.ca/sites/default/files/inline-files/Executive_Summary_English_Web.pdf
- Hoover C, Parker C, Hornby C, Ostertag S, Hansen-Craik, Pearce T, et al. Cultural Relevance in Arctic Food Security Initiatives. In: Sustainable Food Futures: Multidisciplinary Solutions. Routledge; 2017. p. 17–33.
- 15. Food Sovereignty and Harvesting [Internet]. Qikiqtani Inuit Association; 2019. Available from: https://www.qia.ca/wp-content/uploads/2019/03/Food-Sovereignty-and-Harvesting.pdf
- 16. Inuit Tapiriit Kanatami. Inuit Nunangat Map [Internet]. [cited 2022 Nov 18]. Available from: https://www.itk.ca/inuit-nunangat-map/
- 17. Mussell B, Cardiff K, White J. The Mental Health and Well-Being of Aboriginal Children and Youth: Guidance for New Approaches and Services. British Columbia Ministry of Children and Family Development; 2004.
- Land for Healing: Developing a First Nations Land-based Service Delivery Model [Internet]. Bothwell, Ontario: Thunderbird Partnership Foundation; 2018 [cited 2022 Feb 5]. Available from: https://thunderbirdpf.org/wp-content/uploads/2018/07/Thunderbirdpf-LandforHealing-Document-SQ.pdf
- 19. Auger MD. "The Strengths of Our Community and Our Culture": Cultural Continuity as a Determinant of Mental Health for Métis People in British Columbia. Turt Isl J Indig Health. 2021;1(2):71–83.
- 20. Reading C, Wien F. Health inequalities and social determinants of Aboriginal Peoples' Health. Prince George, British Columbia: National Collaborating Centre for Aboriginal Health; 2009 p. 36.
- 21. Karetak J, Tester F, Tagalik S. Inuit Qaujimajatuqangit, What Inuit Have Always Known to Be True. Nova Scotia: Fernwood Publishing; 2017.
- 22. Tagalik S. Inuit Qaujimajatuqangit: The role of Indigenous knowledge in supporting wellness in Inuit communites in Nunavut [Internet]. Inukpaujaq Consulting, National Collaborating Centre for Aboriginal Health; 2010 [cited 2022 Nov 9]. Available from: https://www.ccnsa-nccah.ca/docs/health/FS-InuitQaujimajatuqangitWellnessNunavut-Tagalik-EN.pdf

- 23. Snowshoe A, Crooks CV, Tremblay PF, Craig WM, Hinson RE. Development of a Cultural Connectedness Scale for First Nations youth. Psychol Assess. 2015;27(1):249–59.
- 24. Greenwood M, de Leeuw S. Social determinants of health and the future well-being of Aboriginal children in Canada. Paediatr Child Health. 2012;17(7):381–4.
- 25. Fiedeldey-Van Dijk C, Rowan M, Dell C, Mushquash C, Hopkins C, Fornssler B, et al. Honoring Indigenous culture-as-intervention: Development and validity of the Native Wellness Assessment. J Ethn Subst Abuse. 2017 Apr 3;16(2):181–218.
- 26. Marsh TN, Coholic D, Cote-Meek S, Najavits LM. Blending Aboriginal and Western healing methods to treat intergenerational trauma with substance use disorder in Aboriginal peoples who live in Northeastern Ontario, Canada. Harm Reduct J. 2015 Dec;12(1):14.
- 27. Colquhoun S, Dockery AM. The link between Indigenous culture and wellbeing: Qualitative evidence for Australian Aboriginal peoples. CLMR Discuss Pap Ser 201201.
- 28. Tourand J, Smith A, Poon C, Steward D, Society MC. Ta Saantii: A Profile of Metis Youth Health in BC [Internet]. McCreary Centre Society; 2016 [cited 2022 Jan 4]. Available from: http://www.mcs.bc.ca/pdf/ta_saantii.pdf
- 29. Milligan C. Indigenous Land-Based Healing Programs in Canada: A Scoping Review. Hotil ts'eeda, NWT Recreation and Parks Association; 2019 Aug.
- Author. Changing directions, changing lives: the mental health strategy for Canada [Internet]. Calgary, Alta.: Mental Health Commission of Canada; 2012 [cited 2022 Jan 4]. Available from: https://www.deslibris.ca/ID/232693
- 31. Zimmerman M, Ramirez J, Washienko K, Walter B, Dyer S. Enculturation hypothesis: Exploring direct and positive effects among native American youth. In: Resiliency in ethnic minority families, Vol I: Native and immigrant American families. Maddison, WI: Centre for Excellence in Family Studies, University of Wisconsin; 1994. p. 199–220.
- 32. Chen E, Brody GH, Miller GE. Childhood close family relationships and health. Am Psychol. 2017 Sep;72(6):555–66.
- Burns J, Cooke D, Schweidler C. A Short Guide to Community Based Participatory Action Research [Internet]. Los Angeles, USA: Healthy City; 2011 Dec [cited 2022 Feb 17]. (Community Research Lab Guide). Available from: https://hc-v6static.s3.amazonaws.com/media/resources/tmp/cbpar.pdf
- 34. Storey, Kate. Differences between CBPAR, CBPR, and PAR [Internet]. Message to: Rachel Cassie. 2022 Feb 25 [cited 2022 Feb 27]. [1 paragraph].
- 35. Tremblay MC, Martin DH, McComber AM, McGregor A, Macaulay AC. Understanding community-based participatory research through a social movement framework: a case study of the Kahnawake Schools Diabetes Prevention Project. BMC Public Health. 2018 Dec;18(1):487.

- 36. McTaggart R, Nixon R, Kermis S. Critical Participatory Action Research. In: The Palgrave International Handbook of Action Research. New York: Palgrave Macmillan; 2017.
- 37. About PAR [Internet]. Participatory Action Research Resources for researchers, practitioners, organizers, and communities. 2018 [cited 2022 Feb 17]. Available from: https://participatoryactionresearch.sites.carleton.edu/about-par/
- 38. Mayan M. Essentials of Qualitative Inquiry. New York: Routledge; 2009. (Qualitative Essentials).
- 39. Wilson S. Research Is Ceremony; Indigenous Research Methods [Internet]. Nova Scotia: Fernwood Publishing; 2008. Available from: https://eduq.info/xmlui/handle/11515/35872
- 40. Kovach M. Indigenous Methodologies: Characteristics, Conversations, and Contexts. 2nd ed. University of Toronto Press, Scholarly Publishing Division; 2021.
- 41. Thayer-Bacon BJ. The Nurturing of a Relational Epistemology. Educ Theory. 1997 Jun;47(2):239–60.
- 42. Thayer-Bacon B. A Pragmatist and Feminist Relational (E)pistemology. Eur J Pragmatism Am Philos [Internet]. 2010 Jul 1 [cited 2022 Mar 8];II(1). Available from: http://journals.openedition.org/ejpap/948
- 43. Wilson KJ, Bell T, Arreak A, Koonoo B, Angnatsiak D, Ljubicic GJ. Changing the role of non-Indigenous research partners in practice to support Inuit self-determination in research. Arct Sci. 2020 Sep 1;6(3):127–53.
- 44. Gerlach A. Thinking and Researching Relationally: Enacting Decolonizing Methodologies With an Indigenous Early Childhood Program in Canada. Int J Qual Methods. 2018 Dec 1;17(1):160940691877607.
- 45. Healey G, Tagak Sr. A. PILIRIQATIGIINNIQ 'Working in a collaborative way for the common good': A perspective on the space where health research methodology and Inuit epistemology come together. Int J Crit Indig Stud. 2014 Jan 1;7(1):1–14.
- 46. V. Deloria. Red Earth, White Lies: Native Americans and the myth of scientific fact. New York: Scribner; 1995.
- 47. Brayboy B, Gough H, Leonard B, Roehl II R, Solyom J. Reclaiming Scholarship: Critical Indigenous Research Methodologies. In: Qualitative research: An introduction to methods and designs. Jossey-Bass/Wiley; 2012. p. 423–50.
- 48. Gone JP. Considering Indigenous Research Methodologies: Critical Reflections by an Indigenous Knower. Qual Inq. 2019 Jan;25(1):45–56.
- 49. Antoine D. Pushing the Academy: The Need for Decolonizing Research. Can J Commun [Internet]. 2017 Feb 9 [cited 2022 Mar 2];42(1). Available from: http://cjc-online.ca/index.php/journal/article/view/3091

- 50. Tuck E. Suspending Damage: A Letter to Communities. Harv Educ Rev. 2009 Sep 1;79(3):409–28.
- Beeman-Cadwallader N, Quigley C, Yazzie-Mintz T. Enacting Decolonized Methodologies: The Doing of Research in Educational Communities. Qual Inq. 2012 Jan;18(1):3–15.
- 52. Ristaniemi H. Relational ethics in Indigenous research A reflexive navigation of whiteness and ally positionality. Sámi Lang Cult Assoc. 2019;3(2).
- 53. Peletz N, Hanna K, Noble B. The central role of Inuit Qaujimaningit in Nunavut's impact assessment process. Impact Assess Proj Apprais. 2020 Sep;38(5):412–26.
- 54. Inuit Qaujimajatuqangit [Internet]. Nunavut Impact Review Board; [cited 2022 Jan 6]. Available from: https://www.nirb.ca/inuit-qaujimajatuqangit
- 55. Inuit Qaujimajatuqangit [Internet]. Nunavut Water Board; [cited 2022 Jan 6]. Available from: https://www.nwb-oen.ca/licensing/iq
- 56. Wihak C, Merali N. Culturally Sensitive Counselling in Nunavut: Implications of Inuit Traditional Knowledge. Can J Couns. 2003;37(4).
- 57. McGregor HE. Curriculum Change in Nunavut: Towards Inuit Qaujimajatuqangit. McGill J Educ. 2012;47(3):285–302.
- 58. Laugrand F, Oosten J. Transfer of Inuit qaujimajatuqangit in modern Inuit society. Études/Inuit/Studies. 2009;33(1–2):115–52.
- 59. Docherty-Skippen SM, Woodford ED. Indigenous Knowledge as 21st Century Education: A Taxonomy of 21st Century Learning and Educational Leadership as Inuit Qaujimajatuqangit (IQ). Transform Dialogues Teach Learn J. 2017;10(3):16.
- 60. Inuit Qaujimajatuqangit [Internet]. Government of Nunavut, Department of Culture and Heritage; [cited 2022 Jan 6]. Available from: https://www.gov.nu.ca/information/inuit-qaujimajatuqangit-katimajiit
- 61. INUIT QAUJIMAJATUQANGIT [Internet]. Government of Nunavut; [cited 2022 Jan 6]. Available from: https://www.gov.nu.ca/sites/default/files/iq_brochure_draft_1.pdf
- 62. Johnston P. Constructing a Bridge Between Two Cultures: How Inuit Qaujimajatuqangit is essential to addressing the "modern" child welfare system in Nunavut. Can Assoc Soc Work Rev. 2014;31(2):22.
- 63. △△△^c L←し△^c: ∧^{sb}dr⊃^sb≫^c ⊃[∞]し&▷σ[∞]∩^c △△^c∩⊲^sσ^s J^c Inuit Customary Laws: The Foundation of Inuit Wellbeing [Internet]. Government of Nunavut, Department of Culture and Heritage; 2019 Oct [cited 2022 Oct 5] p. 28. Available from: https://www.gov.nu.ca/sites/default/files/CHildDayPosters/ch-workshop-report-inuktut-v7-2.pdf

- 64. Caughey A, Kilabuk P, Sanguya I, Doucette M, Jaw M, Allen J, et al. Niqivut Silalu Asijjipalliajuq: Building a Community-Led Food Sovereignty and Climate Change Research Program in Nunavut, Canada. Nutrients. 2022 Apr 10;14(8):1572.
- 65. Arviat Wellness Society. AAJIIQATIGIINGNIQ: An Inuit Research Methodology [Internet]. Aqqiumavvik Society; [cited 2022 Oct 10]. Available from: https://www.aqqiumavvik.com/aajiiqatigiingniq-research-methodol
- 66. AAJIIQATIGIINGNIQ: A Wellness Framework [Internet]. Arviani Aqqiumavvik Society; [cited 2022 Oct 11]. Available from: https://www.aqqiumavvik.com/_files/ugd/1f7032_c576197915714d75a6143c0f355d55a3.p df
- 67. Strand KJ, Cutforth N, Stoecker R, Marullo, S, Donohue P. Community-Based Research and Higher Education. 1st ed. Jossey-Bass; 2003. 85, 97 p.
- 68. Hirsch R, Furgal C, Hackett C, Sheldon T, Bell T, Angnatok D, et al. Going Off, Growing Strong: A program to enhance individual youth and community resilience in the face of change in Nain, Nunatsiavut. Études/Inuit/Studies. 2016;40(1):63–84.
- 69. Newell SL, Dion ML, Doubleday NC. Cultural continuity and Inuit health in Arctic Canada. J Epidemiol Community Health. 2020 Jan;74(1):64–70.
- 70. Chandler MJ, Lalonde C. Cultural Continuity as a Hedge against Suicide in Canada's First Nations. Transcult Psychiatry. 1998 Jun;35(2):191–219.
- Koc M. Discourses of Food Security. In: Karaagac B, editor. Accumulations, crises, struggles: capital and labour in contemporary capitalism. Berlin/London: LIT Verlag; 2013. p. 245–65.
- 72. Winkler H, Winthrop-Young G, Wutz M. Discourses, Schemata, Technology, Monuments: Outline for a Theory of Cultural Continuity. Configurations. 2002;10(1):91–109.
- 73. Urban G. An Approach to Culture and Language; Grammatical Parrallelism and Thought. In: A Discourse-Centered Approach to Culture: Native South American Myths and Rituals [Internet]. Austin, Texas: University of Texas Press; 1991 [cited 2022 Jan 28]. (Texas Linguistic Series). Available from: http://newdoc.nccu.edu.tw/teasyllabus/113721259057/Greg%20Urban%20Chapter%201.pd f
- 74. Wexler L. Identifying Colonial Discourses in Inupiat Young People's Narratives as a Way to Understand the No Future of Inupiat Youth Suicide. Am Indian Alsk Native Ment Health Res. 2009;16(1):1–24.
- 75. Ostrowicka H. Regulating Social Life: Discourses on the Youth and the Dispositif of Age [Internet]. Cham: Springer International Publishing; 2019 [cited 2022 Jan 28]. Available from: http://link.springer.com/10.1007/978-3-030-16888-9

- 76. What is Decolonization? What is Indigenization? [Internet]. Queen's University Centre for Teaching and Learning. [cited 2022 Oct 24]. Available from: https://www.queensu.ca/ctl/resources/decolonizing-and-indigenizing/what-decolonizationwhat-indigenization
- 77. Wakeham P. Reconciliation, Indigenization, Decolonization, and Resurgence [Internet]. Western University. [cited 2022 Oct 17]. Available from: https://indigenous.uwo.ca/docs/Indigenous-Initiatives-Key-Terms.pdf
- 78. Coulthard G. Red Skin, White Masks: Rejecting the Colonial Politics of Recognition. Minneapolis, MN: University of Minnesota Press; 2014.
- 79. Tuck E, Yang KW. Decolonization is not a metaphor. Decolonization Indig Educ Soc. 2012;1(1):40.
- 80. Smith LT. Decolonizing Methodologies: Research and Indigenous Peoples. 3rd ed. Bloomsbury Publishing; 2021.
- 81. Waddell CM, Robinson R, Crawford A. Decolonizing Approaches to Inuit Community Wellness: Conversations with Elders in a Nunavut Community. Can J Commun Ment Health. 2017 May;36(1):1–13.
- 82. Leblanc DAM. The Roles of Settler Canadians within Decolonization: Re-evaluating Invitation, Belonging and Rights. Can J Polit Sci. 2021 Jun;54(2):356–73.
- 83. Richards L, Morse J. README FIRST for a User's Guide to Qualitative Methods. 3rd ed. Thousand Oaks, California: Sage Publications; 2012.
- 84. Castleden H, Morgan VS, Lamb C. "I spent the first year drinking tea": Exploring Canadian university researchers' perspectives on community-based participatory research involving Indigenous peoples: Researchers' perspectives on CBPR. Can Geogr Géographe Can. 2012 Jun;56(2):160–79.
- Olshansky E, Sacco D, Braxter B, Dodge P, Hughes E, Ondeck M, et al. Participatory action research to understand and reduce health disparities. Nurs Outlook. 2005 May;53(3):121–6.
- 86. Hyman A, Stacy E, Mohsin H, Atkinson K, Stewart K, Novak Lauscher H, et al. Barriers and Facilitators to Accessing Digital Health Tools Faced by South Asian Canadians in Surrey, British Columbia: Community-Based Participatory Action Exploration Using Photovoice. J Med Internet Res. 2022 Jan 13;24(1):e25863.
- 87. Wallerstein NB, Duran B. Using Community-Based Participatory Research to Address Health Disparities. Health Promot Pract. 2006 Jul;7(3):312–23.
- Community Based Participatory Action Research [Internet]. Youtube; 2020 [cited 2022 Feb 18]. Available from: https://www.youtube.com/watch?v=Ed36qPrpnq8

- 89. Shore N. Community-Based Participatory Research and the Ethics Review Process. J Empir Res Hum Res Ethics. 2007 Mar;2(1):31–41.
- 90. Stiegman ML, Castleden H. Leashes and Lies: Navigating the Colonial Tensions of Institutional Ethics of Research Involving Indigenous Peoples in Canada. Int Indig Policy J [Internet]. 2015 Jun 18 [cited 2022 Oct 18];6(3). Available from: https://ojs.lib.uwo.ca/index.php/iipj/article/view/7465
- 91. Malone RE, Yerger VB, McGruder C, Froelicher E. "It's Like Tuskegee in Reverse": A Case Study of Ethical Tensions in Institutional Review Board Review of Community-Based Participatory Research. Am J Public Health. 2006 Nov;96(11):1914–9.
- 92. Wood L. The Ethical Implications of Community-Based Research: A Call to Rethink Current Review Board Requirements. Int J Qual Methods. 2017 Dec;16(1):160940691774827.
- 93. Kraemer Diaz AE, Spears Johnson CR, Arcury TA. Perceptions That Influence the Maintenance of Scientific Integrity in Community-Based Participatory Research. Health Educ Behav. 2015 Jun;42(3):393–401.
- 94. Lavoie JG, Dwyer J. Implementing Indigenous community control in health care: lessons from Canada. Aust Health Rev. 2016;40(4):453.
- 95. McKenzie C. LGBTQ + community organizations under neoliberalism in Ontario, Canada: Contradictions and disconnects. J Gay Lesbian Soc Serv. 2020 Apr 2;32(2):240–60.
- 96. Shore N, Brazauskas R, Drew E, Wong KA, Moy L, Baden AC, et al. Understanding Community-Based Processes for Research Ethics Review: A National Study. Am J Public Health. 2011 Dec;101(S1):S359–64.
- 97. Inuit Tapiriit Kanatami. National Inuit Strategy on Research [Internet]. Ottawa, ON: Inuit Tapiriit Kanatami; 2018 [cited 2022 Feb 21]. Available from: https://www.itk.ca/wp-content/uploads/2020/10/ITK-National-Inuit-Strategy-on-Research.pdf
- 98. Negotiating Research Relationships: A Guide for Researchers [Internet]. ITK, NRI; 2006 [cited 2022 Oct 14]. Available from: https://www.nri.nu.ca/sites/default/files/public/files/06-068_itk_nrr_booklet.pdf
- 99. Calls for Justice. In: Reclaiming Power and Place: The Final Report of the National Inquiry into Missing and Murdered Indigenous Women and Girls [Internet]. National Inquiry into Missing and Murdered Indigenous Women and Girls; 2019 [cited 2022 Oct 14]. Available from: https://www.mmiwg-ffada.ca/wp-content/uploads/2019/06/Calls_for_Justice.pdf
- 100. Truth and Reconciliation Commission of Canada: Calls to Action [Internet]. Winnipeg, Manitoba: Truth and Reconciliation Commission of Canada; 2012 [cited 2022 Oct 14]. Available from: https://www2.gov.bc.ca/assets/gov/british-columbians-ourgovernments/indigenous-people/aboriginal-peoplesdocuments/calls_to_action_english2.pdf

- 101. Health Research Ethics Workshop and Community Consultation [Internet]. Arctic Health Research Network; 2007 [cited 2022 Oct 14]. Available from: https://www.qhrc.ca/wpcontent/uploads/2019/09/Summary_Notes_FINAL.pdf
- 102. Healey G. QAUJIGIARTIIT FINAL REPORT 2007-08 [Internet]. Arctic Health Research Network; 2008 [cited 2022 Oct 14]. Available from: https://www.qhrc.ca/wpcontent/uploads/2019/09/Qaujigiartiit-Final-Report-2007-8-1_0.pdf
- 103. Government of Canada, Panel on Research Ethics, Canadian Institutes of Health Research, Natural Sciences and Engineering, Social Sciences and Humanities Research. Chapter 9: Research Involving the First Nations, Inuit and Métis Peoples of Canada. In: Tri-Council Policy Statement 2 [Internet]. Ottawa, ON: Government of Canada; 2018 [cited 2022 Oct 14]. Available from: https://ethics.gc.ca/eng/tcps2-eptc2_2018_chapter9-chapitre9.html
- 104. Social Sciences and Humanities Research. Guidelines for the Merit Review of Indigenous Research [Internet]. Government of Canada; 2018 [cited 2022 Oct 14]. Available from: https://www.sshrc-crsh.gc.ca/funding-financement/merit_reviewevaluation_du_merite/guidelines_research-lignes_directrices_recherche-eng.aspx
- 105. Canadian Institutes of Health Research. CIHR Guidelines for Health Research Involving Aboriginal People (2007-2010) [Internet]. Government of Canada; 2013 [cited 2022 Oct 14]. Available from: https://cihr-irsc.gc.ca/e/29134.html
- 106. Graham A, McDonald J, ACUNS Board Committee on Revising the Ethical Principles. Ethical Principles for the Conduct of Research in the North. Assoc Can Univ North Stud. 2003;4(2):114–114.
- 107. Qaujigiartiit Health Research Centre. HEALTH RESEARCH ETHICS CHECKLIST [Internet]. 2019 [cited 2022 Oct 14]. Available from: https://www.qhrc.ca/wpcontent/uploads/2019/09/Ethics-Review-Checklist-OCT-2019.pdf
- 108. Harper S, Edge VL, Cunsolo Willox A, Rigolet Inuit Community Government. 'Changing Climate, Changing Health, Changing Stories' Profile: Using an EcoHealth Approach to Explore Impacts of Climate Change on Inuit Health. EcoHealth. 2012 Mar;9(1):89–101.
- 109. Lichen Consulting. Qaujigiartiit Strategic Plan 2018-2028 [Internet]. Qaujigiartiit Health Research Centre; 2018 [cited 2022 Feb 21]. Available from: https://www.qhrc.ca/wpcontent/uploads/2019/02/QHRC-Strategic-Plan-FINAL-OCT-2018.pdf
- 110. Moore C, Castleden HE, Tirone S, Martin D. Implementing the Tri-Council Policy on Ethical Research Involving Indigenous Peoples in Canada: So, How's That Going in Mi'kma'ki? Int Indig Policy J [Internet]. 2017 Mar 10 [cited 2022 Oct 18];8(2). Available from: https://ojs.lib.uwo.ca/index.php/iipj/article/view/7514
- 111. Day L, Cunsolo A, Castleden H, Martin D, Hart C, Anaviapik-Soucie T, et al. The Expanding Digital Media Landscape of Qualitative and Decolonizing Research: Examining Collaborative Podcasting as a Research Method. 2017;(1):27.

- 112. CIHR Planning and Dissemination Grants ICS Winter 2022 [Internet]. University of Guelph. [cited 2022 Oct 16]. Available from: https://www.uoguelph.ca/research/alerts/content/cihr-planning-and-dissemination-grantsics-winter-2022
- 113. IHSPR Institute Community Support (ICS) Program [Internet]. Canadian Institutes of Health Research. 2022 [cited 2022 Oct 16]. Available from: https://cihrirsc.gc.ca/e/36098.html
- 114. Hackett C, Furgal C, Angnatok D, Sheldon T, Karpik S, Baikie D, et al. Going Off, Growing Strong: Building Resilience of Indigenous Youth. Can J Commun Ment Health. 2016 Oct;35(2):79–82.
- 115. Eggan D. Instruction and Affect in Hopi Cultural Continuity. Southwest J Anthropol. 1956;12(4):25.
- 116. Kirmayer LJ, Aboriginal Healing Foundation (Canada). Suicide among aboriginal people in Canada [Internet]. Ottawa, Ont.: Aboriginal Healing Foundation; 2007 [cited 2022 Jan 31]. Available from: https://www.deslibris.ca/ID/250873
- 117. Social Determinants of Health [Internet]. World Health Organization. [cited 2022 Jan 31]. Available from: https://www.who.int/health-topics/social-determinants-of-health#tab=tab_1
- 118. Social Determinants of Inuit Health in Canada [Internet]. Inuit Tapiriit Kanatami; 2014 [cited 2022 Oct 6]. Available from: https://www.itk.ca/wpcontent/uploads/2016/07/ITK_Social_Determinants_Report.pdf
- 119. Convention on the Rights of the Child [Internet]. United Nations; 1989 Nov [cited 2022 Oct 20]. Available from: https://www.unicef-irc.org/portfolios/general_comments/CRC.GC.C.11.doc.html
- 120. United Nations Declaration on the Rights of Indigenous Peoples [Internet]. United Nations; 2007 Sep. Available from: https://www.un.org/development/desa/indigenouspeoples/wp-content/uploads/sites/19/2018/11/UNDRIP_E_web.pdf
- 121. Implementing the United Nations Declaration on the Rights of Indigenous Peoples Act [Internet]. Government of Canada; 2021 Jun [cited 2022 Jan 31]. Available from: https://www.justice.gc.ca/eng/declaration/index.html
- 122. Submission to the House of Commons Standing Committee on Indigenous and Northern Affairs - BILL C-15: AN ACT RESPECTING THE UNITED NATIONS DECLARATION ON THE RIGHTS OF INDIGENOUS PEOPLES [Internet]. Inuit Tapiriit Kanatami; 2021 [cited 2022 Jan 31]. Available from: https://www.itk.ca/wpcontent/uploads/2021/04/20210412-ITK-submission-to-INAN-Bill-C-15 E.pdf
- 123. Robinson AM. Governments must not wait on courts to implement UNDRIP rights concerning Indigenous sacred sites: lessons from Canada and *Ktunaxa Nation v. British Columbia*. Int J Hum Rights. 2020 Nov 25;24(10):1642–65.

- 124. Cattroll F. Culture and language as social determinants of First Nations, Inuit, and Métis health [Internet]. National Collaborating Centre for Indigenous Health; 2016 [cited 2022 Feb 7]. Available from: https://www.ccnsa-nccah.ca/docs/determinants/FS-CultureLanguage-SDOH-FNMI-EN.pdf
- 125. Oster RT, Grier A, Lightning R, Mayan MJ, Toth EL. Cultural continuity, traditional Indigenous language, and diabetes in Alberta First Nations: a mixed methods study. Int J Equity Health. 2014 Dec;13(1):92.
- 126. Cormier M, Schwartzman K, N'Diaye DS, Boone CE, dos Santos AM, Gaspar J, et al. Proximate determinants of tuberculosis in Indigenous peoples worldwide: a systematic review. Lancet Glob Health. 2019 Jan;7(1):e68–80.
- 127. Møller H. A Problem of the Government? Colonization and the Socio-Cultural Experience of Tuberculosis in Nunavut [Internet] [Master's]. Institute for Anthropology, Copenhagen University; 2005 [cited 2022 Feb 23]. Available from: https://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.488.620&rep=rep1&type=pdf
- 128. Newell SL, Doubleday NC, Community of Chesterfield Inlet, Nunavut. Sharing country food: connecting health, food security and cultural continuity in Chesterfield Inlet, Nunavut. Polar Res [Internet]. 2020 Dec 18 [cited 2022 Jan 31];39. Available from: https://polarresearch.net/index.php/polar/article/view/3755
- 129. Baldwin MW. Relational Schemas and the Processing of Social Information. Pyschological Bull. 1992;112(3):24.
- 130. Lee RM, Draper M, Lee S. Social Connectedness, Dysfunctional Interpersonal Behaviors, and Psychological Distress: Testing a Mediator Model. J Couns Pyschology. 2001;48(3):9.
- 131. Richmond CAM, Ross NA. Social support, material circumstance and health behaviour: Influences on health in First Nation and Inuit communities of Canada. Soc Sci Med. 2008 Nov;67(9):1423–33.
- 132. Vaudry S. "Our connection makes us stronger": Inuit youth's strategies to feel comfortable in Ottawa. Études/Inuit/Studies. 2016;40(1):127–46.
- 133. Murrup-Stewart C, Whyman T, Jobson L, Adams K. "Connection to Culture Is Like a Massive Lifeline": Yarning With Aboriginal Young People About Culture and Social and Emotional Wellbeing. Qual Health Res. 2021 Aug;31(10):1833–46.
- 134. Gottlieb BH. Social Networks and Social Support: An Overview of Research, Practice, and Policy Implications. Health Educ Q. 1985 Mar;12(1):5–22.
- 135. Bond R. Conformity across Cultures. In: Spielberger C, editor. Encyclopedia of Applied Psychology. Academic Press; 2004.
- 136. Qikiqtani Truth Commission, Qikiqtani Inuit Association. Nuutauniq: Moves in Inuit Life. In: Qikiqtani Truth Commission: Thematic Reports and Special Studies 1950–1975

[Internet]. Iqaluit, Nunavut: Inhabit Media; 2013 [cited 2022 Feb 22]. Available from: https://www.qtcommission.ca/sites/default/files/public/thematic_reports/thematic_reports_e nglish_nuutauniq.pdf

- 137. Fraser S, Rousseau C, Kasudluak R, Burmester P, Arauz MJ. Culturally Appropriate Care— A Multicultural Task: Assessing the Needs of Inuit Youth in the Care of Child Welfare Services Δc^{-%}dパ[%]∩J^c Γ⊲σ[%]/σ[%]—⊲Γ/σ^{*} /c∩σ^{*}: ⁶PΓ⁶Pσ^{*} P^{e*}JL' ל⊂D לσ^b ΔΔ^eD^c L^bd^cD^c Γ⊲σ∩↓D^c d^cD^c /P^c⊂∩^bd^eσ^c. J Aborig Health [Internet]. 2015 [cited 2022 Feb 24];9(2). Available from: https://jps.library.utoronto.ca/index.php/ijih/article/view/29094
- 138. Perreault K, Riva M, Dufresne P, Fletcher C. Overcrowding and sense of home in the Canadian Arctic. Hous Stud. 2020 Feb 7;35(2):353–75.
- 139. Healey GK, Noah J, Mearns C. The Eight Ujarait (Rocks) Model: Supporting Inuit Adolescent Mental Health With an Intervention Model Based on Inuit Ways of Knowing. Int J Indig Health. 2016 Jun 30;11(1):92–110.
- 140. Crawford A. Inuit take action towards suicide prevention. The Lancet. 2016 Sep 10;388(10049):1036–8.
- 141. Dowsley M. Identity and the evolving relationship between Inuit women and the land in the eastern Canadian Arctic. Polar Rec. 2014;51(5):536–49.
- 142. Gregory R, Halteman P, Kaechele N, Kotaska J, Satterfield T. Compensating Indigenous social and cultural losses: a community-based multiple-attribute approach. Ecol Soc. 2020;25(4):art4.
- 143. Lines LA, Yellowknives Dene First Nation Wellness Division, Jardine CG. Connection to the land as a youth-identified social determinant of Indigenous Peoples' health. BMC Public Health. 2019 Dec;19(1):176.
- 144. Allen J, Hopper K, Wexler L, Kral M, Rasmus S, Nystad K. Mapping resilience pathways of Indigenous youth in five circumpolar communities. Transcult Psychiatry. 2014 Oct;51(5):601–31.
- 145. Goggin G, Crawford K. Generation Disconnections: Youth Culture and Mobile Communication. In: Mobile Communication: Bringing Us Together and Tearing Us Apart. 1st ed. Routledge; 2011.
- 146. Winstone L, Mars B, Haworth CMA, Kidger J. Social media use and social connectedness among adolescents in the United Kingdom: a qualitative exploration of displacement and stimulation. BMC Public Health. 2021 Dec;21(1):1736.
- 147. Nunavut Tunngavik. THE NUNAVUT AGREEMENT [Internet]. [cited 2022 Oct 19]. Available from: https://nlca.tunngavik.com/

- 148. Laneuville P. Bring Hope and Restore Peace, A Study Report on the Life and Concerns of Inuit Women of Nunavik [Internet]. Saturviit Inuit Women's Association of Nunavik; 2015 [cited 2022 Oct 19]. Available from: https://www.cerp.gouv.qc.ca/fileadmin/Fichiers_clients/Documents_deposes_a_la_Commis sion/P-105.pdf
- 149. Bonesteel S, Anderson E. Canada's relationship with Inuit: a history of policy and program development. Ottawa: Indian and Northern Affairs Canada; 2006. 240 p.
- 150. Our Research Model [Internet]. Qaujigiartiit Health Research Centre. [cited 2022 Feb 21]. Available from: https://www.qhrc.ca/our-approach/our-research-model/
- 151. Inuit Tuberculosis Elimination Framework [Internet]. Ottawa: Inuit Tapiriit Kanatami (ITK); 2018 [cited 2022 Oct 21]. Available from: https://www.itk.ca/wp-content/uploads/2018/12/FINAL-ElectronicEN-Inuit-TB-Elimination-Framework.pdf
- 152. Abdollahi E, Keynan Y, Foucault P, Brophy J, Sheffield H, Moghadas SM. Evaluation of TB elimination strategies in Canadian Inuit populations: Nunavut as a case study. Infect Dis Model. 2022 Dec;7(4):698–708.
- 153. Inuit-Specific Tuberculosis (TB) Strategy [Internet]. Inuit Tapiriit Kanatami; 2013 [cited 2022 Oct 21]. Available from: https://www.itk.ca/wp-content/uploads/2013/03/20130503-EN-FINAL-Inuit-TB-Strategy.pdf
- 154. Møller H. Tuberculosis and Colonialism: Current Tales about Tuberculosis and Colonialism in Nunavut. Int J Indig Health. 2010 Jan;6(1):38–48.
- 155. EXECUTIVE SUMMARY: Nunavut's Regional Action Plan for Tuberculosis Elimination [Internet]. Nunavut Tunngavik Incorporated; 2021 [cited 2022 Oct 21]. Available from: https://www.tunngavik.com/files/2021/03/20210324-NUTBEAP-Executive-Summary_final-ENG.pdf
- 156. Tuberculosis Manual [Internet]. Department of Health, Nunavut; 2017 [cited 2022 Oct 21]. Available from: https://www.gov.nu.ca/sites/default/files/nunavut-tuberculosis-manual-2018.pdf
- 157. Diagnostic and Statistical Manual of Mental Disorders (DSM-5®). Washington, DC: American Psychiatric Association; 2013.
- 158. Fainsinger R, Thai V, Frank, Fergusson J. What's in a Word? Addiction Versus Dependence in DSM-V. Am J Psychiatry. 2006;163(11).
- 159. Johnson G, Chamberlain C. Homelessness and Substance Abuse: Which Comes First? Aust Soc Work. 2008 Dec;61(4):342–56.
- 160. Linton J, Campbell J, Gressick J. What's in a Word? Definitions and Interpretations of Terms Commonly Used to Describe Substance Use Issues. J Alcohol Drug Educ. 2016;60(1).

- Acuff SF, Tucker JA, Murphy JG. Behavioral economics of substance use: Understanding and reducing harmful use during the COVID-19 pandemic. Exp Clin Psychopharmacol. 2021 Dec;29(6):739–49.
- 162. Wendt DC, Hartmann WE, Allen J, Burack JA, Charles B, D'Amico EJ, et al. Substance Use Research with Indigenous Communities: Exploring and Extending Foundational Principles of Community Psychology. Am J Community Psychol. 2019 Sep;64(1–2):146– 58.
- 163. Vowell C. The Myth of the Drunken Indian. In: Indigenous Writes: A Guide to First Nations, Metis, Inuit Issues in Canada. Winnipeg, Manitoba: Highwater Press; 2016. p. 154–5.
- 164. Aboriginal Peoples: Fact Sheet for Canada [Internet]. Statistics Canada; 2012 [cited 2022 Feb 2]. Available from: https://www150.statcan.gc.ca/n1/pub/89-656-x/89-656-x2015001eng.htm
- 165. Taking Steps to Reduce Alcohol-Related Harm in Nunavut [Internet]. Government of Nunavut; 2016 Oct [cited 2022 Feb 3]. Available from: https://www.gov.nu.ca/finance/information/taking-steps-reduce-alcohol-related-harmnunavut
- 166. RCMP Iqaluit drug seizure [Internet]. Royal Canadian Mounted Police. 2022 [cited 2022 Oct 22]. Available from: https://www.rcmp-grc.gc.ca/en/news/2022/rcmp-iqaluit-drugseizure
- 167. Iqaluit man charged with cocaine possession for the purpose of trafficking [Internet]. Royal Canadian Mounted Police. 2019 [cited 2022 Oct 22]. Available from: https://www.rcmp-grc.gc.ca/en/news/2019/iqaluit-man-charged-cocaine-possession-the-purpose-trafficking
- 168. Munchel W. Trauma Is the Gateway Drug an Ace's Informed Approach to Substance Abuse Treatment [Internet]. Children's Mental Health Network. 2017 [cited 2022 Feb 3]. Available from: https://www.cmhnetwork.org/news/trauma-is-the-gateway-drug-an-acesinformed-approach-to-substance-abuse-treatment/
- 169. Gibbard R. Nunavut Liquor and Cannabis Commission's Rankin Inlet store opening [Internet]. Department of Finance, Nunavut. 2021 [cited 2022 Feb 25]. Available from: https://www.gov.nu.ca/finance/news/nunavut-liquor-and-cannabis-commissions-rankininlet-store-opening
- 170. Community Voices: A Report from the Regional Community Health and Wellness Meetings [Internet]. Government of Nunavut, Department of Health and Social Services; 2010 Mar [cited 2022 Feb 3]. Available from: https://www.gov.nu.ca/sites/default/files/files/Final%20Community%20Voices%20Report %20English%20pdf.pdf

- 171. What is an Alcohol Education Committee? [Internet]. ▷'ネ^{sh}/^{sh}⊂ Let's Be Aware Qauyimagiaqaqtugut Ayons les idées claires. 2017 [cited 2022 Feb 3]. Available from: https://www.responsiblenunavut.ca/en/community/what-alcohol-education-committee
- 172. Lepage-Arnakak D. Consolidation of Liquor Act. 2020 Feb 1;80.
- 173. Whitesell M, Bachand A, Peel J, Brown M. Familial, Social, and Individual Factors Contributing to Risk for Adolescent Substance Use. J Addict. 2013;2013:1–9.
- 174. Beman D. Risk factors leading to adolescent substance abuse. Adolescence. 1995;30(117):8.
- 175. Higgins K, McLaughlin A, Perra O, McCartan C, McCann M, Percy A, et al. The Belfast Youth Development Study (BYDS): A prospective cohort study of the initiation, persistence and desistance of substance use from adolescence to adulthood in Northern Ireland. Maggs JL, editor. PLOS ONE. 2018 May 23;13(5):e0195192.
- 176. Muchiri BW, dos Santos MML. Family management risk and protective factors for adolescent substance use in South Africa. Subst Abuse Treat Prev Policy. 2018 Dec;13(1):24.
- 177. Leslie K. Youth substance use and abuse: challenges and strategies for identification and intervention. Can Med Assoc J. 2008 Jan 15;178(2):145–8.
- 178. McMahon B, Rudella JL, McMahon M, Wendling K, Paredes A, Register M. Community-Based Participatory Research: Engaging Youth to Provide Perspective on Risk and Protective Factors. J Sch Health. 2020 Sep;90(9):673–82.
- 179. Taylor OD. Predictors and Protective Factors in the Prevention and Treatment of Adolescent Substance Use Disorders. J Hum Behav Soc Environ. 2010 Aug 5;20(5):601– 17.
- 180. Vowell C. Indigenous Writes: A Guide to First Nations, Metis, Inuit Issues in Canada. Winnipeg, Manitoba: Highwater Press; 2016.
- 181. About D.A.R.E [Internet]. Drug Abuse Resistance Education. [cited 2022 Feb 4]. Available from: https://dare.org/about/
- Leban L, Griffin OH. Disentangling the links between risk factors and adolescent substance use. J Subst Use. 2020 Nov 1;25(6):644–51.
- 183. Allen J, Vallone D, Vargyas E, Healton C, American Legacy Foundation. The truth Campaign: Using Countermarketing to Reduce Youth Smoking. In: The New World of Health Promotion: New Program Development, Implementation, and Evaluation: New Program Development, Implementation, and Evaluation. 1st ed. Jones & Bartlett; 2009.

- 184. McKenna J, Gutierrez K, McCall K. Strategies for an Effective Youth Counter-Marketing Program: Recommendations from Commerical Marketing Experts. J Public Health Manag Pract. 2000;8(1):7–13.
- 185. Lilienfeld S, Arkowitz H. Why "Just Say No" Doesn't Work. Sci Am [Internet]. 2014 Jan 1 [cited 2022 Feb 4]; Available from: https://www.scientificamerican.com/article/why-justsay-no-doesnt-work/
- 186. Friederiksen L. Why Just Say No Typically Doesn't Work [Internet]. CHANGING THE CONVERSATIONS BREAKING THE CYCLES. 2013 [cited 2022 Feb 4]. Available from: https://www.breakingthecycles.com/blog/2013/06/23/why-just-say-no-typicallydoesnt-work/
- 187. Rosenbaum M. Real Drug Education Should Say "Just Say Know" [Internet]. Drug Policy Alliance. 2017 [cited 2022 Feb 4]. Available from: https://drugpolicy.org/blog/real-drugeducation-should-say-just-say-know
- 188. McGrath M. Nancy Reagan and the negative impact of the "Just Say No" anti-drug campaign. The Guardian [Internet]. 2016 Mar 8 [cited 2022 Feb 4]; Available from: https://www.theguardian.com/society/2016/mar/08/nancy-reagan-drugs-just-say-no-dareprogram-opioid-epidemic
- 189. Mackey-Kallis S, Hahn DF. Questions of public will and private action: The power of the negative in the reagans' "just say no" morality campaign. Commun Q. 1991 Jan;39(1):1–17.
- 190. Harvey SA, McKay MT, Sumnall HR. Adolescents' reflections on school-based alcohol education in the United Kingdom: education as usual compared with a structured harm reduction intervention. J Subst Use. 2016 Nov;21(6):640–5.
- 191. Keane H. Critiques of harm reduction, morality and the promise of human rights. Int J Drug Policy. 2003 Jun;14(3).
- 192. Indigenous Harm Reduction = Reducing the Harms of Colonialism [Internet]. Canadian Aboriginal Aids Network, Interagency Coalition on AIDS and Development; 2019 Mar [cited 2022 Feb 7]. Available from: http://www.icad-cisd.com/pdf/Publications/Indigenous-Harm-Reduction-Policy-Brief.pdf
- 193. Desrosiers S. The productivity of culturally important berry species in the Kugluktuk region of Nunavut, and their use in land-based education programs connecting Elders and youth [Internet] [Master's of Science in Geography]. [Vancouver]: University of British Columbia; 2017 [cited 2022 Jan 7]. Available from: https://open.library.ubc.ca/soa/cIRcle/collections/ubctheses/24/items/1.0347289
- 194. Mearns C, Healey G. Makimautiksat Youth Camp, Program Evaluation 2010-2015 [Internet]. Iqaluit: Qaujigiartiit Health Research Centre; 2015 Feb [cited 2022 Oct 4]. Available from: https://www.qhrc.ca/makimautiksat-youth-camp-program-evaluation-2010-2015/

- 195. Attygalle L. Understanding Community-Led Approaches to Community Change. Tamarack Inst. 2020;13.
- 196. What is a community-led approach? How is it different from a community-based approach? [Internet]. Supporting Community-Led Child Protection: An Online Guide and Toolkit. [cited 2022 Feb 8]. Available from: https://communityledcp.org/faqs/what-is-a-community-led-approach-is-it-just-a-community-based-approach#:~:text=A%20community%2Dled%20approach%20involves,a%20fundamental% 20shift%20in%20approach.
- 197. Zimmerman MA. Resiliency Theory: A Strengths-Based Approach to Research and Practice for Adolescent Health. Health Educ Behav. 2013 Aug;40(4):381–3.
- 198. Mathie A, Cunningham G. Mobilizing Assets for Community-Driven Development. Coady International Institute St. Francis Xavier University; 2008.
- 199. Willetts J, Asker S, Carrard N, Winterford K. The practice of a strengths-based approach to community development in Solomon Islands. Dev Stud Res. 2014 Jan;1(1):354–67.
- 200. Wildcat M, McDonald M, Irlbacher-Fox S, Coulthard G. Learning from the land: Indigenous land based pedagogy and decolonization. Decolonization Indig Educ Soc. 2014;3(3):15.
- 201. Education Framework: Inuit Qaujimajatuqangit for Nunavut Curriculum [Internet]. Nunavut Department of Education, Curriculum and School Services Division; 2007 [cited 2022 Jan 7]. 67 p. Available from: https://gov.nu.ca/sites/default/files/inuitqaujimajatuqangit eng.pdf
- 202. McGregor HE, McGregor CA. Behind the Scenes of Inuit Curriculum Development in Nunavut, 2000–2013. Études Inuit Stud. 2016;40(2):109–31.
- 203. 2019 2020 Nunavut Approved Curriculum and Teaching Resources [Internet]. Arviat, NU: Department of Education, Nunavut; 2019 Aug [cited 2022 Jan 7]. Available from: https://gov.nu.ca/sites/default/files/2019-20 nunavut approved curriculum and teaching resources.pdf
- 204. First Canadians, Canadians First: National Strategy on Inuit Education 2011 [Internet]. Inuit Tapiriit Kanatami; 2011 Jun [cited 2022 Oct 4]. Available from: https://itk.ca/wpcontent/uploads/2011/06/National-Strategy-on-Inuit-Education-2011.pdf
- 205. McGregor HE, McGregor CA. Inuit Elder Policy Guidance for System-Wide Educational Change in Nunavut, 2000-2013. 2020;(193):128–42.
- 206. Teachers' Key Essentials Guide Draft May 2020 [Internet]. Department of Education, Nunavut; 2020 [cited 2022 Jan 7]. Available from: https://gov.nu.ca/sites/default/files/teacherskeyessentialguide_eng.pdf

- 207. Kotierk A. On the Inuktut language, the Inuit may have been too welcoming. MacLean's [Internet]. 2018 Feb 22 [cited 2022 Jan 7]; Available from: https://www.macleans.ca/opinion/on-the-inuktut-language-the-inuit-may-have-been-too-welcoming/
- 208. Douglas M. Southern Teachers Working in Canada's North: The Need for a Relevant Curriculum. Learn Landsc. 2009 Feb 2;2(2):243–54.
- 209. Jay DO. Inuit in Nunavut launch lawsuit against English-dominated schooling. The Breach [Internet]. 2021 Oct 14 [cited 2022 Jan 7]; Available from: https://breachmedia.ca/inuit-in-nunavut-launch-lawsuit-against-english-dominated-schooling/
- 210. Newberry N. Southern teachers for the North. Arctic Journal [Internet]. 2013 Nov 1 [cited 2022 Feb 9]; Available from: https://arcticjournal.ca/arts-culture-education/southern-teachers-for-the-north/
- 211. Orientation Supports [Internet]. Department of Education, Nunavut. [cited 2022 Jan 7]. Available from: https://gov.nu.ca/education/information/orientation-supports
- 212. Berger P, Inootik K, Jones R, Kadjukiv J. A Hunger to Teach: Recruiting Inuit Teachers for Nunavut. Études Inuit Stud. 2016 Jan 15;40(2):47–69.
- 213. Abele F, Graham K. The Literature on Inuit-Centred Curriculum and Teaching Approaches [Internet]. National Committee on Inuit Education; 2010 [cited 2022 Feb 25]. Available from: https://www.itk.ca/documentation-national-strategy-on-inuit-education/
- 214. NTI Files Lawsuit Against GN for Violating Equality Rights of Inuit Children and Youth. Nunavut Tunngavik Inc [Internet]. 2021 Oct 13 [cited 2022 Feb 9]; Available from: https://www.tunngavik.com/news/nti-files-lawsuit-against-gn-for-violating-equality-rightsof-inuit-children-and-youth/
- 215. Programs and Partnerships [Internet]. Department of Environment, Nunavut. [cited 2022 Jan 7]. Available from: https://www.gov.nu.ca/environment/information/programs-and-partnerships
- 216. Ljubicic GJ, Mearns R, Okpakok S, Robertson S. Learning from the land (Nunami iliharniq): Reflecting on relational accountability in land-based learning and cross-cultural research in Uqšuqtuuq (Gjoa Haven, Nunavut). Arct Sci. 2021 Dec 29;1–40.
- 217. Land as teacher: understanding Indigenous land-based education [Internet]. Canadian Commission for UNESCO. 2021 [cited 2022 Feb 25]. Available from: https://en.ccunesco.ca/idealab/indigenous-land-based-education
- 218. McGregor HE. One Classroom, Two Teachers? Crit Educ. 2017;8(14):19.
- 219. Madden B, Higgins M, Korteweg L. "Role models can't just be on posters": Re/membering Barriers to Indigenous Community Engagement. Can J Educ. 2013;36(2):37.

- 220. Flaherty L. Nunavut Teacher Education Program [Internet]. Council of Ministers of Education; [cited 2022 Feb 9]. Available from: https://www.cmec.ca/docs/Aboriginal/NU-NunavutTeachersEducationProgram.en.pdf
- 221. d⊂∆ McKitrick PA. A Guide to Teaching in Nunavut [Internet] [Master of Education]. Lakehead University; 2021 [cited 2022 Feb 9]. Available from: https://knowledgecommons.lakeheadu.ca/jspui/bitstream/2453/4892/1/McKitrickP2021mp-1a.pdf
- 222. Bach-Mortensen AM, Lange BCL, Montgomery P. Barriers and facilitators to implementing evidence-based interventions among third sector organisations: a systematic review. Implement Sci. 2018 Dec;13(1):103.
- 223. Chewinski M, Corrigall–Brown C. Channeling advocacy? Assessing how funding source shapes the strategies of environmental organizations. Soc Mov Stud. 2020 Mar 3;19(2):222–40.
- 224. Grover JG. Challenges in Applying Indigenous Evaluation Practices in Mainstream Grant Programs to Indigenous Communities. Can J Program Eval. 2010;23(2):18.
- 225. Clarke VA, Frankish CJ, Green LW. Understanding suicide among indigenous adolescents: a review using the PRECEDE model. Inj Prev. 1997 Jun 1;3(2):126–34.
- 226. Callison C, Ludbrook A, Owen V, Nayyer K. Engaging Respectfully with Indigenous Knowledges: Copyright, Customary Law, and Cultural Memory Institutions in Canada. KULA Knowl Creat Dissem Preserv Stud [Internet]. 2021 Jun 23 [cited 2022 Feb 26];5(1). Available from: https://kula.uvic.ca/index.php/kula/article/view/146
- 227. Saini M, Roche S, Papadopoulos A, Markwick N, Shiwak I, Papadopoulos A, et al. Promoting Inuit health through a participatory whiteboard video. Can J Public Health. 2020 Feb;111(1):50–9.
- 228. Hughes S, Pennington J. Autoethnography: Process, Product, and Possibility for Critical Social Research. SAGE Publications, Inc; 2017. 194 p.
- 229. Anderson L. Analytic Autoethnography. J Contemp Ethnogr. 2006 Aug;35(4):373-95.
- 230. Smith B, Sparkes AC. Narrative and its potential contribution to disability studies. Disabil Soc. 2008 Jan;23(1):17–28.
- 231. Ng SL, Wright SR, Kuper A. The Divergence and Convergence of Critical Reflection and Critical Reflexivity: Implications for Health Professions Education. Acad Med. 2019 Aug;94(8):1122–8.
- 232. Freshwater D, Rolfe G. Critical reflexivity: A politically and ethically engaged research method for nursing. NT Res. 2001;6(1):526–37.

- 233. Viatori MStefan, Ushigua Gloria. Speaking Sovereignty: Indigenous Languages and Self-Determination. Wicazo Sa Rev. 2007;22(2):7–21.
- 234. Hunter A, Richards J, Ali-Joseph A, Camplain C. Between Two Worlds: Impacts of COVID-19 on the AI/AN Health Research Workforce. Am Indian Alsk Native Ment Health Res [Internet]. 2022 [cited 2022 Oct 31];29(2). Available from: https://coloradosph.cuanschutz.edu/research-and-practice/centersprograms/caianh/journal/volume29#Issue-2
- 235. Ownership, Control, Access and Possession (OCAPTM): the Path to First Nations Information Governance (Paper) [Internet]. Ottawa, ON, CA: First Nations Information Governance Centre; 2014 [cited 2022 Nov 1]. Available from: https://fnigc.ca/wpcontent/uploads/2020/09/5776c4ee9387f966e6771aa93a04f389_ocap_path_to_fn_informati on_governance_en_final.pdf
- 236. Valdez ES, Gubrium A. Shifting to Virtual CBPR Protocols in the Time of Corona Virus/COVID-19. Int J Qual Methods. 2020 Jan 1;19:160940692097731.
- 237. The Digital Divide: Broadband Connectivity in Inuit Nunangat [Internet]. Inuit Tapiriit Kanatami; 2021 [cited 2022 Nov 3]. Available from: https://www.itk.ca/wp-content/uploads/2021/08/ITK_Telecomms_English_08.pdf

Appendix A – Health Canada Grant

i. Application

Health Santé Canada Canada

PROTECTED A WHEN COMPLETED

Substance Use and Addictions Program (SUAP) Application Form

2021 Call for Proposals

Completing the form

Health Canada collects information for the purpose of evaluating funding applications for contributions. The information contained in the Application for Funding may be accessible under the provisions of the Access to Information Act. All personal information will be protected in accordance with the Privacy Act.

The SUAP Guidelines for Applicants is available at:

www.canada.ca/en/health-canada/services/publications/healthy-living/substance-use-addictions-program-call-for-prop osals-guidelines-applicants-2021.html

All fields marked with an asterisk * are mandatory.

This PDF form is intended to be completed and submitted using a computer. Please do not print it, fill it in and send it by other means.

The description fields on this form use scroll bars and are limited only by the number of characters specified in each field.

This document includes the following sections:

How to submit your proof of eligibility and the application form

Section 1 – Organization identification

Section 2 - Funding priority(ies)

Section 3 - Project target population(s)

Section 4 - Project information

Section 5 – Evidence and need

Section 6 – Performance measurement and evaluation

Section 7 - Organizational and collaboration capacity

Section 8 - Sex-and gender-based analysis

Section 9 - Summary work plan

Section 10 - Budget and narrative

Section 11 - Approval / Declaration

How to submit your proof of eligibility and the application form

Submitting your proof of eligibility - You will find a button to attach and submit your proof of eligibility in Section 1 – letter (e). When clicking on the button to select the files at your computer, your PDF reader (Foxit, Adobe Reader, etc.) may show a warning message to add the current file to a privileged location or enable all features. Please add this file to a privileged location or enable all features so you can attach the requested documents along with your application form.

Submitting your application form - After filling all the required fields and attaching the requested documents, please click on the "Submit Application" button. Your email client will open with a copy of your filled application form.

HC SUAPCFP2112 (2021-05-001) E

Page 1 of 11



	*	Health Canada	Santé Canada	Submit Form	Reset Form	PROTECTED A WHEN COMPLETED			
Se	ection 1 – C	Organization	identification						
Or	rganization	identificatio	on						
				t appears on legal docume	ents				
b)	Operating	g name If diffe	erent from legal nar	ne					
	Primary a Street nur Building 359	mber and nar	ne*		City or Town*				
3	11 L	or Territory*			Postal code* Telephone number* Extension 257				
	Organizat	ion e-mail ad	ldress*		Website address				
d)	Primary (Full name E-mail ad	*	n		Position* Councillor Telephone numbe	er* Extension			
e)	be found.	Documents		zation's proof of eligil vhere eligibility inform	8 75 083759 - N 12858385	web link(s) where eligibility information can			

Section 2 – Funding priority(ies)*							
Please select all applicable funding priorities which apply to your project:							
Generation Funding priority 1	Funding priority 4						
Funding priority 2							
Funding priority 3							

HC SUAPCFP2112 (2021-05-001) E

Page 2 of 11



Health Santé Canada Canada	PROTECTED A WHEN COMPLETED
Section 3 – Project target populations(s) Please select all the	at apply
 a) Target group(s)* A Indigenous Peoples Racialized peoples and communities 2SLGBTQIA+ peoples and communities Youth People living with pain People who use illegal and toxic drugs People who work in the trades or in physically demanding professions Men aged 25 to 59 People who use drugs alone People in poverty or experiencing economic insecurity People experiencing housing insecurity or homelessness 	
 People experiencing nousing insecurity of nonnelessness People who have been in contact with various justice systems People with lived and living experience of substance us as well as their families 	
b) Official language(s)* c) Gender(s)* ⊠English ⊠Male □French ⊠Female □Both ⊠Non-binary	d) Age group(s)* ∑15 - 24 years 25 - 34 years 35 - 44 years 45 - 54 years 55 - 64 years 65 - 74 years 75 years and over
e) Total anticipated number to be reached* Insert only nu	imbers
will be involved in the project. max. 1000 characters with spaces This project's target population is considered to be at moderate or high risk of harmful substance use (e.g school, have a history of conduct disorder, etc.) Every effort will be mad needs. For example, youth may be involved during data collection and a	t population(s), including both primary and secondary audiences and how they The primary target population is the community's youth, particularly those , youths that are marginalized, have mental health concerns, have dropped out of e to have youth involved in this project's activities based on their interests, skills and analysis, knowledge translation, or in 'pilot activities'. The secondary audience is the ussion surrounding the project's methodology, knowledge translation, and impact of

HC SUAPCFP2112 (2021-05-001) E

Page 3 of 11





PROTECTED A WHEN COMPLETED

Section 4 – Project information

Health

a) Project title*

Using community-driven and process-based research for the development of initiatives that prevent and reduce harmful substance use among youth in

b) Project duration and requested funding amount

Duration* In months 15

Total requested* Only numbers

172010.10

c) List other sources of confirmed and/or anticipated funding (cash and/or in-kind contributions) for the project

Source Type	e the source's legal name	Amount Only numbers	Cash	In-kind ©	
	facility rental	C\$ 9,700.00	0		
	admin fee contribution (5%)	C\$ 7,477.90	0		
Rachel Cassi	e, coordinator salary	C\$ 51,520.00	0	۲	
Business and	Hospitality Services - accommodations	C\$ 12,320.00	0	۲	
			0	0	
			0	0	
			0	0	
			0	0	

Additional sources Use this field to provide additional sources if needed. Max. 1000 characters with spaces

Community volunteers

d) Project summary* Briefly summarize the proposed project. Max. 4000 characters with spaces

This project will be done in partnership between and the University of Alberta's School of Public Health. It is a multi-faceted project aiming to prevent harmful substance use by building capacity in the community's youth. The project will: 1) Scan and identify the needs of youths at risk for substance abuse and 2) identify activities which reduce the risk of substance abuse among vouth.

The research will explore youths' interests in program options and delivery strategies. A community-driven and process-based research methodology will ensure research findings will result in recommendations for effective, practical and meaningful initiatives to address the risk of harmful substance use among youth in the community. This project will directly engage youth to ensure the outcomes of the research are meaningful and relevant. All research will be guided by a community steering team including Elders, community leaders, and concerned community members and youth.

e) Objectives* Describe the objective(s) of the proposed project, including a description of the nature of the problem that the proposed project will aim to address. Max. 4000 characters with spaces

the impacts of colonization have resulted in widespread intergenerational trauma contributing to poverty, food insecurity, and adverse childhood events or Elder care. This trauma has resulted in a break in cultural connectivity and social sphere in which many youth experience underdeveloped life skills and increased likelihood of harmful substance use and criminal activity. While much research has been conducted on intergenerational trauma's role in harmful substance use in Indigenous populations, little research has been done on how to address its effects

In 2018, a Community Needs Assessment was conducted in the impact of intergenerational trauma was identified as a key area of concern. Specifically, concerns for the community's widespread poverty and the prevalence and extent of harmful substance use among youth were highlighted by many community members. Preliminary work with the community, including Elders and community leaders, has confirmed that this concern has only increased since 2018.

Meaningful community-based research that scopes and scans the strategies to build capacity and reduce risk factors for harmful substance use among youth are needed to address the impact of intergenerational trauma. Community-specific research is needed as programs need to be tailored to a) Are of

HC SUAPCFP2112 (2021-05-001) E

Page 4 of 11





Health Santé Canada Canada

PROTECTED A WHEN COMPLETED

f)	Key activities* Describe the key activities that will contribute to achieving your project objectives. Max. 4000 characters with spaces
	Throughout this project, meaningful engagement with community members, particularly youth, will ensure relevant program development. This engagement has already begun, and will continue as research and program methodology develop (the second program development). The key research-specific activities are program oriented and include scoping and scanning exercise data collection, data analysis, and knowledge translation. The process will ensure meaningful and respectful engagement with the community. All research materials related to promotion, data collection, and knowledge translation. The collection and analysis stages. Data collection strategies have not yet been finalized, however, we envision youth to be actively involved through activities such as creating and administering surveys, conducting interviews, or photovoice strategies; we aim to collect both quantitative and qualitative data to best inform our program development. We are also planning to meaningfully engage youth and Elders at the data analysis stage. This engaged, community-based approach aims to not only generate richer data and understanding, but also actively engage the community sy outh in an effort to prevent harmful substance use in younger generations. Critical to respectful and meaningful community-based research, we are planning to utilize integrated knowledge translation. This will entail ongoing discussio
g)	Key outputs* Describe the key outputs (knowledge products and/or learning opportunities) that will be produced. Max. 4000 characters with spaces
	Key knowledge products from the initial research will include descriptions of: 1) is youth's capacity including life skills such as communication and problem-solving, and 2) approaches, such as services or programs, that build capacity to address the risk factors associated with harmful substance use in the prevalence and characteristics of youth's harmful substance use in the will use a process which trains and engages youth to support them as they create and implement effective upstream approaches to prevent harmful substance use. Better understanding of the characteristics and risk factors of harmful substance use among youth in the understanding of the characteristics and risk factors of harmful substance use among youth in the advisory group will seek continual learning opportunities through community engagement throughout program development and implementation. The team will co-develop with youth in multiple formats including in-person group or individual discussion and online open-response forums or surveys to enable anonymity, if desired. Information from youth on program development and implementation will aim to learn from parents, grandparents,
h)	Geographic location(s)* List the cities and provinces/territories in which the project will be implemented. Max. 1000 characters with spaces Nunavut
i)	Project setting(s)* Describe the setting(s) in which the project will be implemented. Max. 2000 characters with spaces
	is located on , in the Nunavut region of . This community has a population of around impeople, with about one third of the population under the age of fourteen. The rates of illicit substances entering the community and crime associated with substance use are increasing. Of note to this project, anoted above, there are many service gaps in . As noted above, there are many service gaps in . including programming for youth. . and
j)	Scope* Please select one
	O National
	O Province/Territory
	Community

HC SUAPCFP2112 (2021-05-001) E

Page 5 of 11





Health

Official language requirements* Describe how the project complies with the Official Language Requirements outlined in the SUAP I) Guidelines for Applicants. Max. 2000 characters with spaces

Nunavut has three official language communities: Inuit languages, English, and French. French is rarely spoken in , especially among youth and Inuit. dialect of Inuktut, is the Inuit language indigenous to . All research promotion, data collection, and knowledge translation materials will be available in English and . Oral interpretation will occur at all community All research meetings related to research or program development. All translation and interpretation will be done by paid community members. We also intend to have a unilingual Elder on the advisory committee to ensure the advisory committee to ensure the second seco

Encouraging usage among Inuit youth has been recognized to be an important part of continuing culture and intergenerational healing. Exploring

m) Sustainability* Health Canada funding is time-limited. Please indicate what measures you will take to ensure that needs addressed by the project are met through alternate measures after the end of the funding period. Please note any anticipated enablers or barriers related to sustainability. If you anticipate that project activities will not continue once the funding period ends, explain what steps you will take to ensure a smooth project wind-down. Max. 4000 characters with spaces

This project aims to build youths' capacity by developing resilience and communication and problem-solving skills. Sustainability is therefore inherent to this project because these programs will facilitate growth of youths' capacity, resulting in lifelong skills youth can utilize to meaningfully contribute to the community through healthy relationships, community leadership, and employment.

The reciprocal partnership between and the University of Alberta will allow for long-term projects that will ultimately have meaningful, lasting contributions to the community. If there is community interest, students could apply for scholarships and grants offered by Canadian Institute of Health Research or other institutions to continue exploring ways to address the risk factors associated with youth engaging in harmful substance use in

, thus the community will determine the most effective usage of funds. Community-determined usage of This funding will be held by

Section 5 – Evidence and need

a) Evidence and need* Describe the evidence and need (overdose data, service gaps, research evidence, statistics, surveys, literature reviews, needs assessments, etc.) for the project's activities, communities of focus and target populations. Explain, if applicable, how your community is underserved. Max. 4000 characters with spaces

As stated previously, a Community Needs Assessment was conducted in 2018. A key area of concern that was identified by many community members was the impact of intergenerational trauma. Many community members noted related concerns for the community's widespread poverty and the prevalence and extent of substance use among youth. Additionally, RCMP reports have shown increasing rates in both alcohol and drug-related criminal activity, as well as illicit drugs entering the community; based on anecdotal evidence, youth involvement in criminal activity and use of illicit drugs appears to be increasing.

After a thorough literature review, there is clear evidence that Indigenous communities in Canada need innovative and culturally-relevant and respectful approaches to heal from intergenerational trauma caused by Canada's historical and ongoing colonization techniques. As highlighted by the Community Needs Assessment, increased rates of harmful substance use in Indigenous communities are one of the main manifestations of the trauma caused by Canada's historical and ongoing colonization.

is underserved on many fronts: healthcare, education, children and youth services to name a few. While much of the current research has focused on the underlying causes of harmful substance use in Nunavut, namely intergenerational trauma, little research and work has been done to address this phenomenon. We have the Community Needs Assessment and literature review to draw on, however, it is not enough to adequately inform new approaches to care. The proposed research will utilize innovative methodologies to generate youth-specific data, thus providing the high quality data that

b) Similar initiatives or new approach* Describe how this project complements or builds on other similar initiatives; or, describe how this project is an innovative/new approach to health promotion, prevention, harm reduction or treatment. Max. 4000 characters with spaces

This project will make use of existing community spaces including the and Centre. This project will complement and build upon previous and concurrent research with other communities in Nunavut, including the work being done at the territorial level. However, this project will utilize novel community-based methodologies and approaches that set it apart from other work currently being undertaken in Nunavut in the area of harmful substance use. To the best of our knowledge, the initial scanning and scoping project is the first of its kind in Canada to utilize community-based research methodology to explore capacity-building programs for youth aimed at addressing the risk factors for harmful substance use. Further, this is the first joint community scan and program development initiative in Canada in which the data will be generated by Indigenous youth and will directly impact the provision of preventative services for Indigenous youth, to the best of our knowledge.

This project will foster a new partnership between the University of Alberta and This is a partnership grounded in reciprocity. respect, and relevant contributions. Going forward, this partnership provides great potential to build capacity in the second se ultimately taking an

HC SUAPCFP2112 (2021-05-001) E

Page 6 of 11





Health

Section 6 – Performance measurement and evaluation

Contribution to SUAP indicators and evaluation requirements* Describe how your project aligns with SUAP outcomes listed in the SUAP Guidelines for Applicants, including appropriate indicators, noting data collection methods. Also explain how the project will positively impact SUAP outcomes. Max. 4000 characters with spaces

This project will increase both immediate and intermediate SUAP program-level outcomes and performance indicators. We aim to increase in-community capacity during both the scanning and program implementation phases. During the scanning phase, we aim to increase community capacity by working with youth throughout the scanning and scoping process. This will increase both the youth's understanding of this process and facilitate deeper understanding of the importance of capacity-building and community engagement; youth-researched programming will be inherently more tailored to the interests of today's youth, thus more effectively reaching our programs' goals. Training of community-based program leaders will increase system is youths' services in thus more effectively reaching our programs goals. Training or community-based program leaders will increase is youthn's services in a culturally-relevant and respectful manner. It will also provide in-community, paid leadership opportunities in a community experiencing devastating poverty. We will collect data to measure program impact on youths' capacity by: 1) recording the number of youth who participate in the research process, 2) interviewing youth to learn about their research experience, 3) recording the number of youth attending the implemented programs, 4) continually seeking feedback from youth on implemented programs, and 5) seeking feedback from adults on the implemented programs. We will also measure increases in the community's leadership capacity by interviewing program leaders on their experience.

As noted throughout this proposal, the scanning and scoping phase will inform us during the program and evaluation development stage on the programming needs and interests of youth in the scanning and scoping phase will function to increase youths' capacity, acting as an upstream approach to prevent harmful substance use. During the program implementation phase, the number of relevant and meaningful programs for youth will increase. Specifically,

Section 7 - Organizational and collaboration capacity

How is your organization well-positioned to carry out this work?* Describe why your organization is best positioned to a) undertake this project. Include information about yourorganization's relevant skills, interests, experience with subject matter and target population(s), financial and/or human resource capacity. Max. 4000 characters with spaces

and the University of Alberta are well-positioned to carry out this work because both organizations are independently committed to and passionate about creating upstream approaches to address harmful substance use in the substance is youth. Together, these institutions are well-positioned to build off each other's strengths to effectively implement programs that address this area of concern. This new relationship is built on the principles of meaningful community-based research that align with IQ principles including: community-ownership, reciprocity, respect, and relevance. To ensure our project embodies these principles, a diverse group of community members will be engaged throughout the research and program development processes including Elders, community leaders such as Council members or school leadership, and people who have been impacted by intergenerational trauma. Many community members have expressed concern over the increasing impact of harmful substance use on youth in this was also highlighted in the Community Needs Assessment conducted in 2018. Critical to this project's relevance and meaningful contribution is that it addresses knowledge and service gaps voiced by the community.

The University of Alberta is a leading knowledge institution in the area of transforming community-based research. Dr. Susan Chatwood, associate professor at the University of Alberta and the main faculty member connected with this project, has worked with remote and northern Indigneous communities for most of

b) Indicate below the names of the partners you will work with during the project and describe their role and contribution.

Name of partner organization Type the partner's legal name	Partner's role
University of Alberta, School of Public Health	Scanning and scoping exercise, health promotion w
Additional partners Use this field to provide additional partners if needed. N	

HC SUAPCFP2112 (2021-05-001) E

Page 7 of 11



Section 8 - Sex and gender-based analysis

Health

Integration of sex and gender-based analysis* How has sex and gender-based analysis (SGBA) been integrated into the project, including data collection? For reference, see the continuum of gender considerations in programs and policies in the SUAP Guidelines for Applicants. Max. 2000 characters with spaces

This project will be inclusive to all genders, including those beyond the binary, and will account for gender-based needs and differences. Accounting for and supporting gender-based needs is well-established in modern Inuit communities; for example, gender-based support such as women's sewing groups and supporting gener-based needs is weinestabilised in modern much communities. During preliminary discussions with an Elder, it was shared that in their experience, this gender division is largely a result of colonization and support groups prior to colonization typically involved the whole community. This is a good example of the importance of designing our programs to have the potential to transform harmful gender norms. There will likely be some interests that follow gendered themes, but it is important to create a space for youth to share and create gender transformative programs. Prior to data collection, we will consult with Elders and other knowledge keepers to ensure we are being respectful of cultural gender needs during data collection. Throughout the data collection process, we plan to explore gender-based differences related to programs that encourage capacity building in youth

to prevent harmful substance use. Project plans specifically related to gender considerations include: 1) exploration of gender-based substance use patterns including setting, substance(s) used, and physical or social impact, and 2) gender-based needs for capacity-building programs. For example, a future research topic could explore the prevalence and experience of gender-based violence in the context of familial substance use and programs that will foster youths' skills

Section 9 – Summary work plan

Activities*	Start date*	End date*	Outputs*
Establishment of Community Partnership	2022-01-01	2022-02-01	Establish partnership understandings and respon
Community Steering Committee Formation	2022-01-01	2022-02-01	Create a community steering committee compose
Formation of Youth Research Committee	2022-01-01	2022-02-01	Creation of Youth Research Committee and esta
Survey Creation and Administration with Youth	2022-02-01	2022-03-01	Help youth creation/administration of a survey to
Training Youth to Analyze Quantitative Data	2022-02-01	2022-02-01	Train youth to input quantitative data into data ma
Qualitative Research Methodology and Analysis Tra	2022-02-01	2022-02-01	Training data collection methods (eg. interview fu
Iterative Qualitative Research Collection by Youth	2022-02-01	2022-12-01	Continually collect data throughout activity planni
Qualitative Data Analysis by Youth	2022-03-01	2023-01-01	With youth, analyze qualitative data to determine
Research Presentation by Youth for the Community	2022-04-01	2022-04-01	Put together and present research findings on the
Program and Activities Development	2022-04-01	2022-06-01	Development and logistical planning of programs
Pilot Activities Implementation	2022-06-01	2022-12-01	Implementing youth pilot projects.
Feedback from Youth on Programs	2022-06-01	2022-12-01	Actively engaging with youth on program content
Program Evaluation	2022-12-01	2023-03-01	Evaluating implemented programs' effectiveness
Survey Re- Administration by Youth	2023-01-01	2023-01-01	Re-administer survey to determine statistics asso
Formal Research Paper	2023-02-01	2023-03-01	Writing the formal research paper to be published
Future Research Projects	2023-03-01		Future projects may include programs to help you

Additional activities Use this field to provide additional activities if needed. Max. 1000 characters with spaces

The activities will include 4 1-week on-the land programs, 2 1-week workshops on introductions to electronics, computers and robotics, 8 weekly 4-hour "Drum-Beat" activities, 8 weekly contemporary Inuit throat-singing/recording session (2 groups, i.e. singers and recorders). 8 weekly traditional Inuit and contemporary cooking classes, 8 weekly sewing sessions for traditional clothing and 8 weekly fashion and e-pattern making sessions, 8 weekly social media story sharing/ creation sessions, 8 communication life-skills sessions (i.e. problem solving skills, communications, planning and visioning exercises.

Page 8 of 11



I ┿ I Health Santé Canada Canada

PROTECTED A WHEN COMPLETED

	Federal fiscal year (FY) = April 1 to March 31	o March 31		
Budget expenditure categories	Narrative (Max. 1000 characters with spaces/narrative)	FY 1	FY 2	Totals
Personnel / Salaries and Benefits	Research Assistant 20 hrs/wk @ \$25/hr + \$13 NA + 14% Mand. Ded. Activity Coordinator 20 hrs/wk @ \$25/hr + \$13 NA + 14% Mand. Ded.	C\$ 10,397.00	C\$ 65,847.00	C\$ 76,244.00
Contractual employees	Outlitter with boat @ \$800(day x 12 days 3 came helpene @ \$5700(day x 12 days Edite & Adv. Mmn. Honoratia Interpreting a \$2000mg. Translation x 40pg x 100 pps	C\$ 3,700.00	C\$ 37,040.00	C\$ 40,740.00
		c		
Travel & Accommodations	Res Coordinator airfare \$4,000/visit Res Supervisor airfare\$4,000/visit Travel perdiems – Coord & svsr Accom. 6 nights enroute	C\$ 6,282.00	C\$ 16,564.00	C\$ 22,846.00
Materials & Supplies	Flipcharts, sewing materials, participant rewards, etc.		C\$ 5,000.00	C\$ 5,000.00
Equipment	Cameras, various activity equipment (eg. cloth for sewing)		C\$ 1,000.00	C\$ 1,000.00
Rent & Utilities	Enroute + = = In-kind			C\$ 0.00
HC SUAPCFP2112 (2021-05-001) E	-05-001) E Page 9 of 11		Ü	Canadä

132

PROTECTED A WHEN COMPLETED	C\$ 0.00	-	C\$ 0.00	C\$ 288.00 C\$ 3,456.00 C\$ 3,744.00	al C\$ 4,487.22 C\$ 17,948.88 C\$ 22,436.10	C\$ 0.00 C\$ 0.00 C\$ 0.00	C\$ 0.00 C\$ 0.00 C\$ 0.00	C\$ 0.00 C\$ 0.00 C\$ 0.00	C\$ 0.00 C\$ 0.00	C\$ 0.00	dget C\$ 25,154.22 C\$ 146,855.88 C\$ 172,010.10	Validate Grand Total	
Santé Canada	Rachel Cassie salary (In kind)		in kind	All participants, meetings & events = \$ 12/particpant	15% of total requested money from Health Canada (admin fees will total \$29,914.80; we are requesting \$22436.10 from Health Canada to cover this)	Facility Rental \$100/event (\$9700 total)	Res. coord salary (\$51,520)	Accom coord & spvr – (\$12, 320)	5% (20% of total = admin costs; will provide \$7477.9)		Total contributions from Health Canada - Total budget		
Health Canada	Performance Measurement & Evaluation	Other	Room/Space rental	Food	Administration	In-Kind	In-Kind	In-Kind	In-Kind				



Section 11 – Approval / Declaration*

Health

The undersigned on behalf of the organization declares that:

• The information in this application and all accompanying documents are accurate and complete;

• No current or former public servant for whom the Health Canada Values and Ethics Code, the Values and Ethics Code for the Public Sector, the Treasury Board Secretariat Policy on Conflict of Interest and Post-Employment and the Conflict of Interest and Post-Employment Code for Public Office Holders applies, shall derive any direct benefit from this funding request including any employment, payment or gifts, unless the provision and receipt of such benefits is in compliance with such codes and policy.

. Individuals who are subject to the provisions of the Conflict of Interest Act (S.C. 2006, c. 9, s. 2), the Conflict of Interest Code for Members of the House of Commons, the Conflict of Interest Code for Senators, the Conflict of Interest and Post-Employment Code for Public Office Holders or any other values and ethics codes applicable within provincial or territorial governments or specific organizations, cannot derive any direct benefit resulting from this funding request unless the provision or receipt of such benefit is in compliance with such legislation and codes;

· Project activities will be undertaken in compliance with all applicable statutes, regulations, orders, standards and guidelines;

· Project activities are not being supported, directly or indirectly (including through funding or in-kind contributions), by entities involved in the manufacture, production, advertising or sale of pharmaceutical, tobacco, vaping or cannabis products; and

• The funding request is made on behalf of the organization named in Section 1 with its full knowledge and consent.

I acknowledge that Health Canada may share this funding request with other organizations (including provincial/territorial Ministries of Health) as part of its review and approval process. I acknowledge that should this funding request be approved, funding will be conditional upon the organization signing a written agreement with Health Canada.

Name:	Title: Mayor of
Signature:	_ Date:

Submit Form **Reset Form**

HC SUAPCFP2112 (2021-05-001) E

Page 11 of 11


Health Santé Canada Canada

APPENDIX A

OVERVIEW <</MANDATORY FOR SUAP>>

A high-level description of the project (approximately 2 paragraphs) including: - Project title

An outline of the objective of the project (XYZ organization will do WHAT for WHOM to achieve WHAT outcome).

Note: In most cases, the Project Description in Section 4 of the FAF may be used.

Suggested structure for the first sentence:

The (LEGAL ORGANIZATION NAME) proposes to (MAIN OBJECTIVE OF THE PROJECT) through the (INITIATIVE TITLE) project.

SAMPLE: The Harm Reduction Network of Sesame Street proposes to decrease the effects of the opioid crisis on the youth of Sesame Street through the Harm Reduction through Community Arts Engagement project.

proposes to prevent harmful substance use among youth through the 'Using community-based research to prevent harmful substance use in youth in project. The proposed project will be done in partnership between and the University of Alberta's School of Public Health. There are two complementary facets of this project: 1) research community needs, and 2) program implementation.

Initially, we will explore how to safely research substance use among youth. Then, we will explore youths' interests in program options and delivery strategies. Community-led research methodology and methods will be used. Ongoing, community-led discussions surrounding research goals, methods, and outcomes will occur to ensure research relevancy and safety.

Research will be used to inform future health promotion programs that aim to address the intergenerational trauma experienced by youth. For example, programs explored during this initial research may include arts- or sports-based programs, or on-the-land programming that fosters a deeper connection to traditional Inuit culture. These programs will be tailored to the community's youth, use modern strategies to engage youth, and be grounded in Inuit cultural values. These programs will aim to build youth's capacity by fostering healthy communication and problem-solving skills, resilience, and programs, and explore the role of culture in program effectiveness and community interest in additional programs.

KEY ACTIVITIES AND DELIVERABLES/OUTPUTS <</mail>

The following is a definition for each table heading:

Fiscal Year: timeframe during which key activities will be carried out; the timeframe can be further defined if desired (i.e., quarters, specific dates)

Key Activities: key operations or tasks that require a significant amount of resources (i.e., time, money). For example, key activities can address:

-

- -
- Sustainability Knowledge Translation and Exchange (KTE) Performance Measurement
- Evaluation

Key Deliverables/Outputs: direct products or services stemming from key activities

NOTE: following table format is optional as long as mandatory items are addressed

Fiscal Yes	ar 2022-23
Key Activities	Key Deliverables/Outputs
Community-based research to explore ways to safely research sensitive topics (eg. substance use or trauma).	Research framework document describing how to safely research sensitive topics with youth. This document may outline what safeguards need to be in place, who to involve, how to gain consent, and other important aspects of research. This document will be used for subsequent research in this project and future research projects with This document will be owned by available in English and and use plain language.
Community-based research on: 1) substance use among youth and, 2) youths' interests in health-promoting initiatives such as on-the- land camps or recreational activities.	Report discussing substance use among youth in and youths' interests in health promotion initiatives. This report will be owned by the available in English and and use plain language. These findings will inform in what activities youth in are interested.
Planning health promotion initiatives that were of interest to youth with Elders, youth, and other community members.	Begin planning logistical aspects and recruiting people to implement the health promotion initiatives. For example, for an on- the-land camp, this includes hiring staff to implement the camps, planning dates for the camp, meetings with Elders and other community members, etc. We will create living documents to track each initiative's progress.
Counselling	Counselling will be provided to adults and youth participating in the research phases. This is a critical component of ensuring the research mitigates any potential harm.
Fiscal Yes	ar 2023-24
Key Activities Continue planning health promotion initiatives.	Key Deliverables/Outputs Planning logistical aspects of initiatives (leaders, location, participant capacity, food, transportation, etc.). We will update the living documents as we continue to plan.
Implement health promotion initiatives.	Carry out the health promotion initiatives (eg. run the camps).
Counselling	Counselling will be provided to youth participating in the activities.
Evaluate health promotion initiatives.	Evaluate the youths' and families' experiences of the initiatives through interviews and other data generation strategies. Create a report summarizing findings and highlighting strengths and opportunities for improvements.

	This report will be owned by
	available in English and and use plain language.
Knowledge sharing.	Knowledge sharing with the larger community. For example, a community feast and/or presentation. The report will be presented orally in English and full-length and summarized versions will be available.

OUTCOMES

This initiative will contribute to and report on the SUAP outcomes and performance indicators that are relevant to their project.

	SUAP OUTCOMES		SUAP PERFORMANCE INDICATORS
	1.1 Primary target population (Canadians ^{1[1]} or Stakeholders ^{2[2]}) have access to information on substance use	a)	# of participants reached (by gender, language, age, location or priority population: Indigenous, 2SLGBTQIA+, Racialized people/communities) in substance use-related learning opportunities delivered by SUAP funded project
S H		b)	# of primary target population (Canadians or stakeholders) with access to substance use-related knowledge products developed by SUAP funded project
O R T - T E		c)	% of primary target population (Canadians or stakeholders) who reported that they gained knowledge/skills about substances as a result of the SUAP funded project by gender, language, age, location or priority population: Indigenous, 2SLGBTQIA+, Racialized people/communities
R M	1.2 Increased availability of harm reduction services ^{3[3]}	a)	# of new services offered by SUAP funded project (including type of service)
			# of clients accessing services offered by SUAP funded projects (by gender, language, age, location, or priority population: Indigenous, 2SLGBTQIA+, Racialized people/communities)
		c)	# of Naloxone kits distributed by SUAP funded project
	2.1 Data and research evidence on drugs, substance use, and emerging drug trends are used by targeted stakeholders	a)	% of targeted stakeholders who reported that they used knowledge/skills related to substance use provided by this project to make evidence informed improvements to policies, programs and practice
M E D I	1088 1032 4	b)	% of targeted stakeholders who reported that they intend to use knowledge/skills provided by this project to make evidence informed improvements to policies, programs and practice
U M - T E	2.2 Reduction in risk-taking behaviour among people with problematic drug and substances use	a)	% of targeted Canadians who reported that they used knowledge/skills related to substance use provided by SUAP funded project by gender, language, age, location, or priority population: Indigenous, 2SLGBTQIA+, Racialized people/communities)
R M		b)	% of targeted Canadians who reported that they intend to use knowledge/skills related to substance use provided by SUAP funded project (by gender, language, age, location, or priority population: Indigenous, 2SLGBTQIA+, Racialized people/communities)

^{1[1]} Targeted Stakeholders may include other levels of governments, pan-Canadian health organizations, non-profit organizations, communities or others at the organizational or system level. This relates to initiatives targeting organization, system or policy and practice change. ^{2[2]} Targeted Canadians may include youth, adults and any other target groups or sub-populations as identified by research and evidence. This relates to initiatives targeting individual behaviour change (or groups of individuals) ^{3[3]} Services Categories currently include: harm reduction / outreach; early intervention; treatment (OAT, iOAT, TiOAT, methamphetamine treatment, nicotine vaping cessation); drug checking; safer supply; surveillance.

	c)	% of clients in SUAP funded treatment and/or harm reduction services projects reporting a positive behaviour/health outcome by gender, language, age, location, or priority population: Indigenous, 2SLGBTQIA+, Racialized people/communities
3.1 People who use drugs and substances experience reduced negative health effects	a)	N/A

PARTNERSHIPS <</MANDATORY>>

Identify key partners and the role each will play in carrying out key activities. A key partner is typically an organization that plays a central role in the implementation of activities and/or that has a vested interest in the successful implementation of activities.

Name of Partner Organization	Partner's Role
University of Alberta	People involved from the University of Alberta include: Rachel Cassie, MSc in Epidemiology candidate; Dr. Susan Chatwood, Associate Professor, School of Public Health. They will be most directly involved during the research phases and report compilation. They will also be available to assist in other ways as needed/wanted.
	Partnership parameters are not yet defined, but likely will assist during research and activity implementation. May also be involved during knowledge sharing activities.

TARGET GROUPS <<<u>MANDATORY</u>>>

Describe the target groups, both direct and indirect, and how they will be reached.

This project's target direct target group is the youth in the

or Youth will be actively involved throughout the research phase and will be the direct program participants. It is a small community of Youth will be reached through community spaces, parents, and snowball sampling (word-of-mouth).

The indirect target group is the larger community. The larger community will be involved through ongoing community discussion around safe research methods, desired program options, and program evaluations. The larger community will be reached through community spaces and snowball sampling.

OFFICIAL LANGUAGES <</MANDATORY>>

Describe how both linguistic communities will be targeted.

Where applicable based on the proposed Initiative, consider the following:

- *i)* Is the targeted population composed of individuals or groups belonging to both linguistic communities?
- *ii)* In the general public, what is the level of potential demand for services in both official languages?

- iii) If, by its very nature, the Initiative is aimed at only one linguistic group, could it be promoted to the other group to encourage increased understanding between the two groups?
- iv) Is the scope national and likely to reach both language groups?
- v) Do the key activities relate to the visibility of Canadian identity abroad?
- vi) Is the recipient a national organization that should clearly have a concern for official languages?

If the answer is yes to any of the above questions, both linguistic communities should be targeted.

Example:

- The acknowledgement of the Department's support will be expressed in English and French;
- All public signage, communication (oral and written), products and services will be developed and offered in English and French; and
- The official-language minority community will be invited to participate where appropriate.

<u>OR</u>

Provide a justification if both linguistic communities will NOT be targeted.

Example:

- The Initiative is designed to address the specific needs of the Francophone population in Manitoba, and as such it does not target the Anglophone population. Or
- The Initiative is piloting a new training approach and curriculum and would like to target a single linguistic community before adapting them and investing in translation.

Nunavut has three official language communities: Inuit languages, English, and French. French is rarely, if ever, spoken in the especially among youth and Inuit. In the second s

Encouraging usage among Inuit youth has been recognized to be an important part of intergenerational healing and cultural continuity. Exploring youths' interest in program delivery in will be an important component of the research phase; if there is hesitancy, root causes of this hesitancy will be explored to see if this hesitancy can be mitigated. Where possible, language during program delivery will occur.

SEX AND GENDER <</MANDATORY>>

Describe how the Initiative will take sex and gender into account.

Where applicable based on the proposed Initiative, consider the following:

i) Briefly describe the sex and gender context in relation to the Initiative. E.g. Are men /women* impacted differently, and if so, how? Does the recipient's focus on one priority over another affect men/women* differently? Is there a disproportionate representation and/or participation of men/women*?

- ii) Describe how key activities and outputs will take into account, or will be tailored to, differences in groups of targeted men/women*. E.g. Is accessibility the same for men /women*? Are outputs, such as reports or training materials, reflective of sex and gendered considerations?
- *iii)* Describe the impact that sex and gender considerations are expected to have in achieving equitable outcomes.
- iv) Describe how knowledge translation and exchange (KTE) strategies have been/will be developed to include all sub-groups being targeted, and to maximize uptake by men/women*. Consider if content/messages/products will need to vary by sex and gender.
- v) Describe how sex and gender-disaggregated data will be collected and used.

*This may also include considerations related to boys/girls and gender diverse individuals. Other social dimensions and experiences such as age and ethnicity, which interact with sex and gender, should also be reflected where relevant.

This project will be inclusive to all genders, including those beyond the binary, and will account for gender-based needs and differences. Accounting for and supporting genderbased needs is well-established in modern lnuit communities; for example, genderbased support such as women's sewing groups and men's groups are an important support element in many communities. During preliminary discussions with an Elder, it was shared that in their experience, this gender division is largely a result of colonization and support groups prior to colonization typically involved the whole community. This is a good example of how we can use IQ principles and Inuit cultural values to transform harmful gender norms caused by colonization.

All genders are impacted by harmful substance use, thus this project will target all youth. It is well-established that men and women experience personal and familial substance use differently, however, these gender differences have not yet been explored in an Inuit community. During research, we will explore gender-based needs and program interests to ensure the programming is inclusive to all.

We will tailor the programming according to what we learn during the research phase. For example, we could develop an on-the-land camp that teaches Inuit hunting and food preparation skills. Depending on youth's interests and Elders' knowledge, these may or may not be broken into gender-based groups. Program evaluations will include gender to reflect program delivery.

We recognize that gender-based interests are present among youth, however, it is important that we also create space for gender transformative programming. For example, we will create space for boys to participate in activities often viewed as feminine, and vice versa. Our gender considerations are expected to create program delivery that is equitably accessible for all youth. We also aim to engage all youth by creating programs that target a wide range of youth's interests.

Prior to data collection, we will consult with Elders and other knowledge keepers to ensure we are being respectful of cultural gender needs during data collection. Throughout the data collection process, we plan to explore gender-based differences related to programs that encourage capacity building in youth to prevent harmful substance use. Research plans specifically related to gender considerations include: 1) gender-based safety needs when researching substance use, 2) gender-based substance use patterns including setting, substance(s) used, and physical or social impact, and 3) gender-based needs for programs.

We will tailor our knowledge sharing based on what we learn from our Program Committee. We plan to specifically discuss if gender-based knowledge sharing strategies are needed, and if they are, ways to facilitate them. We will report on genderbased differences.

iii.

Health Santé Canada Canada

APPENDIX B - BUDGET

		Dudant he	Blood Voor	Budact hu Elevel Vacu (And 1 to March 21)	anal 21)				
		2022-2023	LISCAL L CAL	C T mide	VV02-VV02			Total	JUSTILICATION
BUDGET EXPENDITURE	100	C202-2202		10/10	1 107-1 10		TOTAL	Contribution	
CATEGORIES		09/2022 - 03/31/2023 Other Sources	23 ources		04/01/2023 - 11/30/2023 Other Sources	023 ources	BUDGET	from HC	
	HC	Cash	In-kind	HC	Cash	In-kind			
PERSONNEL SALARIES & BENEFITS	\$14,073			\$56,291			\$76,244	\$70,364	These salaries include the Northern Living Allowance (NLA) of \$13/day. We are requesting the salary for the Fluid Program(s) Coordinator to segmize all positical plantaming of existivity. We are also requesting the Restore Assistance Salary. These one protrime positions at SADhour, for an expected 20-24 hours a work.
CONTRACTUAL PERSONNEL	\$7,548		\$5,400	\$30,192		\$5,400	\$40,740	\$37,740	These funds will pay for honouraria to Elders, translations, and workshop professionals such as leadership experts. Specifically, we are equesting \$500 for honourand \$400 und fract equesting \$4000 empts. Specifically, we are equesting \$4000 for onal interpretation (\$100hour). The in-kind contributions cover the cost of the community logicids coordinator (accommodation, travel, space renal, etc.). The community logistics coordinator would be paid for \$550hour, and we are expecting than to coordinator would be paid for \$550hour, and we are expecting that to coordinator would be paid for \$550hour, and we are expecting that to coordinator would be paid for \$550hour, and we are expecting that a coordinator scale. The examinity logistics coordinator (accounted) to be a set of the source of the s
TRAVEL & ACCOMMODATIONS	\$13,423		\$11,025	\$9,423		\$9,450	\$22,846	522,846	This money will ever the cost of flights to/from Each visit to costs approx. 54000 roundring. The research coordinator plants to travel vise during FY1 and be apprecised plants to travel once. The research coordinator plants to travel vise during FY1 and be apprecised plants to travel avoid uning FY2. The remaining 8 is to cover costs such as travel travel in the angle visit in a during a layover, food on travel dys. The accounted from a reprovided to the regularly changed is \$223,014 [bit main and the angle visit in a during a layover, food on travel dys. The accounted from a are provided to the regularly changed is \$223,014 [bit. During FY1, the research coordinator plants to travel distribution for the research coordinator plants to be in for 5 weeks total.
MATERIALS & SUPPLIES	\$1,500			\$3,500			\$5,000	\$5,000	This money is allotted for consumable materials during the research and/or program phases. This includes items such as flipcharts, posters, arts materials such as fabric.
EQUIPMENT				\$1,000			\$1,000	\$1,000	51,000 This money is alloted for non-consumable materials such as cameras, camping and/or hunting gear, etc.
RENT & UTILITIES			\$1,940			\$7,760	\$9,700	S 0	80 Space rental for the research phase and program implementation will be provided in-kind by the community.
PERFORMANCE MEASUREMENT			\$10,304			\$41,216	\$51,520	S 0	50 This money covers the cost of the research coordinator's salary. It is provided in-kind.
OTHER COSTS (please specify):									
1) Food	\$788			\$2,956			\$3,744	\$3,744	This money is to cover the cost of food provided during the knowledge collection and sharing events. We with also provide food during the youth program. Food provided will am to be mutritous and culturally- upprograms. This may include funditional foods such as berrise or scal meat, or southan foods such as other futus or vegetables. More money is allotted to FY1 as the activities during this year will be with smaller groups of adults. The activities in FY2 will be with larger groups of youth so, we are expecting to need more food.
2) Administration (20% of total = admin costs (529,914.8)	\$8,725		\$2,908	\$13,712		\$4,571	\$29,915	\$22,437	The salary to cover the increased workload for to administer the grant is requested. 15% of the total grant is requested to cover the administrum fee. The grant is requested to cover the administrum fee. The grant is requested from the feet of the grant to add to gap for the administration. If S ⁴ is requested from the fleth Canada, S ⁴ we capter to need 20% of the grant to add to gap for the administration. If S ⁴ is requested from the fleth Canada, S ⁴ we there to need 20% of the grant to add to gap for the administration. If S ⁴ is requested from the fleth Canada, S ⁴ we there to need 20% of the grant to add to gap for the administration. If S ⁴ is requested from the fleth Canada, S ⁴ we there to need 20% of the grant to add to gap for the administration. The state administration for the fleth Canada, S ⁴ we there the need 20% of the grant to add to gap for the administration of the state administration for the fleth Canada, S ⁴ we there are administration for the administration for the fleth Canada, S ⁴ we there administration to the state administration for the administration of the state administration for the state administration fleth administration for the administration for the state administra
3) Counselling	\$4,800			\$4,080			\$8,880	\$8,880	We need to provide counselling services for youth and adults participating in the program. We expect to need about 20 hours a week of commeding during the research phases. In FY1, we will need a research phase in FY1, we will need a counsellor with the youth during the on-the-land earnp (9 days, 8 hours/day, 540/hour). We will also need 10 days of counselling during the other activities (30 hours, 540/hour).
4)							\$0	so	
TOTAL	\$50,857	50	\$31,577	\$121,154	\$0	\$68,397	\$249,589	\$172,011	

COMMENTS:

May 27, 2022

To Whom It May Concern,

Thank you for the funding of our project, Using community-based research to prevent harmful substance use in youth in the

We are requesting a reallocation of funding for our SUAP grant due to exceptional circumstances. Firstly, we were expecting to receive our application's status in December which would have allowed us to begin the project in February. However, we found out in March that we were successful. In the more than 50% of the community goes on-the-land between May – August. Secondly, there was a Covid-19 outbreak that shut down all activity in the community between February – March. As we learned of the project's status in March and were in the middle of a Covid-19 outbreak, we did not have enough time to plan any activities before people left for the summer. Due to this migration, we are not able to begin events until September which drastically changes the allocation of funds.

After more discussions with Council, we learned that research phase 1 (how to do research) must occur prior, and separately, from phase 2 (exploration of substance use among youth and programs in which youth would be interested) to ensure all research is safe and does not cause further harms. This adds complexity to our research licensing process and may lead to further delays caused by the second phase's application's review. This is another complicating factor causing us to ask for funds to be pushed to fiscal year 2.

Thank you for your understanding,

Rachel Cassie

Appendix B – Ethics Approval and Research Licensing Documents

i. Nunavut Research Institute Research License Application

SCIENTIFIC RESEARCH LICENCE APPLICATION HEALTH RELATED RESEARCH						
Before	e completing this application form, please carefully revi	ew the following documer	nts:			
•	(available at https://www.nri.nu.ca/licensing-resource	<u>es</u>); and esearch Involving Human	s – TCPS 2 (2018)			
IMPORTANT						
Please be advised your application cannot be processed until you submit all necessary supporting documents, including:						
÷	confirmation of approval from your institutional resear plain language summary and participant consent for full research protocol;					
These attachments can be uploaded with your online application, or submitted by e-mail, in the following formats: MS Word, Adobe PDF or jpeg.						
	SECTION 1: APPLICANT	NFORMATION				
	SECTION 1: APPLICANT		ng youth in the N/A			
Using	oject Title community-led research to explore how to resear Applicant's full name and mailing address:	ch sensitive topics amo				
Using 2.	Applicant's full name and mailing address: Rachel Cassie Project Supervisor's name and mailing address: Susan Chatwood Edmonton Clinic Health Academy	ch sensitive topics amo				
Using 2.	oject Title g community-led research to explore how to resear Applicant's full name and mailing address: Rachel Cassie Project Supervisor's name and mailing address: Susan Chatwood	ch sensitive topics amo Phone: Fax: Email: Phone:				
Using 2.	Applicant's full name and mailing address: Rachel Cassie Project Supervisor's name and mailing address: Susan Chatwood Edmonton Clinic Health Academy 3-341, 11405 87 Ave NW Edmonton, AB	ch sensitive topics amo Phone: Fax: Email: Phone: Fax: Fax:				

		der and former resident of	
principle		ring project is grounded in IQ	
		SECTION 2: AUTHORIZA	
Institutional E	Desearch E	thics Board Approval	
		eviewed and approved by an Insti	tutional REB2
Yes x	eet been i	No	
			attach documentation to explain why REB
approval has r	ot been ob	otained.	
		thorizations or permits required for	
Authorizatio			Status:
University of	Alberta Re	search Ethics Board	Approved
-		SECTION 3: PROJECT	
		SECTION 3. PROJECT	
	_		
Planned date	s for resea	urch activities in the current calend	dar year:
	s for resea	urch activities in the current calend	dar year:
Planned date: Start date: 01 02	s for resea	arch activities in the current calend	dar year:
Start date:	2022	arch activities in the current calend	lar year:
Start date: 01 02 Day/ Monti	2022	arch activities in the current calend	dar year:
Start date: 01 02 Day/ Monti End Date:	2022 n/ Year	arch activities in the current calend	dar year:
Start date: 01 02 Day/ Montl End Date: 31 12	2022 h/ Year 2022	arch activities in the current calend	dar year:
Start date: 01 02 Day/ Monti End Date:	2022 h/ Year 2022	arch activities in the current calend	dar year:
Start date: 01 02 Day/ Month End Date: 31 12 Day/ Month	2022 n/ Year 2022 n/ Year		dar year: ur research field dates in the current
Start date: 01 02 Day/ Month End Date: 31 12 Day/ Month	2022 n/ Year 2022 n/ Year		
Start date: 01 02 Day/ Month End Date: 31 12 Day/ Month *Please advis year.	2022 h/ Year 2022 h/ Year e NRI imm	ediately if you need to change yo	
Start date: 01 02 Day/ Month End Date: 31 12 Day/ Month *Please advis year.	2022 h/ Year 2022 h/ Year e NRI imm		
Start date: 01 02 Day/ Month End Date: 31 12 Day/ Month *Please advis year.	2022 h/ Year 2022 h/ Year e NRI imm	ediately if you need to change yo	
Start date: 01 02 Day/ Month End Date: 31 12 Day/ Month *Please advis year. Is this applica Yes x	2022 n/ Year 2022 n/ Year e NRI imm ation for a] nediately if you need to change yo <u>new</u> multiyear research project? No	ur research field dates in the current
Start date: 01 02 Day/ Montl End Date: 31 12 Day/ Montl *Please advis year. Is this applica Yes x If Yes, please	2022 h/ Year 2022 h/ Year e NRI imm ation for a provide th	ediately if you need to change yo <u>new</u> multiyear research project? No ne anticipated completion date (m	ur research field dates in the current
Start date: 01 02 Day/ Montl End Date: 31 12 Day/ Montl *Please advis year. Is this applica Yes x If Yes, please Multiyear Pro	2022 h/ Year 2022 h/ Year e NRI imm ation for a provide th ject Comp	ediately if you need to change yo <u>new</u> multiyear research project? No ne anticipated completion date (m	ur research field dates in the current
Start date: 01 02 Day/ Montl End Date: 31 12 Day/ Montl *Please advis year. Is this applica Yes x If Yes, please	2022 h/ Year 2022 h/ Year e NRI imm ation for a provide th ject Comp	ediately if you need to change yo <u>new</u> multiyear research project? No ne anticipated completion date (m	ur research field dates in the current

e this annlication to renew an ex	isting multiyear research license?
Yes	
	a multiyear license if there are changes to your research locations or to your study
	renewed for two consecutive years without a full new application.
Locations of Research in Nunavu	ıt
Community name:	
Тт.	
SECTION 4. NON-TE	CHNICAL PROJECT PROPOSAL DESCRIPTION
	ription of the project proposal, no more than 500 words, in English and . The project description should outline the following:
10 00a 15 2-11144	. The project description should outline the following.
Project Title;	1. 6 (2)
 Lead Researcher's Name a What research questions do 	nd Affiliation; bes the project hope to answer?;
	ctives and why is the study needed?;
	long will the field research be undertaken?;
 What methods will be used What impacts will the research 	
	rch produce to the environment, wildlife, or people?; d by the research be stored and managed?;
	be involved in the research?; and
How, when, and to whom w	ill the research results be shared in Nunavut?
 How, when, and to whom w Upload File 	ill the research results be shared in Nunavut?
Upload File	The relation of the relation of the relation of the second s
Upload File SECTION 5: DETAIL	ED DESCRIPTION OF RESEARCH
Upload File	ED DESCRIPTION OF RESEARCH
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program	ED DESCRIPTION OF RESEARCH detailing the following information:
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Rationale, goals & obje	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Please attach a full project protocol Overall Program Will the research be un	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Actionale, goals & obje Will the research be un initiatives in Nunavut? I Explain the role of Nunavut?	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Actionale, goals & obje Will the research be un initiatives in Nunavut? I Explain the role of Nun: support you will require	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any d from DOH staff or facilities to conduct the research
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Actionale, goals & obje Will the research be un initiatives in Nunavut? I Explain the role of Nun- support you will require Identify the primary sou	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Rationale, goals & obje Will the research be un initiatives in Nunavut? I Explain the role of Nuna support you will require Identify the primary sou Methodology	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any d from DOH staff or facilities to conduct the research rrces of funding for the research project
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Rationale, goals & obje Will the research be un initiatives in Nunavut? I Kunavut? I Explain the role of Nuna support you will require Identify the primary sou Methodology Techniques and protoc Justification for the sele	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any d from DOH staff or facilities to conduct the research irces of funding for the research project ols for sample collection and analysis action of the study methodology
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Rationale, goals & obje Will the research be un initiatives in Nunavut? I Explain the role of Nuna support you will require Identify the primary sou Methodology Techniques and protoc Justification for the sele If interviewing the partic	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any d from DOH staff or facilities to conduct the research urces of funding for the research project ols for sample collection and analysis action of the study methodology cipant is required, provide a list of questions to be posed
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Rationale, goals & obje Will the research be un initiatives in Nunavut? I Explain the role of Nun support you will require Identify the primary sou Methodology Techniques and protoc Justification for the sele If interviewing the partic Location of research ar	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any d from DOH staff or facilities to conduct the research irces of funding for the research project ols for sample collection and analysis action of the study methodology
Upload File SECTION 5: DETAIL Please attach a full project protocol Overall Program Rationale, goals & obje Will the research be un initiatives in Nunavut? I Explain the role of Nuna support you will require Identify the primary sou Methodology Techniques and protoc Justification for the sele If interviewing the partic	ED DESCRIPTION OF RESEARCH detailing the following information: ctives for the project dertaken in conjunction with or in support of any current health f so, reference the programs and explain the relationship. avut's Department of Health (DOH) in the research and describe any d from DOH staff or facilities to conduct the research urces of funding for the research project ols for sample collection and analysis action of the study methodology cipant is required, provide a list of questions to be posed





SECTION 6: COMMUNITY INVOLVEMENT & CONSULTATION

1. List the organizations and individuals in Nunavut that you have consulted with regarding this research:

Community	Name	Organization	Date Contacted
			Aug. 1, 2021

2. Describe the role Nunavut residents or local/regional organizations will play in the research project?

We will form a Research Committee composed of community representatives. We plan to hold six meetings to discuss the practices and policies that need to be enforced to ensure research with Pangnirtung is community-led and researchers have taken all necessary precautions to prevent and minimize any harms. Community-led discussions of research methods will ensure findings result in guidelines for effective, practical, and meaningful steps researchers wishing to work with Pangnirtung must take.

Findings from these meetings will be continually discussed with community representatives. After we have created the research guideline document, we will share it with the community representatives to gain feedback. This will ensure the document is created collaboratively with the community and accurately reflects their needs and priorities.

3. Describe any local services and/or logistic support that you will require to undertake the research (e.g. accommodations, outfitting, translation, sample collection, etc.).

The principal investigator plans to travel to the community in August 2022 (pending NRI approval and Covid-19 related restrictions). With the community, the PI has begun plans for their accommodation, travel, translation and interpretation, and Research Committee meetings. These plans have been in co-drafted with community members and have been budgeted for in the Health Canada grant application.

4. What potential risks does the research pose for Nunavut residents and how will risks be mitigated?

<u>Note:</u> one of the key questions for this project is how to prevent and minimize harms caused by research. Thus, we expect minimal risks caused by the research.

Risks:

P.O. Box 1720 Iqaluit, NU, X0A 0H0 • PHONE: 867-979-7279 • FAX: 867-979-7109 • email mosha.cote@arcticcollege.ca



When participating in the meetings, community representatives may experience psychological or emotional stress triggered by discussing practices researchers need to undertake to prevent and minimize harms caused by research. This may cause community representatives to reflect on past trauma(s) caused by research or bring psychological or emotional insecurities to the forefront. Community representatives may also feel psychological or mental fatigue when recalling past events. There is also the risk for Covid-19 transmission as these research activities will take place in-person.

Risk mitigation:

We will mitigate risks and discomforts by ensuring meeting topics are culturally appropriate and align well with IQ principles. Community-led research will also work to mitigate risks and discomforts of research activities as the meeting topics will be co-determined with the community representatives. We will share meeting topics with the Research Committee prior to meetings to allow people to opt-out of any meetings they do not feel comfortable attending. We will also remind participants throughout the meetings of their right to leave meetings and/or withdraw from the research process, and of the mental health resources available to support them.

We will also mitigate risks and discomforts by making clear the university team's role in this community-led research project. This will function to build trust and reciprocal relationships between Research Committee and university-based researchers. It will also reduce risks of psychological or emotional harm.

We will mitigate any harms by providing referring community representatives to resources that support mental health. These resources include in-person or tele-counselling, and referral to various helplines or online support groups such as the Kamatsiaqtut Helpline or Kids Help Phone's Peer to Peer Community.

The PI has experience navigating emotionally-charged and triggering situations as she is also a mental health worker in Vancouver's Downtown Eastside, working with mainly Indigenous peoples.

We will mitigate the risk of Covid-19 transmission by following all public health recommendations and requirements. This may include, but is not limited to, wearing masks indoors, requiring representatives be vaccinated, limiting group sizes to no larger than 10 people, and more.

5. Will your project require assistance from the Department of Health (DOH) (e.g. funds, time, facilities, data access, etc.)? If so please clearly describe the DOH's role in your research project and include confirmation of departmental support for your research.

No, this project will not require assistance from the Department of Health. However, we plan to create a research guideline document and/or policy brief that may be helpful to the Department of Health.

SECTION 7: GENERAL

P.O. Box 1720 Iqaluit, NU, X0A 0H0 • PHONE: 867-979-7279 • FAX: 867-979-7109 • email mosha.cote@arcticcollege.ca

Page 6 of 7

	An and a start was a start of the start of t					
 Do you give NRI permission to publish project information in the Nunavut Research Institute Annual Compendium of Research Undertaken in Nunavut? 						
C XYES						
Applicant:						
Signature	<u>Rachel Cassie, MSc candidate</u> Title	Jan. 15, 2022 Date				
P.O. Box 1720 Iqaluit, NU, X0A 0H0 • PH	ONE: 867-979-7279 • FAX: 867-979-7109 • en	nail mosha.cote@arcticcollege.ca				
	Page 7 of 7					

ii. Non-technical description of project in English and the local Inuktut language

Non-technical Project Proposal Description

The research project's findings will be a guide for how to do community-led research in It will focus on how to safely explore sensitive topics (eg. substance abuse), and how youth can safely participate in such research. The university team is proposing a long-term research relationship, to ensure solutions are sustainable and lead to lasting change.

Research on how to do culturally-appropriate research that meets the community's needs is needed to prevent and minimize research harms. Historically, research has caused harms among community members in the searchers did little to prevent or address these harms. Research is needed to address many of challenges faces today, however, this research must be culturally appropriate and community-led.

This research project will consist of meetings with community representatives, including Elders, community leaders, and community members who are passionate about making positive change in

These meetings' location will be determined by the community representatives (eg. Office). We propose that these meetings could occur in August 2022. During these meetings, community representatives will lead discussions around research goals, topics, and practices. These meetings will guarantee there is total community direction of research and determine everyone's roles and responsibilities to ensure that this research results in positive change.

Specific meeting discussions will focus around how to anticipate and minimize harms associated with research, including ways to interview people to prevent harm and support those who have been negatively impacted. This could include making sure people doing the interviews know to give breaks or stop if the participant appears uncomfortable or making sure that there are mental health supports available. While community mental health resources do not currently meet the community's needs, the university team is proposing to seek funding to support the development of more supports, such as Elder-youth counselling initiatives.

The knowledge created during these meetings will ensure later research on sensitive topics such as addictions or crime is respectful and minimizes harms.

Meeting notes will be anonymized and stored using data management software that encrypts data. Data will be routinely backed up.

We will have ongoing discussions with the community representatives about research findings and community updates throughout the research in both and English. This is to ensure the knowledge learned is shared with and leads to positive change.

ዄኯእነ[®]ጋ፫ኪσ[®] Δϲ[®]d/^cበ^{*}σ^{*} ΔϲϲΡⁱ³λ[#]νν^b⁴^{*}σ[®]CD⁵σ^{*}d²C²</sub> Γ^ρ^{*} ϲ⁻⁶²³σ² σ⁴²σ³²², ^ρ³σ²², ^ρ³σ²², ⁶³σ²², ⁶³²², ⁶³², ⁶³³, ⁶^{3²², ⁶³², ⁶^{3²², ⁶³², ⁶³², ⁶³², ⁶³², ⁶³², ⁶³³, ⁶³², ⁶³³, ⁶³, ⁶³³, ⁶³, ⁶³³, ⁶³, ⁶³³, ⁶³, ⁶³}}

ቴኮኦዚቄናርኮበታነፋ ወዲ^ቈbበሱህተታ ቴኮኦኣንርፑታናታ ኮቴቴበሱ-ታነፋ Δውናበጋና ቴናጋሲበጋናጋ. ርL-ዲ ቴኮኦኦኮና «Γ՞ቴኦርኮኦሊ⊲ቴናኦኈጋና ለኮተጋር አራቴኒስቲ አምታላይ የአምጋና

iii. Detailed description of the project

Section 5: Detailed description of the research

Overall Program

Background: The proposed project will be done in partnership between and the University of Alberta's School of Public Health.

has a population of around people, with about one third of the population under the age of fifteen.

Inuit Qaujimajatuqangit (IQ) principles, or "that which Inuit have always known to be true", are an important facet of Inuit culture and communities. IQ principles are shared intergenerationally and guide every aspect of the community including family life, wellness, education, and justice.

The impacts of colonization have resulted in widespread intergenerational trauma among 's community members. These include poverty, food insecurity, harmful substance use, adverse childhood events, and unfavourable Elder care. Research on how to address these challenges is necessary, but these topics are sensitive and can be traumatizing to discuss.

Historically, research has been conducted *in*, rather than *with*. Additionally, many research topics centre around the impacts of colonization, requiring careful planning to anticipate and mitigate risks to the participant. Often, researchers do not consider these important aspects of research, leaving the community more drained and traumatized than before. This harmful approach to research has resulted in community members having traumatic relationships with research and being hesitant to participate in further research. However, to address many of the challenges is experiencing, more research into their causes and effective interventions is needed.

Rationale:

This research will explore: 1) how to ensure there is total community control and decisionmaking power throughout the research process, and 2) how to conduct research on sensitive topics that may be traumatizing, especially with youth. This research project is important to ensure future research with for the sense of colonization, but exploration of how to do this research is an important first step. While similar research has been conducted in other communities, sepecific research is important to rebuild the community's trust with the research world and to ensure these practices and policies meet 's specific needs.

In 2018, a Community Needs Assessment was conducted in ______. The impacts of intergenerational trauma were identified as a key area of concern. Specifically, concerns for the community's widespread poverty and the prevalence and extent of harmful substance use

among youth were highlighted by many community members. Preliminary discussion with the community, including Elders and community leaders, has confirmed that these concerns have increased since 2018. Thus, to ensure future research appropriately addresses these growing concerns and challenges and does not add to or amplify the challenges the community is already experiencing, research into how to effectively explore these topics is needed.

Objectives and Goals:

This project aims to answer the question: 'How should research on sensitive topics in be conducted to minimize risks and ensure total community control and

ownership?' The project's objectives are to:

- 1. Explore how to ensure there is total community control and decision-making power throughout the research process;
- Explore how to conduct research in on sensitive topics, especially with youth;
- 3. Discuss community-identified priorities for research topics; and
- 4. Generate guidelines for researchers to follow when working with

These findings will be used by the community to guide research being conducted with

. Community-led, culturally-grounded guidelines for research are needed to ensure research is culturally appropriate and leads to positive change in the community while minimizing all potential risks. As well, this research approach can be adapted for other communities wishing to address the harms of research.

Current Health Initiatives:

This project lays the groundwork for subsequent research projects with _______. Specifically, research aiming to address the impacts of colonization has been discussed as a potential follow-up project. Topics of focus for this project are yet to be determined, but preliminary discussion has focussed around preventing crime and substance use, especially among youth. This subsequent research project, as all other research projects that would be done with ______, would follow the guidelines set forth in this proposed project.

Department of Health:

This research will be independent of the Department of Health. We plan to create a research guideline document and other resources such as a recommendations report or policy brief which may be of use or interest to the Department of Health. We do not expect to require any support from DOH staff or facilities to conduct the research.

Funding:

We have applied to Health Canada's Substance Use and Addictions Program grant. Our application is currently under review - we expect to hear of our application's status soon. If approved, will hold this funding. The university team is also offering to seek other funding that would be held by the determined of the community wishes to proceed.

Methodology

Techniques and protocols for data collection:

We will form a Research Committee composed of Elders and other community leaders and members who are passionate about making positive change in **Community** During these meetings, community representatives will lead discussions around research goals, topics, and practices. We propose that these meetings could occur in September 2022. These meetings will guarantee there is total community direction of the research guideline document. As a primary step, we will also determine everyone's roles and responsibilities to ensure that this research results in positive change.

We will utilize qualitative methods to create a guideline document for research with . These guidelines will be developed through discussion during Research Committee meetings to ensure they accurately reflect 's research goals and needs. The university team will be responsible for compiling these guidelines into a document through iterative analysis of meetings' audio recordings and minutes. Throughout analysis, the university team will continually check the guidelines are representative of the Research Committee's goals and intentions and will ensure this document is available to all community members in both English and . The will own this document and all other research materials.

Techniques and protocols for data analyses:

The university team will conduct iterative thematic analysis of the Research Committee meetings. Our findings will describe the guidelines for research with as voiced by the Research Committee Members. We plan to use NVivo software for qualitative data analysis.

Justification for methodology:

This community-led approach aims to ensure the community's needs and goals guide all subsequent research in **Section**. Research to address the challenges the community is experiencing is needed, but it must be done in a way that is sensitive to the physical, emotional, and spiritual needs of **Section**'s community members. This research project aims to ensure all subsequent research will have these needs at the forefront and reduce the demands on the community to continually assert their needs and priorities regarding research methodology and topic. This project also attempts to rebuild trust between the community and research world.

Qualitative methods will allow for a rich description of the community's needs in the research guidelines. This rich description will lay the foundation for subsequent research, aiming to result in research that meets the community's needs, is culturally appropriate, and results in positive change while minimizing any harms.

Community meetings:

We will co-determine meeting topics with the community representatives. Examples of meeting topics will include research priorities for the search priorities for the search process, and research methods that align well with the community's needs. Specific attention will focus on ways to anticipate and mitigate any risks associated with

the research process. For example, this could include discussing ways to prevent harms when interviewing such as training interviewers to offer breaks or stop if the participant appears uncomfortable. It will also be important to discuss supports that need to be available to participants such as counselling. All meetings will have an interpreter present to ensure everyone is able to participate fully and comfortably. Rachel Cassie will take meeting minutes and member check these minutes throughout the meetings. These meetings will also be audio recorded to ensure accurate data collection. Community representatives will be paid \$40/hour for their time when participating in these meetings.

Location of research + rationale:

This research project has been co-developed by: 1) Rachel Cassie, University of Alberta Master's student 2) 3) Susan Chatwood, University of Alberta Associate Professor, and 4) 3) Susan Chatwood, University of Alberta Associate Professor, and 4) 4, Inuk Elder and former resident of 4 to develop mental health resources between May to December 2021. In this role, she asked if the community would be interested in working with her for her thesis and received significant interest.

's research needs lay the foundation for all research goals and community meetings. Further, will own all research materials and control all funding.

Consent:

We will offer oral and written consent options to all community representatives. After discussion with community members, we are expecting oral consent to be preferred by many representatives as it aligns better with Inuit culture. Both oral and written consent will be available in English and **sector**. An interpreter trained in the principles of informed and free consent will be present for any community representative wishing to provide oral consent in

During oral consent, the researcher guiding the consent process will discuss all points covered on the written consent forms and answer any questions the community representative may have. The researcher will record the name and date of all community representatives who gave consent orally. The written consent forms are available in and English and a translator will be present when signing to facilitate any questions. Research guidelines will also discuss ways to obtain consent that align well with Inuit culture.

Primary Data and Information

Short Term Use

In the short term, research findings will be used to create guidelines that describe how to: 1) ensure there is total community control and decision-making power throughout the research process, and 2) conduct research in on sensitive topics, especially with youth.

Long Term Use

In the long term, the developed research guidelines will ensure research with meets the community's research needs and priorities and minimizes harms caused by research. This project aims to ensure research exploring ways to address sector 's challenges can occur in a way that is emotionally safe and facilitates positive change within the community.

Accessibility of data

Critical to respectful community-led research, we will utilize integrated knowledge translation. The university team will engage in ongoing discussion with the Research Committee to ensure meeting topics and findings are meeting their needs. When compiling the research guidelines document, the university team will ensure there are regular community updates in both

and English. We will share and invite feedback on preliminary research findings and the guidelines document. This document will be available in both and English, and utilize plain language and infographics to increase accessibility and relevance of this document.

The University of Alberta's research team will have access to the data (ie. meeting minutes and transcriptions of audio recordings); this includes, Rachel Cassie, Dr. Susan Chatwood, and Dr. Sherilee Harper. Dr. Susan Chatwood and Dr. Sherilee Harper may be consulted on data analysis techniques throughout the research project, thus access to this data will be on an as-needed basis.

The University of Alberta's Research Ethics Board will also have access to the data, if needed.

Short term data storage

All data will be entered into Word or Excel on the principal investigator's computer for analysis by statistical software such as NVivo. It will be saved only on a password-protected USB memory stick that will also be encrypted.

All physical research materials (eg. if artwork is created to visually describe an idea) will be stored in a locked filing cabinet at the stored 's office. Names of the creators of any research materials will not be attached to the documents, unless consent is given to do so.

Long term data storage

This USB stick will be given tofor administrative storage at the endof the research project. This USB stick will be stored in a locked cabinet at the's office.Asowns this research, they can decide if they wish to destroy thedata after 5 years.

Disposal of data

We will destroy all identifying data (eg. photographs of people who do not consent, names on materials created by community representatives, etc.) once we have confirmed with the person

that they do not consent to the material(s) being used, held, or identifiable. If community representatives wish to withdraw from the Research Committee and wish to have their data revoked, we will retroactively remove their comments and suggestions from the previous meetings' minutes. We will destroy this information by shredding any paper documents and permanently deleting documents off the password-protected USB stick. We will consult with IT professionals to ensure the documents have been permanently deleted.

As owns the research, they can choose to destroy the data after 5 years. If they wish to do so, we will share information regarding how to permanently delete or 'wipe' a USB stick.

Other uses of data

There are no future uses for the data. The password-protected USB stick will be given toto be stored for 5 years. Asowns this research, they canchoose to destroy it after 5 years.

Intellectual property rights/ownership of data

will own all research data. They will also hold all funding for research activities. The principal investigator, Rachel Cassie, will be basing her master's thesis on this project. All knowledge sharing materials will be co-created with the Research Committee and require their approval prior to sharing.

Biological Samples

No biological materials will be taken.

Medical Chart Review

No medical chart review will occur.

iv. Consent form

Using community-led research to explore how to research sensitive topics among youth in

Participant Consent Form

Project Title:

Using community-led research to explore how to research sensitive topics among youth in

Principal Investigator

Rachel Cassie University of Alberta Edmonton Clinic Health Academy 3-341, 11405 87 Ave NW Edmonton, AB T6G 1C9

Research Supervisor

Susan Chatwood Edmonton Clinic Health Academy 3-341, 11405 87 Ave NW Edmonton, AB T6G 1C9

Background:

We invite you to participate in a research study being conducted by and the University of Alberta. Taking part in the research is entirely your choice. Even if you do take part, you may leave the study at any time for any reason. Withdrawal or refusing to participate will not affect your relationship with

or the University of Alberta. The information below describes the research, tells you about what is involved in the research, what you will be asked to do and about any benefit, risk, inconvenience or discomfort that you might experience. Please ask as many questions as you like. If you have any questions later, please contact the research coordinator, Rachel Cassie (

's Council gave direction for this project. This project explores how to do research with . Research on how to do culturally-appropriate research that meets the community's needs is needed to prevent and minimize research harms. Historically, research has caused harms among community members in

, and researchers did little to prevent or address these harms. Research is needed to address many of challenges faces today, however, this research must be culturally appropriate and community-led.

We are recruiting adults to participate in this study as community representatives. Community representatives are asked to participate in a minimum of three (3) meetings between September __ to __ at _____ Office. During these meetings we will discuss how research should be done in ______ to meet the community's needs and priorities, and how to prevent and minimize harms caused by research. We will identify potential community representatives through social media posts, posters in the community, and radio segments sharing the research opportunity.

U of A Ethics Number:

Purpose

This project aims to explore the research questions:

- 1) How can we ensure that community members direct all research activities and research results are owned by ?
- 2) How should community-led research on sensitive topics with youth be done to prevent and minimize any harms resulting from the research?

Specific questions we hope to answer include:

- 1) What practices should researchers use to ensure that all research activities are directed by the community?
- 2) Are there any policies that need to be implemented by to ensure all research is community-led?
- 3) What safeguards (ie. practices or policies) need to be enforced to ensure that research on sensitive topics prevents and minimizes any harms caused by research?
- 4) Are there any specific safeguards researchers need to consider when doing research with youth?
- 5) What are the community's current research priorities and interests?

Data will be used to create a research guideline document to make sure any future research with sensitive to the community's needs and priorities. Data will also be used by Rachel Cassie for her graduate thesis.

What you will be asked to do - study procedures:

- · We will form a Research Committee composed of community representatives
- We will be holding 6 meetings between September ____ and ____ for community representatives
- Each meeting will focus on a different question to guide the creation of the research guideline document
 - Example: What practices researchers use to ensure that all research activities are directed by the community?
 - See the above section "specific questions we hope to answer" for other potential meeting topics
- Community representatives will be asked to share their knowledge and time during Research Committee meetings
- These meetings will be audio recorded to help the researcher during data analysis. The researcher will also take minutes during the meeting.
- We will draft terms of reference for the Research Committee. This will create mutual understanding of roles and responsibility.
- We are expecting to hold 6 meetings between September and that are 1-2 hours in duration. Community representatives are asked to attend at least 3 of these meetings.
- Every meeting will begin with a brief review of previous meetings' discussions to allow representatives who were not present to be able to contribute to this discussion. Meeting format will be flexible to allow for continued discussion of topics when necessary.

U of A Ethics Number:

Type of information collected:

- Age and gender
- Your knowledge and thoughts on how to do research on sensitive topics with youth in

Reviewing the projects' findings:

 When the research results have been synthesized, you will be invited to a focus group session with other participants to review the main findings and ensure they are accurate.

Statement of informant rights:

"I have been fully informed of the objectives of the project being conducted. I understand these objectives and consent to being interviewed for the project. I understand that steps will be undertaken to ensure that this interview will remain confidential unless I consent to being identified. I also understand that, if I wish to withdraw from the study, I may do so without any repercussions."

Benefits of the project

- You will guide the creation of a research guideline document to ensure future
 research with is community-led and safe
- You will have the opportunity to share your perspective on what the research process and priorities should be for
- · You will have the opportunity to learn about the research process

Risks

- · There are no major risks, discomforts or inconveniences expected.
- Some people may be concerned about the time required to take part in the research process or activities. This is why community representatives are not required to attend all meetings and are asked to attend at least 3 meetings.
- The investigators will minimize these risks by ensuring that your participation in this research project remains completely voluntary and you may choose to remain anonymous.
- If you decide to leave before or during the project, you are free to do so.

Payment

- All community representatives will be paid for sharing their time and knowledge.
- · Community representatives will receive an honorarium of \$40/hour.
 - Most meetings are expected to be 90 minutes in duration. Members would receive \$60 for attending a 90 minute meeting.

U of A Ethics Number:

Confidentiality

- · Confidentiality is important in all research settings.
- No identifiable information will be recorded in Research Committee minutes
- All personal information shared at Research Committee meetings will not be shared beyond the group (or at later meetings) and will only be used if the representative agrees
- The University of Alberta Research Ethics Board (REB) has the right to review all study data.
- All research data (ie. meeting minutes and transcriptions) will be stored on a
 password-protected USB stick that will be kept in a locked filing cabinet.
 - All research data will also be encrypted to further ensure participants' confidentiality.
 - This USB stick will be given to filing cabinet.
 - All identifiers will be removed from the data, unless consent was given for it to be retained.
 - After 5 years, can decide if they wish to destroy the data.

Conditions for release of recorded information:

There are no plans to release recorded information (eg. recorded interview audio). If this changes, we will ask participants for consent <u>before</u> releasing any recorded information.

Voluntary Participation

Your participation in the Research Committee is entirely voluntary and will not affect your relationship with or the University of Alberta. How you participate will have no legal or administrative consequences.

Freedom to Withdraw

- · You are free to leave the Research Committee at any time.
- Leaving the Research Committee will not affect your relationship with
 or the University of Alberta.
- If you decide to stop participating at any point during this study, your contributions will be considered to guide subsequent research activities.

How to obtain results:

We will provide ongoing discussion of project findings. We will also have a community gathering at the end of the project where we will present the research guideline document and invite feedback to ensure the document accurately reflects the community's needs and priorities. This document will be owned by and available to all community members in English and the shared in a similar way and owned by accurately will be shared in a similar way and owned by a share a community is needs.

U of A Ethics Number:

Research Results may be used/disseminated in the following ways:

- Planned uses: Guidelines and Graduate Thesis
- · Other possible uses: Reports, Presentation, and Publications

Additional Contacts

- If you have any ideas, questions or concerns about this research project, please contact the principal investigator, Rachel Cassie, at or
- · You may also contact her research supervisor, Dr. Susan Chatwood at
- You may also contact
 Councilor
 at
- The plan for this study has been reviewed by a Research Ethics Board at the University of Alberta. If you have questions about your rights or how research should be conducted, you can call (780) 492-2615. This office is independent of the researchers.
- You can also contact the Nunavut Research Institute at 867-979-7280 with any concerns or questions about the research.

U of A Ethics Number:

Consent:

If you sign this form, you do not give up your legal rights nor release the investigators or resolved institutions from their legal and professional responsibilities.

Your signature:

- I have read what this study is about and understood the risks and benefits. I
 have had adequate time to think about this and had the opportunity to ask
 questions and my questions have been answered.
- I agree to participate in the research project, understanding the risks and contributions of my participation, that my participation is voluntary, and that I may end my participation at any time.

For the purpose of this research there are options for you to be Audio/Video-tapped **OR** Participate and remain anonymous

Please check one:

 I agree to be audio-recorded during the meetings 	Yes 🛛 No 🗆
 I agree to be <u>video-recorded</u> during the meetings or photographed during project activities 	Yes 🛛 No 🗆
 I agree to the use of quotations 	Yes 🛛 No 🗆
 I allow my name to be identified in any publications resulting from this study 	Yes 🛛 No 🗆
For Future Research Projects (please check one):	
 I agree that my data may be used for future projects related to community-led research 	Yes 🛛 No 🗆

A copy of this Informed Consent Form (with signatures) has been given to me for my records.

I hereby consent to participate in this study:

Name of Participant (print)

Signature

Date

U of A Ethics Number:



Participant Contact Information (please fill out):

Name :_____

Telephone : _____ E-Mail : _____

Mailing address:

Co-Lead/Research Coordinator's Signature:

I have explained this study to the best of my ability. I invited questions and gave answers. I believe that the participant fully understands what is involved in being in the study, any potential risks of the study and that he or she has freely chosen to be in the study.

Name of Research Coordinator or Project Co-Lead (print)

Signature

Date

If you have any questions, please do not hesitate to contact the Research Coordinator:

Rachel Cassie, Principal Investigator c/o University of Alberta Edmonton Clinic Health Academy 3-341, 11405 87 Ave NW Edmonton, AB T6G 1C9

U of A Ethics Number: 7 of 7

ፅ⊳ት\ናምሞ ወዲሮ*ም የትግድላዊው ንር ማግግራ መድግግ የትግድ ብም የትግድላዊ ዘላዛሬ የሆም

ABCD to ATPA CCALAC'

<u> እר⊾⊲∿Jל< ⊲∩ኈ'ም∿し:</u>

ᡃ᠋ᢐ᠌᠋ᡋᠵ᠋᠋᠋ᠰ᠋᠋ᠮ᠆ᢧ᠋ᢆᡆᡄ᠆ᡷᠣ᠊᠊ᢐ᠋ᡋᡔᢣ᠋᠋ᡅᡄᠡᠯ᠋᠋ᡘ᠋᠋ᠬᢒ᠖᠋ᢆ᠆᠘ᡱᠠᡃ᠋ᡗ᠋ᢅ᠋ᠴᡄ᠆ᢐ᠖᠋ᢂ᠋ᠺ᠆᠅ᠤᡏ ᡰᡰᡐ᠋ᡃ᠘ᡄ᠋᠋᠋᠋ᡶᠴ

ק∆<י≻ הֹץ University of Alberta Edmonton Clinic Health Academy 3-341, 11405 87 Ave NW Edmonton, AB T6G 1C9

ظ^ر نرده د

Edmonton Clinic Health Academy 3-341, 11405 87 Ave NW Edmonton, AB T6G 1C9

<u>ለՐ⊲₽∩Ր/Lን∿Ն፡:</u>

 ۵
 ۵
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲
 ۲</td

H4L2. 1 b0L2 1 d1>Da(2D 2) CLDT 1 U Arad 1 A 1 A 2 A $^$

U of A Ethics Number:

Δ°¬ε'σ' Δ°b¬Δγ°∩c∩σ<

«ጋJc Λ°bCP'd'_)C CLጋΓ°L %D2\'σDσ

«ጋΓ' ∧∿しł⊿ˤ ⊲໋Րႆł⊳֊ച ዘ⊲୳Lஉነሪና . ሀሀር ሬንግር ርነዋው እይለት ይረጉ እ የ'ሁ፞፝®ጋ∆ት⋗σ⊲፝®ጋም ∆ຼຼະም ⊲⊇ና⊇⊂ ኄና∖₽ሃነኇ ጋ∖ናኁ∖በሇ, ຼຼຼຼຼຼຼຼຼ ລຼຼຼຼ ~ ๙๙๖๙๗ ⊇∖ናኁ\ኄና⊇⊂, $\Delta C = \Delta C$

<u>ገግሆኑ የ</u>

ለল∿ፈብብት ርΓኖሚ ያ⊳ታፖኒዲቪ ርዋዋገዮል ወላምዋር።

- 1) ዄዾ^ቈ bጚ/በናበናበላናታላ^ቈለር ርL[₺]dd ዄዾትኣናታና ዄዾትኣዖርዾታሲ^ቈጋ-ዾ ᡄᡃᡏᠣᡅᡃᢂ᠋᠅ᠵ᠋᠆᠋ᠴ᠋ᡃᠵ
- ᠋᠋ᢉ᠋᠋᠙᠆᠆᠅᠆ᢕᢂ᠆᠋᠃᠆᠅᠆ᠺ᠘᠋᠕᠆᠆

⊲∧%d⊂><< ?>>

- DCCD 150 diLC?
- 2) ⊲⊃⊲し⊂⊳ռ⊲७°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 4°°
 <li
- 10-500%-20°?
- 3) $\Delta^{c} = \Lambda^{c} =$ <u>᠆</u>᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆᠆
- 4) ጜ⊳ኦኣኈሰና ጜዾፚናጋቍ ፚ/ደቦ⊲ኪ⊲ጜኈ<ና ዾዾ/ናጋቍና ጜ⊳ኦኣጜበጜኆኈ<<?</p>
- 5) ዾዹ፞፟፟፟ ዾዾ፟፟፟፟ ዾዾ 5) ዾዺ፟፟ 5) ዾ 5) 5) ዾ 5) ዾ 5) 5) ዾ 5) ዾ 5) ዾ 5) ዾ 5) 5) ዾ 5) ዾ 5) 5) ዾ 5

ቴ⊳ኦኣኄቫናየLና ፈ^ናዮነረላንረርሥ~~ህቃና Lーሁነአーሥነጋበት ለቃውቢውፈምርበው ቴ⊳<u>ት</u>ኣΓ፞፟፝፝፦ና ናΔ^cł ႱłϷ< Δ<u>ϲ</u>·σ<u>⊲</u>[®]C[®]₂ レ₂^c ለታሲ<mark></mark>2∩Րσ⊲ናΓ^LL^c.

- ¬σ-%PUL,J4c PUL≯D4c
- ኄ⊳ትኣኄ₽ዾ ፐ⊂ቦነላ∿Ր°₽
 - ▷<
 ○
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □
 □</li ᠣᡆ᠋ᢞᡃᠥ᠋ᢕᡤ᠈ᡃᢧᡕᠴᡄ
 - Cd⊃J⊃ 'd¿ör CD * ΔLΔ
 Cd⊃J⊃ 'd¿ör CD * ΔLΔ
 Cd⊃J⊃ 'd¿ör CD * ΔLΔ Ֆ⊳ֈֈշՍրրշ
- <<u>י_</u>ח׳ פטררפע<<C

U of A Ethics Number: 2 of 8



- <u>b∩L%CPLσ
 <u>b∩L%CPLσ
 <u>dCD/10
 6-Ġ%∩1_C
 <u>dCD/10
 4br
 <u>dCD/10
 <u>d</u>
 <u>d</u></li</u></u></u></u></u></u></u></u></u></u></u></u></u></u></u>
- bnlc-sa<<< °Pr>?*b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 b
 c
 b
 c
 b
 c
 b
 c
 b
 c
 b
 c
 b
 c
 b
 c
 c
 b
 c
 c
 b
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 c
 <p

<u>∽¬¬~¬`````</u>14'₽ <u></u>

- ኻ▷ኦLơ∩ና Δ/LՐኑՈ·_> ኻዾ[፨] ኻ▷ኦ\ናምናኮ ኻ▷ኦLኦ▷ኦኪ⊲<">> ∿▷◊▷/ኣ\ም՝ Δዾ/ናጋና ለ

<u> 'የГ'?~` 'b>+\?∩Г~>+~:</u>

"DYPLACPYN4"cn'-3"U Nard "CPYL4" N'-5" $d^{9}P'^{8}CPrL\sigma' d''$. DPY9"U Nard "CPYL4" d"YYL5"5-3 dN"7"6CP5"5" CLDJU NondUtJ. DPY9"U b"Ua"D"401">P549"UC dN"7"6CP5" a.3a Δ "?"YD4PL drd5". DPY9"U, 8"UD4"a" Acr>P4"572 8P2Y55J5, CALAJ*a5L ArL"?"

<u>∧⊳∽∿Ր°</u>

- Δb
 Δb
 CP%
 Δb
 CP%
 CP%
- Δ
 Δ
 σ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ
 Δ

<u>ᠳᠵ᠘ᢣᠵ᠘ᡩ᠘</u>

- Δ<[°]Υ^c ΔΔΔ^c Δ/L

 ΔΔ^c Δ/L

 ΔΔ^c

U of A Ethics Number:

- <u>ረ</u>⊂ βαραιγΔ^c %ργΓία[∞]Γ⁶αβρ⁶ CΔΓγααζΓC.

<u>⊲₽⊂⁰C⊅ჾ⁰</u>

- هد- خنی ۵۵ ماله ۵۰ ماله ۵۵ ماله ۵۰ ماله ماله ۵۰ ماله ۵۰ ماله ۵۰ ماله ۵۰ ماله ۵۰ ماله ۵۰ ماله ماله ۵۰ ماله ۵۰ ماله ۵۰ ماله ۵۰ ماله ماله ۵۰ مال ۲۰ ماله ۵۰ ماله ۵
- Δ655 ΦCD25 ΦCD20 \$40-Γ
 - ር∆L∆^LL^c b∩Lⁱ לל⊲^c ב∩^k \$60-Γⁱ C∆L^c ב∩Ր b∩Lσ^c Γ^c ב^c

ႱჼႮႭჼႻჼႱჿ

- ᡏᠫᡥᢗᢦ᠋ᠫ᠕᠂ᠳ᠘᠕᠕᠘
- Δ- σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ
 σ<
- - %>>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 %>
 <li ₽₺₫σ℃~σ ₺₽₽₺₽₺₫ፈՐና.
 - C՟~L໑° Ͻჼჾჾ Ϸ٬L⊀Γ.
 - ᢐ▷›ኣናჾ▷< ℾካኳፍ Δረቦን▷ ‹‹፲୮ጵፍ ለን የር▷/ Lሁን ጋና, የ/ ⊲σ ~ ር∆ ነ ፈው ህ
 - ▷የ▷ና Ϲ·ϲͰΔና ⊲σJኈ∩·_ጋዮና, ዘ⊲୳Lຼ୰ኇ ኈ▷ዖነንዖበΓσ▷ላσ՝ ለና∆ላL°σኈϧϹ CALAJea Gy & DC.

<u>ב- כסאת אל- איףיכאיאאאיאר האראיטלרי:</u>

 $< \mathbf{C} \in \mathcal{C}$ <u>/୭୦ଏ/</u> ୦۸୦ଏଧ/L៩ ነ፨ዖር▷[∞]Ր°୦°.

<u>م ۲۵۵٬۹۵٬۷۵۵ م. ۲۹۵</u>

ለቴርኮላLσቢታΔ° ር፦ዕዉσ ቴኮትኣናσናዮ ለላLσናጋ° የረላσ Lc<> « «ጋΔነትዮርናጋ ᠆ᡔᡟᢣᡃ᠋ᡃᠺ᠊᠉ᡣᡄ᠋ᠴ᠆ᡱ᠆ᡄ᠋᠘᠋ᢆ᠆᠆ᡄ᠋᠆ᡧᡄ᠋ᡃ᠕ᡃᡆᢕᢧ᠋᠆ᡱ᠂ᡠᡃ

<u>ምንትየምሆኑ VeCp4Γ4-ም58</u>c <u>Phra</u>

%%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%
 %%

U of A Ethics Number: 4 of 8

- ለ የ C D ላ L ላ የ 6 ና 6 ና C b d 2 * V ና የ D P \\ * 0 G C d 2 * Y ና G C A 2 * Y ና G C A 2 * Y ና G C A 2 * C

<u> የቀትለንሀቢ ቀቅ የዓትም ይባሩ የቆይ የ</u>

<u>ʹϧϷϟϯͻͶϹϷϲͺϥϽϣϹϷϥͼϭϟͺϷͺϟͺϲͺ</u>

₽₽⋖っС₀ъ₊€⊅₅₽4⊲&ъ∪с⊳9⊲

- Δ/L<>P
 D
 ללב
 ח
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
 ה
- که نماته مرسمه می از می مرسمه می از می مرسمه می مرسم مرسمه می م مرسمه می مرسم مرسمه می مرسمه م مرسمه می مرممه می مرسمه می مرممه می مرمومه می مرسمه می مرسمه م
- •
- ኻ▷ኦኮ⊲ጘልቦ๙-ዹናፑኦቦና ዾዹቃፑ ኻ▷ኦኣና ጋ፦ኪኇና፲ና bበLኦኁና ▷ቫሩ Dበኄር (867) 979-7280 ኻዾጋΔ-ዹናበ⊲ጘ Δ/أ_ጏበኻጘልና ⊲ለጘልበኁዾጏኁኇና ኻ▷ኦኣናውኦና Γነ፟፞ጏዾ፞ጚዮ.

U of A Ethics Number:

<u>⊲∿rp∩∘:</u>

 $C^*a < A \cap C > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a > A < a$

⊲∪⊂⊳,יַכ:

⊲⊂⊳୵'r`∩∩"≀r⊂:

- ४५८०°८०४८२४ అందంలి అందంలి అందంలి అందంలి
 ४५७०९८०४८४८४८ ఉందంలి
 ४५८९४८४८४८४८ ఉందంలి

ィ≫ჾჀჾ⊲ჼႠႶჼჾ ለႠჀզჼJჾզჼႦჂႽ (⊲⊂⊳Ⴤჼℾჼ ჾჇ⊲ჀႽ):

U of A Ethics Number:

م⊃م⊽₀به	▶∿L ⊲∿Րነ/	_ح`ح• ∆د	6'40% 4th	~ ~~ し Ċ`ł」~し	ቴ⊳ኦኣ'σ'J°:

⊲∩۶ (ח∩۶⊃۵۰۵`م))	40 <i>~</i> >?/5	⊳₋۲	

رد ۲۰٫۹۵۵ (۲۳۰) ۲۰٫۷۵۲ ×۲۹

⊲∩° :______

₽₩₽₽₽₽₽₽₽

\$P\$429-00/24405

ᢐᠵᢣᠺ᠉ᡣᠵ᠈᠆ᡣ᠋᠋᠅᠆ᡔ ᠈᠆᠆ᡔ᠅ᠵ᠆ᡔ᠋᠆ᡷ᠆᠘᠆ᡐ᠘ 40-D722

⊳₋∟

٦-∽۲

Rachel Cassie, Principal Investigator c/o University of Alberta Edmonton Clinic Health Academy 3-341, 11405 87 Ave NW Edmonton, AB T6G 1C9 rcassie@ualberta.ca

U of A Ethics Number:

	SCIENTIFIC RESEARCH LICENSE
	LICENSE NUMBER
SSUED TO:	Rachel Cassie
550LD 10.	Edmonton Clinic Health Academy
	University of Alberta
EAM MEMBERS:	S.Chatwood.
LAW MENDERS.	5.Chatwood,
	nunity-driven research to develop programs that prevent and reduce ostance use among youth in
BJECTIVES OF RE	SEARCH:
	lore: 1) how to ensure there is total community control and decision-
	nout the research process, and 2) how to conduct research on sensitive
sure future research	umatizing, especially with youth. This research project is important to with second is community-led and minimizes all potential harms.
	ded to address the harms of colonization, but exploration of how to do this
	nt first step. While similar research has been conducted in other
ommunities,	specific research is important to rebuild the community's trust with
ne research world and	to ensure these practices and policies meet sector 's specific needs.

DATA COLLECTION IN NU: DATES: March 1,2022 to December 31,2022 LOCATION:

Scientific Research License expires on December 31,2022 Issued at Iqaluit, NU on June 30,2022

Jamal Shirley Science Advisor



1/17/22, 11:43 AM

Notification of Approval

Date:	January 5, 2022		
Study ID:			
Principal Investigator:	Rachel Cassie		
Study Supervisor:	Susan Chatwood		
Study Title:	Using community-driven research to develop initiatives that prevent and reduce harmful substance use among youth in the		
Approval Expiry Date:	Wednesday, January 4, 2023		
Sponsor/Funding Agency:	Health Canada	HC	

Thank you for submitting the above study to the Research Ethics Board 1. Your application has been reviewed and approved on behalf of the committee.

Approved Documents:

Recruitment Materials
Adult recruitment poster - example V1 Oct 12, 2021
Youth recruitment poster - example V1 Oct 12, 2021
Letter of Initial Contact
Letter of Initial Contact - V2 Dec 20, 2021
Consent Forms
Adult Informed Consent Forms - community steering committee, clean V3 Dec 20, 2021
Adult Informed Consent Forms - qualitative data participants, clean V3 Dec 20, 2021
Assent Forms
Youth Assent Form - research committee, clean V3 Dec 20, 2021
Youth Assent Form - survey, clean V3 Dec 20, 2021
Questionnaires, Cover Letters, Surveys, Tests, Interview Scripts, etc.
Youth Substance Use Survey - EXAMPLE V1 Oct 12, 2021

Any proposed changes to the study must be submitted to the REB for approval prior to implementation. A renewal report must be submitted next year prior to the expiry of this approval if your study still requires ethics approval. If you do not renew on or before the renewal expiry date, you will have to re-submit an ethics application.

Approval by the REB does not constitute authorization to initiate the conduct of this research. The Principal Investigator is responsible for ensuring required approvals from other involved organizations (e.g., Alberta

1/17/22, 11:43 AM

Health Services, Covenant Health, community organizations, school boards) are obtained, before the research begins.

Sincerely,

Theresa Garvin, Ph.D. Chair, Research Ethics Board 1

Note: This correspondence includes an electronic signature (validation and approval via an online system).

